

A Thesis Submitted for the Degree of PhD at the University of Warwick

Permanent WRAP URL:

http://wrap.warwick.ac.uk/183254

Copyright and reuse:

This thesis is made available online and is protected by original copyright. Please scroll down to view the document itself. Please refer to the repository record for this item for information to help you to cite it.

Our policy information is available from the repository home page.

For more information, please contact the WRAP Team at: wrap@warwick.ac.uk



Microscopic Derivation of Gibbs Measures for the Focusing One Dimensional Nonlinear Schrödinger Equation

by

Andrew Rout

Thesis

Submitted to the University of Warwick

for the degree of

Doctor of Philosophy

Maths Institute

September 2023



Contents

List of	List of Figures				
Acknow	Acknowledgments				
Declar	ations	vi			
Abstra	let	vii			
Chapte	er 1 Introduction	1			
1.1	General introduction	1			
1.2	The nonlinear Schrödinger equation	2			
1.3	Deterministic well-posedness of the NLS	3			
1.4	Gibbs measures	4			
1.5	Many-body quantum mechanics	5			
1.6	Notation and conventions	7			
1.7	Auxiliary results	8			
1.8	Outline of thesis	15			
Chapte	er 2 Main Results	16			
2.1	Cubic problem	16			
	2.1.1 Classical problem	16			
	2.1.2 Quantum problem	21			
	2.1.3 Main results	23			
2.2	Quintic problem	26			
	2.2.1 Classical problem	26			
	2.2.2 Quantum problem	30			
	2.2.3 Main results	32			
2.3	Previously known many body results				
2.4	Outline of the proof	36			

	2.4.1	Cubic problem	36
	2.4.2	Quintic problem	37
Chapt	er3N	Vicroscopic Derivation of the Gibbs Measure for the Cu-	
bic	NLS		39
3.1	Prelin	ninary results and basic estimates	39
	3.1.1	Preliminary results	39
	3.1.2	Basic estimates	40
3.2	The t	ime-independent problem with bounded interaction potential.	
	Proof	of Theorem 2.1.7.	41
	3.2.1	Duhamel expansion	42
	3.2.2	Bounds on the explicit terms	43
	3.2.3	Bounds on the remainder term	44
	3.2.4	The classical setting	46
	3.2.5	Convergence of the explicit terms	47
	3.2.6	Convergence of correlation functions. Proof of Theorem 2.1.7	54
3.3	The t	ime-independent problem with unbounded interaction poten-	
	tials.	Proofs of Theorems 2.1.8 and 2.1.9	57
	3.3.1	L^1 interaction potentials. Proof of Theorem 2.1.8	57
	3.3.2	The delta function. Proof of Theorem 2.1.9 \ldots	59
3.4	Rema	rks about the cut-off function $f \ldots \ldots \ldots \ldots \ldots \ldots$	60
	3.4.1	Interaction potentials of positive type	60
	3.4.2	General L^{∞} interaction potentials $\ldots \ldots \ldots \ldots \ldots \ldots$	61
3.5	The ti	ime-dependent problem	62
	3.5.1	Bounded interaction potentials. Proof of Theorem 2.1.11. $\ . \ .$	62
	3.5.2	Unbounded interaction potentials. Proofs of Theorems 2.1.12	
		and 2.1.13	63
Chant	or 1 N	Aicroscopic Derivation of the Gibbs Measure for the Quin-	_
tic	NLS	incroscopie Derivation of the Gibbs Measure for the Quin-	70
4.1	Analy	sis of the quintic Hartree equation (2.45)	70
	4.1.1	Deterministic local well-posedness and invariance of the Gibbs	.0
		measure	70
	4.1.2	Proof of the approximation lemma from Step 2	82
4.2	The t	ime-independent problem with bounded interaction potentials.	-
	Proof	of Theorem 2.2.6	87
	4.2.1	Basic estimates	88
	4.2.2	Power series expansions of the classical and quantum states .	88

	4.2.3	Analysis of the quantum series (4.83)	90
	4.2.4	Analysis of the classical series (4.88)	92
	4.2.5	Proof of Theorem 2.2.6	93
	4.2.6	Graphical analysis of the untruncated explicit terms. Proof of	
		Proposition 4.2.13	95
4.3	The ti	me-independent problem with unbounded interaction poten-	
	tials.	Proof of Theorem 2.2.7	116
4.4	The ti	me-dependent problem. Proofs of Theorems 2.2.11 and 2.2.12	119
	4.4.1	Bounded interaction potentials. Proof of Theorem 2.2.11 \therefore	119
	4.4.2	Unbounded interaction potentials. Proof of Theorem $2.2.12$.	120
_			
Appen	dix A	Outline of the Construction of the Gibbs Measure for	r
\mathbf{the}	Cubic	NLS	127
A.1	Forma	l arguments	128
A.2	Rigorous arguments		
A.3	Focusing case		
A.4	Auxiliary results		
A.5	Normi	ng sets	133
	A.5.1	Norming sets and ε -nets	133
	A.5.2	Conclusion of the proof of Lemma A.5.1	134
A.6	Proof	of Lemma 3.1.1	135
A.7	Higher	dimensional constructions $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	140
Annen	div P	Creation and annihilation operators	1/19
лрреп Д 1	Dagia	computations	149
D.1	Dasic		144

List of Figures

4.1	An unpaired graph from Definition 4.2.17 with $m = p = 3$. The	
	black dots correspond to factors of $\varphi_{\tau}(\cdot)$ and $\varphi_{\tau}^{*}(\cdot)$. The wavy lines	
	correspond to factors of the interaction potential w	98
4.2	A graph with $m = p = 3$ with a valid pairing from Definition 4.2.20.	98
4.3	An example of the coloured graph from Definition $4.2.27$ with the	
	same pairing as Figure 4.2. The wider dotted edges have colour -1 ,	
	and the finer dotted lines have colour 1. \ldots \ldots \ldots \ldots \ldots \ldots	101
4.4	An unpaired graph with $m = 2, p = 3$ corresponding to Definition	
	4.2.43	115

Acknowledgments

Firstly, I would like to thank my supervisor Dr. Vedran Sohinger, for introducing me to lots of interesting new areas of mathematics. I thank him for all the time and ideas he dedicated to me, and his unwavering enthusiasm for learning and teaching. Without his patience and support, I don't doubt my PhD experience would have been much less rewarding and successful.

I also thank the members of my research group, especially Daniel Ueltschi for his helpful comments and physical intuition. I also thank Georgios Athanasopoulos and Grega Saksida for their helpful conversations. Thanks also to my examiners José Rodrigo and Antti Knowles for their comments on thesis.

Thank you to Mark Pollicott for supervising my MMath masters project and helping me obtain my PhD position.

Special thanks go to those members of the Mathematics Institute who have made my PhD experience especially enjoyable. Particular thanks to Simon Gabriel, David Parmenter, Diana Mocanu, Julie Rasmusen, and Pierre Stepanov. Many thanks to Charleston Winfield, whose friendship during my undergraduate I remember fondly.

I want thank to my many friends from Quiz Society for reminding me there is a world outside of maths. Special thanks to Ethan Webb and Luke Beresford.

I would like to thank my parents for their unending support, for which I am eternally grateful. Finally, I would like to thank all the cats of Coventry who stood in as poor substitutes for my own cats.

Declarations

This thesis is submitted to the University of Warwick in support of my application for the degree of Doctor of Philosophy. It has been composed by myself and has not been submitted in any previous application for any degree. The work has been carried out by myself under the supervision of Vedran Sohinger.

Parts of this thesis have been published or submitted for publication by the author. Parts of Chapter 2 and Chapter 3, as well as Appendix A appear in an article co-authored with Vedran Sohinger, A microscopic derivation of Gibbs measures for the 1D focusing cubic nonlinear Schrödinger equation, due to appear in Communications in Partial Differential Equations [69]. Parts of Chapter 2 and Chapter 4 form a paper with Vedran Sohinger, A microscopic derivation of Gibbs measures for the 1D focusing quintic nonlinear Schrödinger equation, which has been submitted for publication [70].

Abstract

In this thesis, we give a mircoscopic derivation of Gibbs measures for the focusing cubic and quintic (nonlocal) nonlinear Schrödinger equations (NLS) on \mathbb{T} from many body quantum Gibbs states. In the cubic case, this corresponds to taking a twobody interaction, whereas the quintic case corresponds to a three-body interaction. Since we are not making positivity assumptions on the interaction potential, it is necessary to truncate in the mass in the classical setting and the rescaled particle number in the quantum setting.

Our methods are based on the perturbative expansion developed in the work of Fröhlich, Knowles, Schlein, and Sohinger [29]. We obtain results in both the time independent and time dependent cases. These are the first known results in the focusing regime and for any quintic regime. In particular, we give the first microscopic derivation of time-dependent correlation functions for Gibbs measures corresponding to the quintic NLS, as studied in the work of Bourgain [9]. In the quintic case, we can only study a suitable nonlocal quintic NLS, preventing us from obtaining a derivation of the local NLS in the quintic case.

Chapter 1

Introduction

1.1 General introduction

The nonlinear Schrödinger equation (NLS) is a nonlinear partial differential equation which arises in the physics of quantum optics and Bose-Einstein condensation. It is also an example of a nonlinear dispersive equation, which intuitively means its solutions spread out in physical space unless boundary conditions are imposed. This phenomenon is called dispersion.

The well-posedness of the NLS is an interplay between the dispersive properties of the linear Schrödinger equation and the nonlinearity, with the sign playing an integral role. For a positive (defocusing) nonlinearity, one can use conservation of energy and mass, as well as the dispersion to show that the the NLS is well-posed for any sufficiently regular initial condition. For a non-positive (focusing) nonlinearity, blow up of solutions can occur for large initial conditions which are regular. For less regular initial conditions, even in the defocusing case, one cannot expect wellposedness for all initial conditions and the NLS is ill-posed. To get around this, one introduces an invariant (Gibbs) measure, which is supported at low regularity. One is then able to use the invariance of the measure to show that the NLS is globally well-posed for any initial condition in the support of the measure.

On the other hand, the NLS is an effective equation for a many-body quantum system satisfying the many-body linear Schrödinger equation. This means that, under certain conditions, as the number of particles goes to infinity, the solution of the many-body Schrödinger equation is well approximated by the solution of the NLS. This is made rigorous in the sense of reduced density matrices, see [35,41,78]. A natural question is what does the Gibbs measure for the NLS correspond to on the many-body side? This question has been well studied in the case of a defocusing (repulsive) two-body interaction, where it has been shown that the grand canonical ensemble (or quantum Gibbs state) converges to the Gibbs measure for the defocusing cubic NLS in the mean-field limit; see [29–31, 33, 34, 49–53, 80].

In this thesis, we study this question in the case of focusing (attractive) interactions in one dimension – in particular for two and three-body interactions. In this case, the (truncated) grand canonical ensemble corresponds to the (truncated) Gibbs measures for the appropriate focusing cubic and quintic NLS respectively.

1.2 The nonlinear Schrödinger equation

The general nonlinear Schrödinger equation (NLS) is given by

$$\begin{cases} i\partial_t u = -\Delta u + N(u), \\ u(x,0) = u_0 \in H^s(\mathbb{T}), \end{cases}$$
(1.1)

where $u : X \times \mathbb{R} \to \mathbb{C}$ is a function of time and space, N is a nonlinearity, and $H^s(X)$, defined in (1.14) below, denotes the regularity of the initial condition u_0 . We denote by $x \in X$ and $t \in \mathbb{R}$ the spatial and time variables respectively. We will restrict our study to the case where $X := \mathbb{T} \equiv \mathbb{T}^1 \equiv \mathbb{R}/\mathbb{Z} \equiv [-1/2, 1/2)$. We will typically consider consider the following nonlinearities.

$$\begin{split} 1. \ & N(u) := \int_T w(x-y) |u(y)|^2 u(x). \\ 2. \ & N(u) := \int_{\mathbb{T}} dy \, dz \, w(x-y) w(y-z) w(z-x) |u(y)|^2 |u(z)|^2 u(x). \ , \end{split}$$

where $w : \mathbb{T} \to \mathbb{R}$. Throughout we use * to denote convolution with respect to the spatial variable, i.e. $u * v(x) := \int_{\mathbb{T}} u(y)v(x-y)$. When $w \in L^1(\mathbb{T})$, (1.1) is called the (one dimensional) Hartree equation. To differentiate between nonlinearities 1. and 2., we will describe the corresponding equations as the (one dimensional) cubic and quintic Hartree equations respectively. We note that although both nonlinearities depend on time, we will suppress the time from our notation throughout.

Taking $w = \pm \delta$, we recover the local NLS

$$\begin{cases} i\partial_t u = -\Delta u \pm |u|^{2k} u, \\ u(x,0) = u_0 \in H^s(\mathbb{T}), \end{cases}$$
(1.2)

which is so-called because its nonlinearity at (x, t) depends only on the value of u(x, t), which is not true for nonlinearities 1. and 2.

Associated with (1.1) and (1.2) are two invariant quantities

$$\mathcal{N}(u) := \int_{\mathbb{T}} dx \, |u(x)|^2,\tag{1.3}$$

$$H(u) := \int_{\mathbb{T}} dx \, |\nabla u(x)|^2 + E_p(N), \tag{1.4}$$

where \mathcal{N} is the mass and H is the energy, which we also call the Hamiltonian. We split the Hamiltonian into two terms, the kinetic energy and the potential energy, which we denote E_p . The potential energy is given by

1.
$$E_p(N) = \frac{1}{2} \int_{\mathbb{T}} dx \, dy \, |u(x)|^2 w(x-y) |u(y)|^2,$$

2. $E_p(N) = \frac{1}{3} \int_{\mathbb{T}} dx \, dy \, dz |u(x)|^2 |u(y)|^2 |u(z)|^2 w(x-y) w(y-z) w(z-x),$

for the nonlinearities 1. and 2. respectively. Putting $w = \pm \delta$, we recover the potential energy for the local NLS, namely

$$E_p = \pm \frac{1}{2} \int_{\mathbb{T}} |u(x)|^p dx,$$

for p = 4 or p = 6 respectively. The invariance of (1.3) and (1.4) for smooth solutions and potentials can be shown by differentiating under the integral sign. For solutions corresponding to initial conditions below the energy space, one uses a density argument and the persistence of regularity of solutions¹.

Where positivity assumptions are made on the interaction potential, we call (1.1) the defocusing problem. Standard positivity assumptions on the interaction potential are $w \ge 0$ pointwise, or that it is of positive type, meaning $\hat{w} \ge 0$ pointwise. These positivity assumptions mean that the potential energy is positive, which in general simplifies the analysis of the well-posedness properties of (1.1). In contrast, if we make no positivity assumptions on the interaction potential, the problem is called focusing.

1.3 Deterministic well-posedness of the NLS

The deterministic well-posedness of the NLS depends on an interplay between the conservation laws (1.3) and (1.4), the dispersive effects of the linear equation, which gives rise to *Stichartz estimates*, and the strength and sign of the nonlinearity. These estimates give bounds on the solution to the linear Schrödinger equation in terms of the initial condition u_0 . In the periodic case, these estimates rely on techniques from

¹This is only possible above a certain regularity of initial condition; see for example [55,82]

analytic number theory. For a statement of the Strichartz results in the periodic setting, see Lemma 1.7.3 and the discussion after Remark 1.7.4. These estimates are captured by the *dispersive Sobolev spaces*, which are defined in Definition 1.7.5 below.

The Strichartz estimates are used to establish local well-posedness, and the conservation laws are used to show the solutions do not blow up and obtain global well-posedness. Even in the case of a positive nonlinearity, this approach only works up to a certain regularity of initial condition; see [36] for more details on finite time blow up. Precise statements of well-posedness results for various NLS equations can be found for example in [55, Sections 5 and 6] and [82, Section 4]. Moreover, the range of nonlinearities that can be dealt with can be understood with a scaling heuristic, for example see [82, Section 3]. Heuristically, the conservation laws are only defined for sufficiently regular functions, for example H^1 for (1.4). As such, we can only expect to get deterministic global well-posedness above a certain critical regularity. To get around this, we introduce an invariant measure, explained below.

Remark 1.3.1. We note here that the local cubic NLS in one spatial dimension is an integrable system, so has infinitely many conservation laws associated with it. We will only consider the NLS as the limit of the Hartree equation for suitable interaction potentials, so this integrability will not be used, and we make no further comments on it.

1.4 Gibbs measures

To consider the well-posedness of functions with low regularity, we need to introduce a probability measure on the space of initial conditions. To this end, for a defocusing potential, we consider the (heuristically) defined *Gibbs measure* on $L^2(\mathbb{T})$ given by

$$d\mathbb{P}_{\text{Gibbs}} := \frac{1}{z_{\text{Gibbs}}} e^{-H(u)} du, \qquad (1.5)$$

where z_{Gibbs} is a normalisation constant, H is the Hamiltonian defined in (1.4), and du is the (ill-defined) infinite dimensional Lebesgue measure. By drawing analogy to the finite dimensional case, Liouville's theorem suggests this measure should be invariant under the dynamics of the NLS. To make this rigorous, we will realise the Gibbs measure as a weighted *Wiener measure*, which is supported on functions of low regularity. The invariance of this measure acts as a replacement for the conservation laws, and implies probabilistic well-posedness results. The rigorous construction of these measures was first considered in the constructive field theory

literature, for example [37, 58, 59, 75], and they were studied further in [5, 17, 18, 20, 39, 47, 56, 57, 65] and references within. Further details on the rigorous construction of such measures are given in Appendix A. The invariance of the measure was first proved by Bourgain in [9–11], with some preliminary results proved by Zhidkov [89]. This invariance has been extensively studied in the PDE community; we direct the reader to the expository works [19, 60, 64], and for further results [14–16, 24, 40, 65] and the references within. The idea of an invariant measure has been generalised to the study of quasi-invariant measures; see for example [27, 85] and the references therein.

In the case of a focusing potential, H is no longer positive definite, so formally one can have $z_{\text{Gibbs}} = \infty$. Instead one considers the following modification of (1.5)

$$d\mathbb{P}^{f}_{\text{Gibbs}} := \frac{1}{z^{f}_{\text{Gibbs}}} \mathrm{e}^{-H(u)} f(\mathcal{N}) du, \qquad (1.6)$$

where $f \in C_c^{\infty}(\mathbb{R})$ is a suitable cut off function, and z_{Gibbs}^f is a normalisation constant. Such measures were considered in [3, 9, 20, 25, 37, 84], and will be the main Gibbs measures discussed in this thesis.

Given the measure $d\mathbb{P}^f_{\text{Gibbs}}$ as in (1.6) and a well defined time evolution S_t , for example as in (2.5) below, one can consider the corresponding *time-dependent correlation functions*. Namely, for $m \in \mathbb{N}^*$, times $t_1, \ldots, t_m \in \mathbb{R}$, and functions $X^1, \ldots, X^m \in C^{\infty}(\mathfrak{h})$, we define

$$\mathcal{Q}^{f}_{\text{Gibbs}}(X^{1},\ldots,X^{m};t_{1},\ldots,t_{m}) := \int d\mathbb{P}^{f}_{\text{Gibbs}}(\varphi) X^{1}(S_{t_{1}}\varphi) \cdots X^{m}(S_{t_{m}}\varphi) , \quad (1.7)$$

which we call the *m*-particle time-dependent correlation associated with H and $X^{j}, t_{j}, j = 1, ..., m$.

1.5 Many-body quantum mechanics

We can view (1.1) with a cubic non-local linearity as the *classical limit* of a manybody quantum system. Namely, given $n \in \mathbb{N}$, we consider the *n*-body Hamiltonian given by

$$H^{(n)} := \sum_{j=1}^{n} -\Delta_j + \frac{\lambda}{2} \sum_{i \neq j}^{n} w(x_i - x_j), \qquad (1.8)$$

where Δ_j denotes the Laplacian in the j^{th} component and $\lambda > 0$ is a coupling constant. The *n*-body Hamiltonian is defined on the *bosonic n-particle space* $L^2_{\text{sym}}(\mathbb{T}^n)$.

This is the subset of $u \in L^2(\mathbb{T}^n)$ satisfying

$$u(x_1,\ldots,x_n) = u(x_{\sigma(1)},\ldots,x_{\sigma(n)})$$

for any permutation $\sigma \in S_n$. To make both terms in (1.8) have the same order, we take the *interaction strength* $\lambda > 0$ to be of order $\frac{1}{n}$. This scaling regime is known as the *mean-field regime*. The mean-field regime corresponds to short-range (weak) interactions, and is necessary to obtain a non-trivial limit as the number of particles tends to infinity.

We define the n-body Schrödinger equation as

$$i\partial_t \Psi_{n,t} = H^{(n)} \Psi_{n,t},\tag{1.9}$$

where $H^{(n)}$ is as in (1.8). For suitably chosen initial conditions, one wants to compare the dynamics of (1.9) to the dynamics of (1.1) as $n \to \infty$. Rigorous results of this kind were first proved in [35,41,78], where the authors proved convergence in terms of reduced one-particle density matrices. For an expository view on these results, we direct the reader to [6,72]

In this thesis, we study the relationship between the Gibbs measure associated with (1.1) and the *Gibbs states* of the (1.9), which are the equilibrium states of (1.9) at temperature $\tau > 0$. The Gibbs state at temperature $\tau > 0$ is defined as

$$P_{\tau}^{(n)} \equiv P_{\tau}^{f,(n)} := \mathrm{e}^{-H^{(n)}/\tau} f\left(\frac{n}{\tau}\right),$$

which acts on $L^2_{\text{sym}}(\mathbb{T}^n)$, where we recall the cut-off function from (1.6). These states correspond to a cut-off grand canonical ensemble operator.

The main goal of this thesis is to show that one can obtain (1.7) as a *mean-field* limit of corresponding many-body quantum objects, which we henceforth refer to as a *microscopic derivation*. We do this in two steps.

- (i) **Step 1:** Analysis of the *time-independent problem*, i.e. when $t_1 = \cdots = t_m = 0.$
- (ii) Step 2: Analysis of the *time-dependent problem*. This is the general case.

The precise statements of our results can be found in Chapter 2.

For interaction potentials with positivity assumptions, grand canonical ensembles without cut-off were studied in [29–31, 33, 34, 49–53, 80]. An overview of these results can be found in [32]. For the quintic nonlinearity, we instead consider the n-body Hamiltonian defined by

$$H^{(n)} := \sum_{i=1}^{n} -\Delta_j + \frac{\lambda}{3} \sum_{\substack{i \neq j \neq k \\ i \neq k}}^{n} w(x_i - x_j) w(x_j - x_k) w(x_k - x_i),$$
(1.10)

where in this case $\lambda \sim \frac{1}{n^2}$. In this case, the classical limits of the cubic equation should be seen as a guiding heuristic. Some results for the quintic case were proved in [21, 22, 48, 88], however these are not directly related to the quintic equation studied in this thesis. For further details, we direct the reader to Section 2.3.

1.6 Notation and conventions

We adopt the convention that $\mathbb{N} = \{0, 1, 2, ...\}$ is the set of non-negative integers and we write $\mathbb{N}^* := \{1, 2, ...\}$. We write C to denote generic positive constants that can change line to line. If C depends on a set of parameters $\{a_1, ..., a_n\}$, we will write $C(a_1, ..., a_n)$. We also sometimes write $a \leq b$ to denote $a \leq Cb$, moreover we write $a \leq_{a_1,...,a_n}$ is $a \leq C(a_1, ..., a_n)b$. Similarly we use \sim to denote equality up to a constant. $C_c^{\infty}(U)$ will be used to denote the set of compactly supported smooth functions $f: U \to \mathbb{R}$.

We write **1** to denote the identity operator on a Hilbert space \mathcal{H} and write $\mathcal{L}(\mathcal{H})$ for the space of bounded linear operators on \mathcal{H} . Moreover, if \mathcal{H} is separable and $p \in [1, \infty]$, we define the Schatten space $\mathfrak{S}^p(\mathcal{H})$ to be the set of $\mathcal{A} \in \mathcal{L}(\mathcal{H})$ with $\|\mathcal{A}\|_{\mathfrak{S}^p(\mathcal{H})} < \infty$, where

$$\|\mathcal{A}\|_{\mathfrak{S}^{p}(\mathcal{H})} := \begin{cases} (\mathrm{Tr}|\mathcal{A}|^{p})^{1/p} & \text{if } p < \infty \\ \sup \operatorname{sup spec}|\mathcal{A}| & \text{if } p = \infty. \end{cases}$$
(1.11)

Here $|\mathcal{A}| := \sqrt{\mathcal{A}^* \mathcal{A}}$ and spec denotes the spectrum of \mathcal{A} .

For $p \in \mathbb{N}$, we define $\mathfrak{h}^{(p)} := L^2_{\text{sym}}(\mathbb{T}^p)$, and we note $\mathfrak{h} := \mathfrak{h}^{(1)} \equiv L^2_{\text{sym}}(\mathbb{T}) \equiv L^2(\mathbb{T})$. We also write

$$\mathfrak{B}_p := \{ \xi \in \mathfrak{S}^2(\mathfrak{h}^{(p)}) : \|\xi\|_{\mathfrak{S}^2(\mathfrak{h}^{(p)})} \le 1 \},$$
(1.12)

$$\mathcal{C}_p := \mathfrak{B}_p \cup \mathbf{1}_p, \tag{1.13}$$

where $\mathbf{1}_p$ is the identity operator on $\mathfrak{h}^{(p)}$.

If ξ is a closed linear operator on $\mathfrak{h}^{(p)}$, we can identify it with its Schwartz

kernel, which we write as $\xi(x_1, \ldots, x_p; y_1, \ldots, y_p)$; see for example [66, Corollary V.4.4].

For a set A, the indicator function on A is given by

$$\chi_A(x) := \begin{cases} 0 & \text{if } x \notin A \\ 1 & \text{if } x \in A. \end{cases}$$

Where clear, we also adopt the convention of dropping the space we are integrating over from our integrals.

1.7 Auxiliary results

Harmonic analysis

We take the convention that the *Fourier coefficients* for a function $u \in L^1(\mathbb{T})$ are given by

$$\hat{u}(k) := \int dx \, u(x) \mathrm{e}^{-2\pi i k x}$$

Then we define the Sobolev space of regularity $s \in \mathbb{R}$ to be the space

$$H^{s}(\mathbb{T}) := \{ u \in \mathcal{D}'(\mathbb{T}) : \sum_{k \in \mathbb{Z}} |\hat{u}(k)|^{2} (1 + |k|^{2})^{s} < \infty \},$$
(1.14)

where $\mathcal{D}'(\mathbb{T})$ is the space of distributions with test functions taken in $C_c^{\infty}(\mathbb{T})$. Then

$$\|u\|_{H^{s}(\mathbb{T})}^{2} := \sum_{k \in \mathbb{Z}} |\hat{u}(k)|^{2} (1 + 2\pi |k|^{2})^{s} \equiv \left\|\hat{u}(k)(1 + 2\pi |k|^{2})^{s/2}\right\|_{\ell_{k}^{2}}^{2}$$

We also use the notation

$$H^{s-}(\mathbb{T}) := \bigcup_{s' < s} H^{s'}(\mathbb{T}).$$

The following embedding theorem for Sobolev spaces into L^p spaces holds; see for example [7].

Theorem 1.7.1 (Sobolev embedding theorem for d = 1). Let $p \in (1, \infty]$, s > 0, and $\frac{1}{p} = \frac{1}{2} - s$. Then

$$\|u\|_{L^p(\mathbb{T})} \lesssim_{p,s} \|u\|_{H^s(\mathbb{T})}.$$

Finally, we will also need to use the theory of Fourier multipliers on the torus, so we need the following extension of the Mikhlin multiplier theorem. The statement and proof of the following theorem in full generality can be found in [81, VII, Theorem 3.8]. Here we use the notation $(L^p(X), L^p(X))$ to denote the class of bounded operators from $L^p(X)$ to $L^p(X)$ which commute with translations.

Lemma 1.7.2 (Mikhlin multiplier theorem for \mathbb{T}). Let $p \in [1, \infty]$ and $T \in (L^p(\mathbb{R}), L^p(\mathbb{R}))$ be a Fourier multiplier operator. Let \hat{u} be the multiplier corresponding to T and suppose that \hat{u} is continuous at every point of \mathbb{Z} . For $k \in \mathbb{Z}$, let $\lambda(k) := \hat{u}(k)$. Then there is a unique periodised lattice operator \tilde{T} defined by

$$\tilde{T}f(x) \sim \sum_{k \in \mathbb{Z}} \lambda(k) \hat{f}(k) \mathrm{e}^{2\pi i k x}$$

such that $\tilde{T} \in (L^p(\mathbb{T}), L^p(\mathbb{T}))$ and $\|\tilde{T}\|_{L^p \to L^p} \le \|T\|_{L^p \to L^p}$.

Mixed L^p spaces

Throughout, we will make use of so called "mixed L^p spaces." Formally, for $p, q \in [1, \infty]$, Banach spaces X, Y, and $u: Y \times X \to \mathbb{C}$, we define

$$\|u\|_{L^p_X L^q_Y} := \|\|u(\cdot, x)\|_{L^q_Y}\|_{L^p_X}.$$
(1.15)

In other words, for each $x \in X$, we compute the L^q norm of $y \mapsto u_x(y) := u(y, x)$, and then we compute the L^p norm of the function $x \mapsto ||u(x, y)||_{L^q_Y}$. This is all formal because we are assuming that the quantity in (1.15) is well defined. This can be made rigorous using the theory of Bochner integrals and Bochner spaces, for example see [73]. Where p = q, we sometimes write $L^p_X L^q_Y \equiv L^p_{X,Y}$. One important duality result which holds is that, for $p, q \in (1, \infty)$ and Banach spaces X and Y

$$(L^p L^q (X \times Y))^* \cong L^{p'} L^{q'} (X \times Y),$$

where p' and q' denote the Hölder conjugate exponents of p and q respectively (1/p + 1/p' = 1).

Strichartz estimates and $X^{s,b}$ spaces

A solution to the linear Schrödinger equation is a function $u: \mathbb{T} \times \mathbb{R} \to \mathbb{C}$ satisfying

$$\begin{cases} i\partial_t u = -\Delta u, \\ u(x,0) = u_0(x) \in H^s. \end{cases}$$
(1.16)

Solutions to (1.16) can be written as $u(x,t) = e^{it\Delta}u_0(x)$, where

$$e^{it\Delta}u_0(x) = \sum_k \hat{u}_0(k)e^{-4\pi^2ik^2t + 2\pi ikx}$$

We call $e^{it\Delta}$ the *free Schrödinger kernel*. Solutions to the linear Schrödinger equation satisfy the following Strichartz estimates, as proven in [8].

Lemma 1.7.3 (Strichartz estimates for T). Suppose that $u_0 = \sum_k a_k e^{2\pi i kx} \in L^2(\mathbb{T})$. Let the Dirichlet projection $P_N : L^2(\mathbb{T}) \to L^2(\mathbb{T})$ be defined by $P_N u_0(x) = \sum_{|k| < N} a_k e^{2\pi i kx}$. Suppose that $u : \mathbb{T} \times \mathbb{R} \to \mathbb{C}$ is a solution of the linear Schrödinger equation. Then

- 1. $\|e^{it\Delta}u_0\|_{L^4_t L^4_x} \lesssim \|u_0\|_{L^2_x}$.
- 2. Suppose that $\operatorname{supp}(\hat{u}) \subset \{k : |k| \leq N\}$ (in other words $a_k = 0$ for any |k| > N). Then

$$\|\mathrm{e}^{\imath t\Delta} P_N u_0\|_{L^6_{t,x}} \lesssim N^{\varepsilon} \|P_N u_0\|_{L^2_x}.$$

3. Under the same assumptions as in (2), we have

$$\|\mathrm{e}^{it\Delta}u_0\|_{L^q_{t,x}} \lesssim N^{\frac{1}{2}-\frac{3}{q}+\varepsilon}\|P_N u_0\|_{L^2_x},$$

for any $q \in (6, \infty]$.

Remark 1.7.4. We note that the ε loss of derivatives in 2. cannot be omitted. Indeed, in [8, (2.45)], Bourgain proved

$$\frac{1}{N^{1/2}} \| e^{it\Delta} \sum_{n=0}^{N} e^{2\pi i (nx+n^2t)} \|_{L^6_{t,x}} \to \infty$$

as $N \to \infty$.

In *n* dimensions, for a function $f \in L^2(\mathbb{T}^n)$ with $P_N f = f$, we expect to find estimates of the form (since the pair $\left(\frac{2(n+2)}{n}, \frac{2(n+2)}{n}\right)$ is admissible in dimension *n*).

$$\|\mathbf{e}^{it\Delta}f\|_{L^{q}_{t}L^{q}_{x}} \lesssim \|f\|_{L^{2}_{x}}, \quad q < \frac{2(n+2)}{n}.$$
(1.17)

$$\|e^{it\Delta}f\|_{L^{q}_{t}L^{q}_{x}} \lesssim N^{\varepsilon}\|f\|_{L^{2}_{x}}, \quad q = \frac{2(n+2)}{n}.$$
(1.18)

$$\|\mathrm{e}^{it\Delta}f\|_{L^{q}_{t}L^{q}_{x}} \lesssim N^{\frac{n}{2} - \frac{n+2}{q} + \varepsilon} \|f\|_{L^{2}_{x}}, \quad q > \frac{2(n+2)}{n}.$$
 (1.19)

For n = 1, Bourgain proved these estimates in [8]. In the same paper, he also proved (1.17) for n = 2, 3, and he proved (1.18) and (1.19) for n = 3 under the assumption that $q \ge 4$. They were proven for all n, q under the assumption that the Fourier coefficients lie on a paraboloid and similar results proven for irrational tori in [12, Section 2].

When proving the well-posedness of dispersive PDEs, one makes use of the *dispersive Sobolev spaces*, or $X^{s,b}$ spaces. For a function $u : \mathbb{T} \times \mathbb{R} \to \mathbb{C}$, we define its spacetime Fourier transform to be

$$\tilde{u}(k,\eta) := \int_{\mathbb{R}} dt \int_{\mathbb{T}} u(x,t) e^{-2\pi i x k - 2\pi i \eta t}$$

Definition 1.7.5. Let $s, b \in \mathbb{R}$ and $u : \mathbb{T} \times \mathbb{R} \to \mathbb{C}$, we define

$$\|u\|_{X^{s,b}} := \left\| (1+|2\pi k|)^s (1+|\eta+2\pi k^2|)^b \tilde{u}(k,\eta) \right\|_{\ell^2_k(\mathbb{Z})L^2_\eta(\mathbb{R})} = \left\| e^{it\Delta} u \right\|_{H^s_x(\mathbb{T})H^b_x(\mathbb{R})}.$$
(1.20)

The final characterisation means that the $X^{s,b}$ norm heuristically measures how far a function is from being a solution of the linear Schrödinger equation. For an interval $I \subset \mathbb{R}$, the local $X^{s,b}$ space, denoted $X_I^{s,b}$ is defined as

$$\|u\|_{X_{I}^{s,b}} := \inf_{v} \|v\|_{X^{s,b}}, \tag{1.21}$$

where the infimum is taken over the set of functions $v : \mathbb{T} \times \mathbb{R} \to \mathbb{C}$ with $v|_{\mathbb{T} \times I} = u|_{\mathbb{T} \times I}$. We recall the following results about local $X^{s,b}$ spaces.

Lemma 1.7.6 (Properties of $X^{s,b}$ spaces). Let us fix $b = \frac{1}{2} + \varepsilon$ for $\varepsilon > 0$ small and $s \in \mathbb{R}$. The following estimates hold for all $t_0 \in \mathbb{R}$ and $\delta > 0$.

- (i) For all $u \in X^{s,b}_{[t_0,t_0+\delta]}$, we have $\|u\|_{L^{\infty}_{t\in[t_0,t_0+\delta]}H^s_x} \lesssim_b \|u\|_{X^{s,b}_{[t_0,t_0+\delta]}}$.
- (ii) For all $\Phi \in H^s$, we have

$$\|\mathrm{e}^{\mathrm{i}(t-t_0)\Delta}\Phi\|_{X^{s,b}_{[t_0,t_0+\delta]}} \lesssim_b \|\Phi\|_{H^s}.$$
 (1.22)

(iii) For all $F \in X^{s,b-1}_{[t_0,t_0+\delta]}$, we have

$$\left\| \int_{t_0}^t dt' \,\mathrm{e}^{\mathrm{i}(t-t')\Delta} F(\cdot,t') \right\|_{X^{s,b}_{[t_0,t_0+\delta]}} \lesssim_b \|F\|_{X^{s,b-1}_{[t_0,t_0+\delta]}}.$$
 (1.23)

Proof of Lemma 1.7.6. The estimates above are standard. Claim (i) follows by

Sobolev embedding in the time variable, using the assumption that $b > \frac{1}{2}$. The particular form of the estimates given by (ii) and (iii) is proved in a self-contained way in [26, Lemma 3.10] and [26, Lemma 3.12] respectively². In particular, the arguments from the proof of [26, Lemma 3.10] imply that for all $\Phi \in H^s$ and $\psi \in C_c^{\infty}(\mathbb{R})$, we have

$$\|\psi(t) e^{it\Delta} \Phi\|_{X^{s,b}_{[t_0,t_0+\delta]}} \lesssim_{b,\psi} \|\Phi\|_{H^s}.$$

We now consider ψ which is equal to 1 on $[t_0, t_0 + \delta]$ to deduce that

$$\|\mathrm{e}^{\mathrm{i}t\Delta}\Phi\|_{X^{s,b}_{[t_0,t_0+\delta]}} \lesssim_b \|\Phi\|_{H^s} \,. \tag{1.24}$$

The estimate (1.24) follows from (1.22) and the unitarity of $e^{it\Delta}$.

The argument for (iii) is similar. When $t_0 = 0$, it follows from the proof of [26, Lemma 3.12]. More precisely, it suffices to show that for all nonnegative $\psi \in C_c^{\infty}(\mathbb{R})$, we have

$$\left\|\psi(t)\int_{0}^{t} dt' \,\mathrm{e}^{\mathrm{i}(t-t')\Delta} F(\cdot,t')\right\|_{X^{s,b}} \lesssim_{b,\psi} \|F\|_{X^{s,b-1}}.$$
 (1.25)

The claim of (iii) then follows from (1.25) by taking infima over F as in (1.21). Using the fact that $\psi(t)$ and $e^{it\Delta}$ commute, as well as (1.20), we have that the expression on the left-hand side of (1.25) equals

$$\left\|\psi(t)\int_0^t dt' \,\mathrm{e}^{-\mathrm{i}t'\Delta} F(\cdot,t')\right\|_{H^s_x H^b_t} \le \left\|\psi(t)\int_0^t dt' \left\|\mathrm{e}^{-\mathrm{i}t'\Delta} F(\cdot,t')\right\|_{H^s_x}\right\|_{H^b_t}.$$
 (1.26)

Using (1.20) and (1.26), we note that (1.25) follows from the following inequality.

$$\left\|\psi(t) \int_{0}^{t} dt' g(t')\right\|_{H_{t}^{b}} \lesssim_{b,\psi} \|g\|_{H_{t}^{b'}}.$$
(1.27)

The estimate (1.27) corresponds to [26, (3.18)], which is shown in the proof of [26, Lemma 3.12]. This shows (1.23) when $t_0 = 0$.

The proof of (1.23) for general t_0 follows by translation. In order to explain the last step in more detail, let us consider $F \in X^{s,b-1}_{[t_0,t_0+\delta]}$. We then consider $G(x,t) := F(x,t+t_0)$ and observe that

$$\|G\|_{X^{s,b-1}_{[0,\delta]}} = \|F\|_{X^{s,b-1}_{[t_0,t_0+\delta]}}.$$
(1.28)

²The results in [26] are proved for the Airy semigroup $W_t = e^{-t\delta_x^3}$ on the real line. The arguments for the Schrödinger semigroup $e^{it\Delta}$ on the torus follow analogously.

A direct calculation shows that

$$\int_{t_0}^t dt' \,\mathrm{e}^{\mathrm{i}(t-t')\Delta} F(t') = \int_0^{t-t_0} dt' \,\mathrm{e}^{\mathrm{i}(t-t_0-t')\Delta} \,G(t') \,, \tag{1.29}$$

from where we deduce

$$\left\|\int_{t_0}^t dt' \operatorname{e}^{\operatorname{i}(t-t')\Delta} F(\cdot,t')\right\|_{X^{s,b}_{[t_0,t_0+\delta]}} = \left\|\int_0^t dt' \operatorname{e}^{\operatorname{i}(t-t')\Delta} G(t')\right\|_{X^{s,b}_{[0,\delta]}}.$$
 (1.30)

The result of claim (iii) then follows from (1.28)–(1.30) and (1.23) when $t_0 = 0$. \Box

Remark 1.7.7. We refer the reader to [30, Lemma 5.3], [30, Appendix A] for a self-contained summary of similar estimates in $X^{s,b}$ spaces, based on [43, 44]. We also refer the reader to [83, Section 2.6]. Note that, in contrast to [30, Lemma 5.3], in (ii)–(iii), we are working in local $X^{s,b}$ spaces instead of with a cut-off function ψ in the t variable.

We also have the following standard result, which states that $X^{s,b}$ and consequently local $X^{s,b}$ spaces interpolate nicely. We prove it for completeness, but the reader could also find a proof in [38]. We use it to interpolate between estimates for $X^{s,b}$ estimates.

Lemma 1.7.8. Suppose $s, b \in \mathbb{R}$. Let $s = \theta s_1 + (1 - \theta)s_2$ and $b = \theta b_1 + (1 - \theta)b_2$. Then we have the following inequality.

$$\|\cdot\|_{X^{s,b}} \le \|\cdot\|_{X^{s_1,b_1}}^{\theta}\|\cdot\|_{X^{s_2,b_2}}^{1-\theta}.$$

Proof.

$$\begin{split} \|u\|_{X^{s,b}}^{2} &= \left\| |\tilde{u}(k,\eta)|^{2} (1+|2\pi k|^{2})^{2s} (1+|\eta+2\pi k^{2}|)^{2b}) \right\|_{\ell_{k}^{1}(\mathbb{Z})L_{\eta}^{1}(\mathbb{R})} \\ &= \left\| \left(|\tilde{u}(k,\eta)|^{2} (1+|2\pi k|^{2})^{2s_{1}} (1+|\eta+2\pi k^{2}|)^{2s_{1}} \right)^{\theta} \\ &\times \left(|\tilde{u}(k,\eta)|^{2} (1+|2\pi k|^{2})^{2s_{2}} (1+|\eta+2\pi k^{2}|)^{2s_{2}} \right)^{1-\theta} \right\|_{\ell_{k}^{1}(\mathbb{Z})L_{\eta}^{1}(\mathbb{R})} \\ &\leq \left\| \left(|\tilde{u}(k,\eta)|^{2} (1+|2\pi k|^{2})^{2s_{1}} (1+|\eta+2\pi k^{2}|)^{2s_{1}} \right)^{\theta} \right\|_{\ell_{k}^{\frac{1}{\theta}}(\mathbb{Z})L_{\eta}^{\frac{1}{\theta}}(\mathbb{R})} \\ &+ \left\| \left(|\tilde{u}(k,\eta)|^{2} (1+|2\pi k|^{2})^{2s_{2}} (1+|\eta+2\pi k^{2}|)^{2s_{2}} \right)^{1-\theta} \right\|_{\ell_{k}^{\frac{1}{1-\theta}}(\mathbb{Z})L_{\eta}^{\frac{1}{1-\theta}}(\mathbb{R})} \\ &= \|u\|_{X^{s_{1},b_{1}}}^{2\theta} \|u\|_{X^{s_{2},b_{2}}}^{2(1-\theta)}, \end{split}$$

where the inequality follows from Hölder's inequality.

Results for Gibbs measures

In the construction of Gibbs measures, we will need the following result about finite dimensional Hamiltonian systems; see for example [4]. Recall we say that for a map $T: X \to X$, we say a measure μ is *invariant under* T if $d\mu(A) = d\mu(T^{-1}A)$, for any measurable set A.

Theorem 1.7.9 (Liouville's theorem). Let $H := H(p_1, \ldots, p_n, q_1, \ldots, q_n)$ be a Hamiltonian. For a finite dimensional Hamiltonian system with evolution equations

$$\begin{cases} \dot{p}_j = \frac{\partial H}{\partial q_j} \\ \dot{q}_j = -\frac{\partial H}{\partial p_j} \end{cases}$$

and a non-negative smooth function g, g(H) dLeb is invariant under the flow map, S_t .

We will also use the following result to compute the expectations of Gaussian random variables. For an example of a self-contained introduction, see [31, Lemma 2.4].

Theorem 1.7.10 (Wick's theorem). Let $(g_k)_{k=1}^n$ be i.i.d. centred complex Gaussian random variables. Let $(g_k)^*$ denote either g_k or $\overline{g_k}$. Then

$$\mathbb{E}\left[\prod_{j=1}^{n} (g_k)^{*_i}\right] = \sum_{\Pi \in \mathcal{M}(n)} \prod_{(i,j) \in \Pi} \mathbb{E}\left[(g_i)^{*_i} (g_j)^{*_j}\right],\tag{1.31}$$

where the sum is taken over all complete pairings of $\{1, \ldots, n\}$, and where edges of Π are denoted by (i, j) with i < j.

Remark 1.7.11. If g_1 and g_2 are independent centred complex random variables, we note that $\mathbb{E}[g_1g_2] = \mathbb{E}[\overline{g_1g_2}] = 0$. This observation allows us to simplify the sum on the right hand side of (1.31).

Feynman-Kac formula

In our analysis, we make use of the Feynman-Kac formula. To this end, let T > 0and let Ω^T denote the space of continuous paths $\omega : [0,T] \to \mathbb{T}$. Given t > 0, we define

$$\psi^{t}(x) := e^{t\Delta}(x) = \sum_{n \in \mathbb{Z}^{d}} (4\pi t)^{-1/2} e^{-|x-n|^{2}/4t}$$
(1.32)

to be the periodic heat kernel on \mathbb{T} . For $x, \tilde{x} \in \mathbb{T}$, we characterise the Wiener measure $\mathbb{W}_{x,\tilde{x}}^T$ by its finite-dimensional distribution. Namely for $0 < t_1 < \ldots < t_n < T$ and $f: \mathbb{T}^n \to \mathbb{R}$ continuous

$$\int \mathbb{W}_{x,\tilde{x}}^{T}(d\omega) f(\omega(t_{1}), \dots, \omega(t_{n}))$$

= $\int dx_{1} \dots dx_{n} \psi^{t_{1}}(x_{1} - \tilde{x}) \psi^{t_{2}-t_{1}}(x_{2} - x_{1}) \dots$
 $\times \psi^{t_{n}-t_{n-1}}(x_{n} - x_{n-1}) \psi^{T-t_{n}}(x - x_{n}) f(x_{1}, \dots, x_{n}).$

Then we have the following result, see for example [67, Theorem X.68].

Proposition 1.7.12 (Feynman-Kac Formula). Let $V : \mathbb{T} \to \mathbb{C}$ be continuous and bounded below. For t > 0

$$e^{t(\Delta-V)}(x;\tilde{x}) = \int \mathbb{W}_{x,\tilde{x}}^t(d\omega) e^{-\int_0^t ds \, V(\omega(s))} \, .$$

1.8 Outline of thesis

The thesis consists of two main parts, namely the cubic and the quintic problems. In Chapter 2, we state the main results of the thesis, and explain the main differences in the analysis of the two problems. In Chapter 3, we deal with the cubic case, and in Chapter 4 we analyse the quintic problem. Appendix A is devoted to an exposition of the construction of the one dimensional Gibbs measure for the focusing 1D NLS and to a self-contained proof of the integrability of the weight function. Finally, Appendix B includes some basic computations within the second quantisation framework.

Chapter 2

Main Results

In this section we give the setup of our problem and a statement of the main results proved in the thesis. We begin by setting up the problem and stating our results for the cubic nonlinearity and then explain the differences for the quintic nonlinearity. In Section 2.3 we give an outline of the previously known results in the field.

2.1 Cubic problem

2.1.1 Classical problem

We fix the Hilbert space $\mathfrak{h} := L^2(\mathbb{T}; \mathbb{C}) \equiv L^2(\mathbb{T})$. To define the Hamiltonian we work with, we first make the following assumption.

Assumption 2.1.1 (The interaction potential). We consider an *interaction potential* which is of one of the following types.

- (i) $w: \mathbb{T} \to \mathbb{R}$ is even and belongs to $L^1(\mathbb{T})$.
- (ii) $w = -\delta$, where δ is the Dirac delta function.

Let us note that, in Assumption 2.1.1, we do not assume any conditions on the sign of \hat{w} (pointwise almost everywhere).

With w as in Assumption 2.1.1, the Hamiltonian that we consider is given by

$$H(\varphi) := \int dx \left(|\nabla \varphi(x)|^2 + \kappa |\varphi(x)|^2 \right) + \frac{1}{2} \int dx \, dy \, |\varphi(x)|^2 \, w(x-y) |\varphi(y)|^2 \,. \tag{2.1}$$

In (2.1), and throughout the sequel, we fix $\kappa > 0$ to be the *(negative) chemical* potential. On the space of fields $\varphi : \mathbb{T} \to \mathbb{C}$, we consider a Poisson bracket defined

$$\{\varphi(x),\overline{\varphi}(y)\} = i\delta(x-y), \quad \{\varphi(x),\varphi(y)\} = \{\overline{\varphi}(x),\overline{\varphi}(y)\} = 0.$$
(2.2)

We note that, by Assumption 2.1.1, the Hamiltonian (2.1) is not necessarily positive-definite. Hence, when studying the associated Gibbs measure, one has to use the modification given by (1.6), instead of (1.5). This setup was previously used in [9, 20, 47].

The Hamiltonian equation of motion associated with Hamiltonian (2.1) and Poisson bracket (2.2) is the time-dependent nonlocal *nonlinear Schrödinger equation* (NLS)

$$i\partial_t\varphi(x) = (-\Delta + \kappa)\varphi(x) + \int dy \,|\varphi(y)|^2 \,w(x-y)\varphi(x)\,. \tag{2.3}$$

Here, we abbreviate the notation $\varphi(x) \equiv \varphi(x,t)$ with $\varphi : \mathbb{T} \times \mathbb{R} \to \mathbb{C}$. For $w \in L^1$, as in Assumption 2.1.1 (i), one usually refers to (2.3) as the Hartree equation. We will also consider the focusing local cubic NLS

$$i\partial_t\varphi(x) = (-\Delta + \kappa)\varphi(x) - |\varphi(x)|^2\varphi(x), \qquad (2.4)$$

which corresponds to (2.3) with $w = -\delta$, as in Assumption 2.1.1 (ii). We refer to¹ (2.3) and (2.4) as the *focusing cubic nonlinear Schrödinger equation (NLS)*.

The arguments in [8] show that the focusing cubic NLS (2.3)–(2.4) is globally well-posed for initial data in $\mathfrak{h} \equiv L^2(\mathbb{T})$. In particular, there exists a well-defined solution map S_t that maps any initial data $\varphi_0 \in \mathfrak{h}$ to the solution at time t given by

$$\varphi(\cdot) \equiv \varphi(\cdot, t) := S_t \varphi_0(\cdot) \in \mathfrak{h} \,. \tag{2.5}$$

Moreover, $||S_t\varphi_0||_{\mathfrak{h}} = ||\varphi_0||_{\mathfrak{h}}$.

The one-particle space on which we work is $\mathfrak{h} = L^2(\mathbb{T})$. We use the following convention for the scalar product.

$$\langle g_1, g_2 \rangle_{\mathfrak{h}} := \int dx \,\overline{g_1}(x) \, g_2(x) \, .$$

We consider the one-body Hamiltonian given by

$$h := -\Delta + \kappa, \tag{2.6}$$

¹When one has suitable positivity (in other words *defocusing*) assumptions on w, the analysis of the problem we are considering for (2.3) has already been done in [29]; see Section 2.3 below for an overview. Our main interest lies in the case when these assumptions are relaxed, which we refer to as the *focusing* regime.

where $\kappa > 0$ is as in (2.1). This is a positive self-adjoint densely defined operator on \mathfrak{h} . We can write *h* spectrally as

$$h := \sum_{k \in \mathbb{N}} \lambda_k u_k u_k^*, \tag{2.7}$$

where

$$\lambda_k := 4\pi^2 |k|^2 + \kappa \tag{2.8}$$

are the eigenvalues of h and

$$u_k := e^{2\pi i k x} \tag{2.9}$$

are the normalised eigenvalues of h on \mathfrak{h} . Since we are working on \mathbb{T} , we have

$$\operatorname{Tr}(h^{-1}) = \sum_{k \in \mathbb{N}} \frac{1}{4\pi^2 |k|^2 + \kappa} < \infty,$$
(2.10)

where the trace is taken over \mathfrak{h} .

For each $k \in \mathbb{N}$, we define μ_k to be a standard complex Gaussian measure. In other words, $\mu_k := \frac{1}{\pi} e^{-|z|^2} dz$, where dz is the Lebesgue measure on \mathbb{C} . Let $(\mathbb{C}^{\mathbb{N}}, \mathcal{G}, \mu)$ be the product probability space with

$$\mu := \bigotimes_{k \in \mathbb{N}} \mu_k. \tag{2.11}$$

We denote elements of the probability space $\mathbb{C}^{\mathbb{N}}$ by $\omega = (\omega_k)_{k \in \mathbb{N}}$. Let the *classical* free field $\varphi \equiv \varphi^{\omega}$ be defined by

$$\varphi := \sum_{k \in \mathbb{N}} \frac{\omega_k}{\sqrt{\lambda_k}} u_k. \tag{2.12}$$

Note that (2.10) implies (2.12) converges almost surely in $H^{\frac{1}{2}-\varepsilon}(\mathbb{T})$ for $\varepsilon > 0$ arbitrarily small.

Since μ is a Gaussian measure, it satisfies the following Wick theorem, which follows from Theorem 1.7.10.

Proposition 2.1.2. Let φ be as in (2.12). Given $g \in H^{-\frac{1}{2}+\varepsilon}$ for $\varepsilon > 0$, we let $\varphi(g) := \langle g, \varphi \rangle$ and $\overline{\varphi}(g) := \langle \varphi, g \rangle$. Furthermore, we let $(\varphi)^*(g)$ denote either $\varphi(g)$ or $\overline{\varphi}(g)$. Then, given $n \in \mathbb{N}^*$ and $g_1, \ldots, g_n \in H^{-\frac{1}{2}+\varepsilon}$, we have

$$\mathbb{E}_{\mu}\left[\prod_{i=1}^{n} \left(\varphi(g_{i})\right)^{*_{i}}\right] = \sum_{\Pi \in \mathcal{M}(n)} \prod_{(i,j) \in \Pi} \mathbb{E}_{\mu}\left[\left(\varphi(g_{i})\right)^{*_{i}} \left(\varphi(g_{j})\right)^{*_{j}}\right],$$
(2.13)

where the sum is taken over all complete pairings of $\{1, \ldots, n\}$, and where edges of Π are denoted by (i, j) with i < j.

We note that, by gauge invariance, for all $(i, j) \in \Pi$

$$\mathbb{E}_{\mu}\left[\varphi(g_i)\varphi(g_j)\right] = \mathbb{E}_{\mu}\left[\overline{\varphi}(g_i)\,\overline{\varphi}(g_j)\right] = 0\,.$$

Therefore, each non-zero factor arising on the right-hand side of (2.13) can be computed using

$$\int d\mu \,\overline{\varphi}(\tilde{g})\varphi(g) = \langle g, h^{-1}\tilde{g} \rangle,$$

for $g, \tilde{g} \in H^{-\frac{1}{2}+\varepsilon}$. Here, the Green function h^{-1} is the covariance of μ . We note that, under a suitable pushforward, we can identify μ with a probability measure on H^s ; see for example [29, Remark 1.3]. As in [29], we work directly with the measure μ as above and do not use this identification.

Remark 2.1.3. This is a fairly brief introduction to the Gibbs measure so that we can state our results. A longer introduction and details of the construction of the Gibbs measure in one spatial dimension is given in Appendix A.

Given $p \in \mathbb{N}^*$, the *p*-particle space $\mathfrak{h}^{(p)}$ is defined as the symmetric subspace of $\mathfrak{h}^{\otimes p}$, i.e. $u \in \mathfrak{h}^{(p)}$ if and only if for any permutation π ,

$$u(x_{\pi(1)},\ldots,x_{\pi(p)})=u(x_1,\ldots,x_p)$$

For ξ a closed linear operator on $\mathfrak{h}^{(p)}$ and φ as in (2.12), we define the random variable

$$\Theta(\xi) := \int dx_1 \dots dx_p \, dy_1 \dots dy_p \, \xi(x_1, \dots, x_p; y_1, \dots, y_p) \\ \times \overline{\varphi}(x_1) \dots \overline{\varphi}(x_p) \varphi(y_1) \dots \varphi(y_p) \,. \tag{2.14}$$

Recall we denote by $\mathcal{L}(\mathcal{H})$ the set of all bounded operators on a Hilbert Space \mathcal{H} . If $\xi \in \mathcal{L}(\mathfrak{h}^{(p)})$, then $\Theta(\xi)$ defined in (2.14) is almost surely well-defined, since $\varphi \in \mathfrak{h}$ almost surely.

Given w as in Assumption 2.1.1, we define the *classical interaction* as

$$\mathcal{W} := \frac{1}{2} \int dx \, dy \, |\varphi(x)|^2 \, w(x-y) |\varphi(y)|^2 \,. \tag{2.15}$$

The *free classical Hamiltonian* is given by

$$H_0 := \Theta(h) = \int dx \, dy \,\overline{\varphi}(x) h(x; y) \varphi(y) \,. \tag{2.16}$$

The interacting classical Hamiltonian is given by

$$H := H_0 + \mathcal{W} \,. \tag{2.17}$$

The mass is defined as

$$\mathcal{N} := \int dx \, |\varphi(x)|^2 \,. \tag{2.18}$$

At this stage, we have to introduce the cut-off f that appears in (1.6). We now state the precise assumptions on f that we use in the sequel.

Assumption 2.1.4. In the cubic case, we fix $f \in C_c^{\infty}(\mathbb{R})$, which is not identically equal to zero such that $0 \leq f \leq 1$ and

$$\operatorname{supp}(f) \subset [-K, K], \qquad (2.19)$$

for some K > 0.

All of our estimates depend on K in (2.19), but we do not track this dependence explicitly.

We define the *classical state* $\rho^f(\cdot) \equiv \rho(\cdot)$ by

$$\rho(X) := \frac{\int d\mu \, X e^{-\mathcal{W}} f(\mathcal{N})}{\int d\mu \, e^{-\mathcal{W}} f(\mathcal{N})} \equiv \mathbb{E}_{\mathbb{P}^f_{\text{Gibbs}}}(X) \,, \tag{2.20}$$

where X is a random variable. Let the classical partition function $z \equiv z_{\text{Gibbs}}^{f}$ be defined as

$$z := \int d\mu \,\mathrm{e}^{-\mathcal{W}} f(\mathcal{N}) \,. \tag{2.21}$$

Note that both ρ and z are well defined by Lemma 3.1.1 and Corollary 3.1.4 below. We characterise $\rho(\cdot)$ through its moments. Namely, we define the *classical p-particle* correlation function $\gamma_p \equiv \gamma_p^f$, which acts on $\mathfrak{h}^{(p)}$ through its kernel

$$\gamma_p(x_1, \dots, x_p; y_1, \dots, y_p) := \rho(\overline{\varphi}(y_1) \dots \overline{\varphi}(y_p) \varphi(x_1) \dots \varphi(x_p)).$$
(2.22)

For the time-dependent result, we will also need the following notion of *classical evolution*.

Definition 2.1.5. Let $p \in \mathbb{N}^*$ and $\xi \in \mathcal{L}(\mathfrak{h}^{(p)})$ be given. For $t \in \mathbb{R}$, we define the random variable

$$\Psi^{t}\Theta(\xi) := \int dx_{1} \dots dx_{p} \, dy_{1} \dots dy_{p} \, \xi(x_{1}, \dots, x_{p}; y_{1} \dots y_{p}) \\ \times \overline{S_{t}\varphi}(x_{1}) \dots \overline{S_{t}\varphi}(x_{p}) S_{t}\varphi(x_{p}) \dots S_{t}\varphi(y_{p}), \quad (2.23)$$

where S_t is the flow map defined in (2.5). This is well defined since $\varphi \in \mathfrak{h}$ almost surely and S_t is norm preserving on \mathfrak{h} .

2.1.2 Quantum problem

We use the same conventions as in [29, Section 1.4]. For more details and motivation, we refer the reader to the aforementioned work. In the quantum setting, we work on the *bosonic Fock space*, which is defined as

$$\mathcal{F} \equiv \mathcal{F}(\mathfrak{h}) := \bigoplus_{p \in \mathbb{N}} \mathfrak{h}^{(p)}.$$

Let us denote vectors of \mathcal{F} by $\Psi = (\Psi^{(p)})_{p \in \mathbb{N}}$. For $g \in \mathfrak{h}$, let $b^*(g)$ and b(g) denote the bosonic creation and annihilation operators, defined respectively as

$$(b^*(g)\Psi)^{(p)}(x_1,\ldots,x_p) := \frac{1}{\sqrt{p}} \sum_{i=1}^p g(x_i)\Psi^{(p-1)}(x_1,\ldots,x_{i-1},x_{i+1},\ldots,x_p), \quad (2.24)$$

$$(b(g)\Psi)^{(p)}(x_1,\ldots,x_p) := \sqrt{p+1} \int dx \,\overline{g(x)}\Psi^{(p+1)}(x,x_1,\ldots,x_p) \,. \tag{2.25}$$

These are closed, densely-defined operators which are each other's adjoints; see for example [13]. The creation and annihilation operators satisfy the canonical commutation relations, i.e.

$$[b(g_1), b^*(g_2)] = \langle g_1, g_2 \rangle_{\mathfrak{h}}, \quad [b(g_1), b(g_2)] = [b^*(g_1), b^*(g_2)] = 0, \quad (2.26)$$

for all $g_1, g_2 \in \mathfrak{h}$. The computation of (2.26) is included in Appendix B.

We define the rescaled creation and annihilation operators

$$\varphi_{\tau}^*(g) := \tau^{-1/2} b^*(g), \quad \varphi_{\tau}(g) := \tau^{-1/2} b(g), \quad (2.27)$$

for $g \in \mathfrak{h}$. Here, we think of φ_{τ}^* and φ_{τ} as operator valued distributions, and we denote their distribution kernels by $\varphi_{\tau}^*(x)$ and $\varphi_{\tau}(x)$, respectively. Formally, $\varphi_{\tau}^*(x)$ and $\varphi_{\tau}(x)$ correspond to taking $g = \delta_x$ (the Dirac delta function centred at x) in

(2.27). In analogy to (2.12), we call φ_{τ} the quantum field.

As before, let ξ be a closed linear operator on $\mathfrak{h}^{(p)}$. The *lift* of ξ to \mathcal{F} is defined by

$$\Theta_{\tau}(\xi) := \int dx_1 \dots dx_p \, dy_1 \dots dy_p \, \xi(x_1, \dots, x_p; y_1, \dots, y_p) \\ \times \varphi_{\tau}^*(x_1) \dots \varphi_{\tau}^*(x_p) \varphi_{\tau}(y_1) \dots \varphi_{\tau}(y_p) \,. \tag{2.28}$$

For $w \in L^{\infty}(\mathbb{T})$ real-valued and even, we define the quantum interaction as²

$$\mathcal{W}_{\tau} := \frac{1}{2} \Theta_{\tau}(W) = \frac{1}{2} \int dx \, dy \, \varphi_{\tau}^*(x) \varphi_{\tau}^*(y) w(x-y) \varphi_{\tau}(x) \varphi_{\tau}(y) \,. \tag{2.29}$$

Here W is the two particle operator on $\mathfrak{h}^{(2)}$ which acts by multiplication by $w(x_1-x_2)$ for $w \in L^{\infty}$. We define the *free quantum Hamiltonian* as

$$H_{\tau,0} := \Theta_{\tau}(h) = \int dx \, dy \, \varphi_{\tau}^*(x) h(x;y) \varphi_{\tau}(y) \,, \qquad (2.30)$$

where h is as in (2.6). We define the interacting quantum Hamiltonian as

$$H_{\tau} := H_{\tau,0} + \mathcal{W}_{\tau} \,.$$

We also define the rescaled particle number as

$$\mathcal{N}_{\tau} := \int dx \, \varphi_{\tau}^*(x) \varphi_{\tau}(x) \,. \tag{2.31}$$

In Appendix B.1 it is shown that (2.67) acts on the p^{th} sector of Fock space as multiplication by $\frac{p}{\tau}$.

The *(untruncated) grand canonical ensemble* is defined as

$$P_{\tau} := \mathrm{e}^{-H_{\tau}} \tag{2.32}$$

and the *(truncated) quantum state* $\rho_{\tau}^{f}(\cdot) \equiv \rho_{\tau}(\cdot)$ is defined as

$$\rho_{\tau}(\mathcal{A}) := \frac{\operatorname{Tr}(\mathcal{A}P_{\tau}f(\mathcal{N}_{\tau}))}{\operatorname{Tr}(P_{\tau}f(\mathcal{N}_{\tau}))}, \qquad (2.33)$$

where the traces are taken over Fock space. Let the quantum partition function and

²In principle, we could consider w as in Assumption 2.1.1 in the quantum setting at the level of the definition. In practice, we take the interaction potential to be bounded; see Section 2.1.3 below for the precise statements.

the free quantum partition function, $Z_{\tau} \equiv Z_{\tau}^{f}$, $Z_{\tau,0}$ be defined respectively as

$$Z_{\tau} := \operatorname{Tr}(\mathrm{e}^{-H_{\tau}} f(\mathcal{N}_{\tau})), \quad Z_{\tau,0} := \operatorname{Tr}(\mathrm{e}^{-H_{\tau,0}}).$$
 (2.34)

With $Z_{\tau}, Z_{\tau,0}$ as in (2.34), we define the relative quantum partition function $\mathcal{Z}_{\tau} \equiv \mathcal{Z}_{\tau}^{f}$ by

$$\mathcal{Z}_{\tau} := \frac{Z_{\tau}}{Z_{\tau,0}} \,. \tag{2.35}$$

In analogy to (2.22), we characterise the quantum state through its correlation functions. Namely, for $p \in \mathbb{N}^*$, we define the quantum *p*-particle correlation function $\gamma_{\tau,p}^f \equiv \gamma_{\tau,p}$, which acts on $\mathfrak{h}^{(p)}$ through its kernel

$$\gamma_{\tau,p}(x_1,\ldots,x_p;y_1,\ldots,y_p) := \rho_{\tau}(\varphi_{\tau}^*(y_1)\ldots\varphi_{\tau}^*(y_p)\varphi_{\tau}(x_1)\ldots\varphi_{\tau}(x_p)).$$
(2.36)

For the time-dependent problem, we also define the *quantum time evolution* of an operator on Fock space.

Definition 2.1.6. Suppose $\mathcal{A} : \mathcal{F} \to \mathcal{F}$. Define the quantum time evolution of \mathcal{A} as

$$\Psi^t_{\tau} \mathcal{A} := \mathrm{e}^{it\tau H_{\tau}} \mathcal{A} \, \mathrm{e}^{-it\tau H_{\tau}} \, .$$

Throughout the sequel, for a given quantity $Y = \rho, \mathcal{N}, H, \ldots$, we will use the abbreviation $Y_{\#}$ to denote either Y_{τ} or Y.

2.1.3 Main results

We now state our results for the cubic problem. The first result that we prove concerns bounded interaction potentials.

Theorem 2.1.7 (Convergence for $w \in L^{\infty}(\mathbb{T})$). Let $w \in L^{\infty}(\mathbb{T})$ be real-valued and even. Given $p \in \mathbb{N}^*$, we recall the quantities $\gamma_{\tau,p}$ and γ_p defined in (2.36) and (2.22) respectively. We then have

$$\lim_{\tau \to \infty} \|\gamma_{\tau,p} - \gamma_p\|_{\mathfrak{S}^1(\mathfrak{h}^{(p)})} = 0.$$
(2.37)

Moreover, recalling (2.21) and (2.35), we have

$$\lim_{\tau \to \infty} \mathcal{Z}_{\tau} = z \,. \tag{2.38}$$

By applying an approximation argument, we prove results for $w \in L^1(\mathbb{T})$ and $w = -\delta$ as in Assumption 2.1.1. Throughout the sequel, any object with a superscript ε is the corresponding object defined by taking the interaction potential to be w^{ε} , which will be a suitable bounded approximation of w. In what follows, we always assume that all the approximating interaction potentials w^{ε} are *real-valued and even*, without mentioning this explicitly. We can now state the result for $L^1(\mathbb{T})$ interaction potentials.

Theorem 2.1.8 (Convergence for $w \in L^1(\mathbb{T})$). Let w be as in Assumption 2.1.1 (i). Suppose that (w^{ε}) is a sequence of interaction potentials in $L^{\infty}(\mathbb{T})$ such that $w^{\varepsilon} \to w$ in $L^1(\mathbb{T})$. Then there exists a sequence (ε_{τ}) satisfying $\varepsilon_{\tau} \to 0$ as $\tau \to \infty$ such that for any $p \in \mathbb{N}^*$

$$\lim_{\tau \to \infty} \|\gamma_{\tau,p}^{\varepsilon_{\tau}} - \gamma_{p}\|_{\mathfrak{S}^{1}(\mathfrak{h}^{(p)})} = 0$$
(2.39)

and such that

$$\lim_{\tau \to \infty} \mathcal{Z}_{\tau}^{\varepsilon_{\tau}} = z \,. \tag{2.40}$$

Before considering $w = -\delta$ as in Assumption 2.1.1 (ii), we need to define the sequence more w^{ε} precisely. We fix $U : \mathbb{R} \to \mathbb{R}$ to be a continuous even function, with $\operatorname{supp} U \subset \mathbb{T}$ satisfying

$$\int_{\mathbb{R}} dx \, U(x) = \int_{\mathbb{T}} dx \, U(x) = -1 \,. \tag{2.41}$$

For $\varepsilon \in (0, 1)$, we define

$$w^{\varepsilon} := \frac{1}{\varepsilon} U\left(\frac{[x]}{\varepsilon}\right), \qquad (2.42)$$

where [x] is defined to be the unique element in $(x+\mathbb{Z})\cap\mathbb{T}$. In particular, $w^{\varepsilon} \in L^{\infty}(\mathbb{T})$ and w^{ε} converges to $-\delta$ weakly, with respect to continuous functions.

Theorem 2.1.9 (Convergence for $w = -\delta$). With notation as in (2.42), there exists a sequence (ε_{τ}) satisfying $\varepsilon_{\tau} \to 0$ as $\tau \to \infty$ such that for any $p \in \mathbb{N}^*$

$$\lim_{\tau \to \infty} \|\gamma_{\tau,p}^{\varepsilon_{\tau}} - \gamma_p\|_{\mathfrak{S}^1(\mathfrak{h}^{(p)})} = 0 \tag{2.43}$$

and such that

$$\lim_{\tau \to \infty} \mathcal{Z}_{\tau}^{\varepsilon_{\tau}} = z \,. \tag{2.44}$$

Remark 2.1.10. We make the following observations about Theorems 2.1.7, 2.1.8, and 2.1.9.

1. For a pointwise almost everywhere non-negative, bounded, even interaction potential w, Theorem 2.1.7 holds without the need for a cut-off function f.

This is the content of [29, Theorem 1.8]. Moreover, by working with the non-normal ordered quantum interaction \mathcal{W}'_{τ} defined in (3.72), for a bounded, real-valued, even interaction potential w of positive type (i.e. \hat{w} pointwise almost everywhere non-negative), the same proof as [29, Theorem 1.8] again shows that Theorem 2.1.7 holds without the need for a cut-off function f. We include the details of the proof of this claim in Section 3.4.1.

- 2. We conjecture that the results hold for f a characteristic function of an interval. The method that we apply in Lemma 3.2.10 of Section 3.2.5 requires suitable smoothness assumptions on f. This is a technical assumption.
- 3. For an individual $w \in L^{\infty}$, Theorem 2.1.7 holds with a cut-off function of the form $f(x) = e^{-cx^2}$, for c > 0 sufficiently large depending on $||w||_{L^{\infty}}$. This is also proved by working with a non-normal ordered quantum interaction. The details are given in Section 3.4.2. We note this c cannot be chosen uniformly in the L^{∞} norm of the interaction potential. So we cannot treat the unbounded interactions as in Theorems 2.1.8 and 2.1.9 using this kind of truncation.
- 4. One could consider the questions from Theorems 2.1.7 and 2.1.11 in the nonperiodic setting when the spatial domain is \mathbb{R} for the one-body Hamiltonian $h = -\Delta + \kappa + v$, where $v : \mathbb{R} \to [0, \infty)$ is a positive one-body potential such that h has compact resolvent and $\operatorname{Tr} h^{-1} < \infty$ holds (as in (2.10)). The analysis that we present in the periodic setting would carry through to this case, provided that we know that the time evolution S_t given in (2.5) is welldefined on the support of the Gibbs measure. We do not address this question further in the thesis.
- 5. By following the duality arguments in [29, Section 3.3], we can get the equivalents of equations (2.37), (2.39), and (2.43) in terms of ρ_{τ} and ρ . For more details when $w \in L^{\infty}$, see Corollary 3.2.13, Lemma 3.3.1, and Lemma 3.3.2 below. For the time-independent problem, we state the convergence as above in the trace class. For the time-dependent problem, we need to use the alternative formulation, which can be seen as a generalisation of the time-independent analysis. For more details, see Remark 2.1.14 below.
- 6. Our method works for more general interaction potentials. In particular, we can consider linear combinations of interaction potentials as in Assumption 2.1.1 (i) and (ii) with the same arguments.

Recalling the quantities ρ_{τ} and ρ defined as in (2.33) and (2.20) respectively, we now state the time-dependent results for the cubic problem. **Theorem 2.1.11** (Convergence for $w \in L^{\infty}(\mathbb{T})$). Let w be as in Theorem 2.1.7. Given $m \in \mathbb{N}^*$, $p_i \in \mathbb{N}^*$, $\xi^i \in \mathcal{L}(\mathfrak{h}^{(p_i)})$, and $t_i \in \mathbb{R}$, we have

$$\lim_{\tau \to \infty} \rho_{\tau} \left(\Psi_{\tau}^{t_1} \Theta_{\tau}(\xi^1) \dots \Psi_{\tau}^{t_m} \Theta_{\tau}(\xi^m) \right) = \rho \left(\Psi^{t_1} \Theta(\xi^1) \dots \Psi^{t_m} \Theta(\xi^m) \right) \,.$$

Theorem 2.1.12 (Convergence for $w \in L^1(\mathbb{T})$). Let w, w^{ε} be as in the assumptions of Theorem 2.1.8. Then, there exists a sequence (ε_{τ}) satisfying $\varepsilon_{\tau} \to 0$ as $\tau \to \infty$ such that, given $m \in \mathbb{N}^*$, $p_i \in \mathbb{N}^*$, $\xi^i \in \mathcal{L}(\mathfrak{h}^{(p_i)})$, and $t_i \in \mathbb{R}$, we have

$$\lim_{\tau \to \infty} \rho_{\tau}^{\varepsilon_{\tau}} \left(\Psi_{\tau}^{t_1} \Theta_{\tau}(\xi^1) \dots \Psi_{\tau}^{t_m} \Theta_{\tau}(\xi^m) \right) = \rho \left(\Psi^{t_1} \Theta(\xi^1) \dots \Psi^{t_m} \Theta(\xi^m) \right).$$

Theorem 2.1.13 (Convergence for $w = -\delta$). Let w, w^{ε} be as in the assumptions of Theorem 2.1.9. Then, there exists a sequence (ε_{τ}) satisfying $\varepsilon_{\tau} \to 0$ as $\tau \to \infty$ such that, given $m \in \mathbb{N}^*$, $p_i \in \mathbb{N}^*$, $\xi^i \in \mathcal{L}(\mathfrak{h}^{(p_i)})$, and $t_i \in \mathbb{R}$, we have

$$\lim_{\tau \to \infty} \rho_{\tau}^{\varepsilon_{\tau}} \left(\Psi_{\tau}^{t_1} \Theta_{\tau}(\xi^1) \dots \Psi_{\tau}^{t_m} \Theta_{\tau}(\xi^m) \right) = \rho \left(\Psi^{t_1} \Theta(\xi^1) \dots \Psi^{t_m} \Theta(\xi^m) \right) \,.$$

Remark 2.1.14. Theorems 2.1.11–2.1.13 can indeed be seen as generalisations of the results given in Theorems 2.1.7–2.1.9 respectively (the latter of which correspond to setting m = 1 and $t_1 = 0$). Namely, we use Remark 2.1.10 (3) above and note that the proofs show that the convergence is uniform in $\|\xi^1\| \leq 1$.

2.2 Quintic problem

2.2.1 Classical problem

The primary difference between the cubic and quintic classical cases is that we consider an equation with a nonlinearity which does not have a convolution structure. We begin by stating our assumptions on the interaction potential in the quintic model.

Assumption 2.2.1. Let $w : \Lambda \to \mathbb{R}$ be even and such that $w \in L^{\infty}(\Lambda)$.

Some of our analysis applies to more singular interaction potentials.

Assumption 2.2.2. Let $w : \Lambda \to \mathbb{R}$ be even and such that $w \in L^{\frac{3}{2}}(\Lambda)$.

In Assumptions 2.2.1–2.2.2 above, we make no condition on the (pointwise) sign of w or \hat{w} .

Convention: When working in the classical setting, we consider w as in Assumption 2.2.2. When working in the quantum setting, we will consider w as in Assumption 2.2.1. See Section 2.2.2 for more details.

For w as in Assumption 2.2.2, we study the following form of the *nonlinear* Schrödinger equation (NLS).

$$i\partial_t u + (\Delta - \kappa)u = \int dy \, dz \, w(x - y) \, w(y - z) \, w(z - x) \, |u(y)|^2 \, |u(z)|^2 \, u(x) \,. \quad (2.45)$$

We refer to (2.45) as the quintic Hartree equation or nonlocal quintic NLS. As is shown in Section 4.1 below, Assumption 2.2.2 is the natural setting for the interaction potential when studying (2.45); see Remark 2.2.8 below.

On the space of fields $u : \Lambda \to \mathbb{C}$, we consider the Poisson structure where the Poisson bracket is given by

$$\{u(x), \bar{u}(y)\} = i\delta(x-y), \qquad \{u(x), u(y)\} = \{\bar{u}(x), \bar{u}(y)\} = 0.$$
(2.46)

Lemma 2.2.3. With Poisson structure given by (2.46), we have that (2.45) corresponds to the Hamiltonian equation of motion associated with Hamiltonian

$$H(u) = \int dx \left(|\nabla u(x)|^2 + \kappa |u(x)|^2 \right) + \frac{1}{3} \int dx \, dy \, dz \, w(x-y) \, w(y-z) \, w(z-x) \, |u(x)|^2 \, |u(y)|^2 \, |u(z)|^2 \,.$$
(2.47)

Proof. By a direct calculation using (2.46), we obtain that

$$\{H, u\}(x) = i(\Delta u(x) - \kappa u(x)) - i \int dy \, dz \, w(x - y) \, w(y - z) \, w(z - x) \, |u(y)|^2 \, |u(z)|^2 \, u(x) \,.$$
(2.48)

Using the framework of [8], we show in Section 4.1 that (2.45) is locally wellwith w as in Assumption 2.2.2 and for initial data in $H^s(\Lambda)$ with s > 0; for

posed with w as in Assumption 2.2.2 and for initial data in $H^s(\Lambda)$ with s > 0; for details see Proposition 4.1.1 below. For the Gibbs measure given by

$$d\mathbb{P}^{f}_{\text{Gibbs}}(u) := \frac{1}{z^{f}_{\text{Gibbs}}} e^{-H(u)} f(\|u\|^{2}_{L^{2}}) du, \qquad (2.49)$$

where $f \in C_0^{\infty}(\mathbb{R})$ is a suitable cut-off function and

$$z_{\text{Gibbs}}^f = \int du \, \mathrm{e}^{-H(u)} \, f(\|u\|_{L^2}^2),$$
we show that (2.45) is globally well-posed for initial conditions in the support of (2.49). For a precise statement, see Proposition 4.1.2 below. In particular, there is a well-defined solution map S_t that maps any initial condition u_0 in the support of (2.49) to the solution of (2.45) at time t given by

$$u(\cdot) \equiv u(\cdot, t) := S_t(u_0).$$
 (2.50)

The map S_t preserves regularity in the sense that for $s \in (0, \frac{1}{2})$ and $u_0 \in H^s$, we have $S_t(u_0) \in H^s$.

The Wiener measure in the quintic case is defined in the same way as the cubic case; see (2.11) for more details. Similarly, the classical free field is given by (2.12), and the Wiener measure still satisfies the Wick theorem given in Proposition 2.1.2. The mass $\mathcal{N} \equiv \mathcal{N}^{\omega} = \|\varphi\|_{\mathfrak{h}}^2$ is also the same as the cubic case.

With w as in Assumption 2.2.2, the *classical interaction* $\mathcal{W} \equiv \mathcal{W}^{\omega}$ is given by

$$\mathcal{W} := \frac{1}{3} \int dx \, dy \, dz \, w(x-y) \, w(y-z) \, w(z-x) \, |\varphi(x)|^2 \, |\varphi(y)|^2 \, |\varphi(z)|^2 \,. \tag{2.51}$$

Since $\varphi \in H^{\frac{1}{3}}$ μ -almost surely, by Lemma 4.1.3 below, it follows that \mathcal{W} is finite μ -almost surely. We also define the *classical free Hamiltonian* $H_0 \equiv H_0^{\omega}$ as

$$H_0 := \int dx \, dy \,\overline{\varphi}(x) \, h(x; y) \,\varphi(y) \,, \qquad (2.52)$$

where h(x; y) is the kernel corresponding to (2.6), which is the same as in the cubic case. The *classical interacting Hamiltonian* $H \equiv H^{\omega}$ is given by

$$H := H_0 + \mathcal{W} \,. \tag{2.53}$$

Assumption 2.2.4. Whenever working in the quintic case, we fix $f \in C_0^{\infty}(\mathbb{R})$ a cut-off function, which is not identically zero, satisfying $0 \le f \le 1$, and

$$f(s) = 0 \text{ for } s > K,$$
 (2.54)

where K > 0 is a sufficiently small positive constant.

Remark 2.2.5. The choice of K is dictated by Proposition 4.1.2 (i) below. Throughout, our estimates will depend on the K in (2.54), but we will not keep explicit track of this dependence.

With μ as in (2.11), \mathcal{N} as in (2.18), \mathcal{W} as in (2.51), and f as in Assumption

2.2.4, we now define the classical Gibbs measure

$$\mathbb{P}^{f}_{\text{Gibbs}} := \frac{1}{z^{f}_{\text{Gibbs}}} e^{-\mathcal{W}} f(\mathcal{N}) \, \mu \,, \qquad (2.55)$$

where the classical partition function $z \equiv z_{\text{Gibbs}}^{f}$ is the normalisation constant

$$z := \int d\mu \,\mathrm{e}^{-\mathcal{W}} f(\mathcal{N}) \,. \tag{2.56}$$

We note that the probability measure $\mathbb{P}^{f}_{\text{Gibbs}}$ in (2.55) and the classical partition function (2.56) are well-defined for sufficiently small K in (2.54) by Proposition 4.1.2 (i) below. In particular, this gives us a rigorous construction of (2.49) above.

The remainder of the classical model is defined analogously to the cubic case, though we include it for completeness.

For a random variable $X \equiv X^{\omega}$, the *classical state* $\rho \equiv \rho^f$ is given by

$$\rho(X) := \mathbb{E}_{\mathbb{P}^{f}_{\text{Gibbs}}}(X) = \frac{\int d\mu \, X e^{-\mathcal{W}} f(\mathcal{N})}{\int d\mu \, e^{-\mathcal{W}} f(\mathcal{N})} \,.$$
(2.57)

We define the classical *p*-particle correlation function $\gamma_p \equiv \gamma_p^f$ as the operator on $\mathfrak{h}^{(p)}$ with kernel given by

$$\gamma_p(x_1,\ldots,x_p;y_1,\ldots,y_p) = \rho(\bar{\varphi}(y_1)\ldots\bar{\varphi}(y_p)\varphi(x_1)\ldots\varphi(x_p)).$$
(2.58)

For a closed densely-defined linear operator ξ on $\mathfrak{h}^{(p)}$, we will consider the random variable $\Theta(\xi) \equiv \Theta^{\omega}(\xi)$ defined as

$$\Theta(\xi) := \int dx_1 \dots dx_p \, dy_1 \dots dy_p \, \xi(x_1, \dots, x_p; y_1, \dots, y_p) \\ \times \overline{\varphi}(x_1) \dots \overline{\varphi}(x_p) \varphi(y_1) \dots \varphi(y_p) \,. \tag{2.59}$$

Let us note that

$$H = \Theta(h) + \frac{1}{3}\Theta(W), \qquad (2.60)$$

where W is the operator corresponding to p = 3 and operator kernel

$$W(x_1, x_2, x_3; y_1, y_2, y_3)$$

:= $w(x_1 - x_2) w(x_2 - x_3) w(x_3 - x_1) \delta(x_1 - y_1) \delta(x_2 - y_2) \delta(x_3 - y_3)$. (2.61)

In particular, W acts as multiplication by $w(x_1 - x_2)w(x_2 - x_3)w(x_3 - x_1)$. Note

that W is a linear operator on $\mathfrak{h}^{(3)}$ since by Assumption 2.2.2 we have that w is even. It is densely defined³. As was noted earlier, $\Theta(W) = 3W$ is finite μ -almost surely.

With S_t defined as in (2.50) and ξ a closed densely-defined linear operator on $\mathfrak{h}^{(p)}$, we define for $t \in \mathbb{R}$ the random variable

$$\Psi^t \Theta(\xi) := \int dx_1 \dots dx_p \, dy_1 \dots dy_p \, \xi(x_1, \dots, x_p; y_1, \dots, y_p) \\ \times \overline{S_t \varphi}(x_1) \dots \overline{S_t \varphi}(x_p) S_t \varphi(y_1) \dots S_t \varphi(y_p) \,, \quad (2.62)$$

which corresponds to the time evolution of (2.59).

2.2.2 Quantum problem

The major difference between the cubic and quintic problems is we that we now consider the three-body interaction given in (2.63). When working in the quantum setting, we consider w as in Assumption 2.2.1 (i.e. $w \in L^{\infty}$ and even). This assumption is needed for technical reasons, as will be clear from the analysis in Section 4.2 below.

For w as in Assumption 2.2.1, the quantum interaction is defined as

$$\mathcal{W}_{\tau} := \frac{1}{3} \int dx \, dy \, dz \, w(x-y) \, w(y-z) \, w(z-x) \, \varphi_{\tau}^*(x) \varphi_{\tau}^*(y) \varphi_{\tau}^*(z) \varphi_{\tau}(x) \varphi_{\tau}(y) \varphi_{\tau}(z) \,.$$

$$(2.63)$$

For h as in (2.6) the quantum free Hamiltonian is defined as

$$H_{\tau,0} := \int dx \, dy \, \varphi_{\tau}^*(x) \, h(x,y) \, \varphi_{\tau}(y) \,. \tag{2.64}$$

Then the quantum interacting Hamiltonian is given by

$$H_{\tau} := H_{\tau,0} + \mathcal{W}_{\tau} \,. \tag{2.65}$$

Using (2.27) and (2.63)-(2.65), we have

$$H_{\tau} = \bigoplus_{n=0}^{\infty} H_{\tau}^{(n)} \,,$$

³It would be bounded if we were considering w as in Assumption 2.2.1.

where

$$H_{\tau}^{(n)} = \frac{1}{\tau} \sum_{i=1}^{n} (-\Delta_i + \kappa) + \frac{1}{3\tau^3} \sum_{\substack{i,j,k\\i \neq j \neq k \neq i}}^{n} w(x_i - x_j) w(x_j - x_k) w(x_k - x_i). \quad (2.66)$$

Note that $H_{\tau}^{(n)} = \frac{1}{\tau} H^{(n)}$ for $H^{(n)}$ as in (1.10) with $\lambda = \frac{1}{\tau^2}$. The rest of the quantum setting is defined analogously to the cubic case. For self-containedness, we include the definitions below.

We define the *rescaled particle number* as

$$\mathcal{N}_{\tau} := \int dx \, \varphi_{\tau}^*(x) \, \varphi_{\tau}(x) \,. \tag{2.67}$$

The (untruncated and unnormalised) grand canonical ensemble is given by

$$P_{\tau} := \mathrm{e}^{-H_{\tau}} = \bigoplus_{n=0}^{\infty} \mathrm{e}^{-H_{\tau}^{(n)}},$$

with $H_{\tau}^{(n)}$ as in (2.66). For a closed operator $\mathcal{A} : \mathcal{F} \to \mathcal{F}$, the quantum state $\rho_{\tau} \equiv \rho_{\tau}^{f}$ is defined as

$$\rho_{\tau}(\mathcal{A}) := \frac{\operatorname{Tr}_{\mathcal{F}} \left(\mathcal{A} P_{\tau} f(\mathcal{N}_{\tau}) \right)}{\operatorname{Tr}_{\mathcal{F}} \left(P_{\tau} f(\mathcal{N}_{\tau}) \right)} \,. \tag{2.68}$$

We define the quantum partition function and quantum free partition function $Z_{\tau} \equiv Z_{\tau}^{f}$ and $Z_{\tau,0}$ respectively as

$$Z_{\tau} := \operatorname{Tr}\left(P_{\tau}f(\mathcal{N}_{\tau})\right), \qquad Z_{\tau,0} := \operatorname{Tr}\left(e^{-H_{\tau,0}}\right), \qquad (2.69)$$

and the quantum relative partition function $\mathcal{Z}_{\tau} \equiv \mathcal{Z}_{\tau}^{f}$ as

$$\mathcal{Z}_{\tau} := \frac{Z_{\tau}}{Z_{\tau,0}} \,. \tag{2.70}$$

For $p \in \mathbb{N}^*$ we define the *p*-particle quantum correlation function $\gamma_{\tau,p} \equiv \gamma_{\tau,p}^f$ as operator which acts on $\mathfrak{h}^{(p)}$, with kernel given by

$$\gamma_{\tau,p}(x_1,\ldots,x_p,y_1,\ldots,y_p) := \rho_\tau(\varphi_\tau^*(y_1)\ldots\varphi_\tau^*(y_p)\varphi_\tau(x_1)\ldots\varphi_\tau(x_p)).$$
(2.71)

Note that (2.71) is a quantum analogue of (2.58).

By analogy with (2.59), for a closed linear operator $\xi \in \mathcal{L}(\mathfrak{h}^{(p)})$, we define

the lift of ξ to Fock space as

$$\Theta_{\tau}(\xi) := \int dx_1 \dots dx_p \, dy_1 \dots dy_p \, \xi(x_1, \dots, x_p; y_1, \dots, y_p) \\ \times \varphi_{\tau}^*(x_1) \dots \varphi_{\tau}^*(x_p) \varphi_{\tau}(y_1) \dots \varphi_{\tau}(y_p) \,. \tag{2.72}$$

Analogously to (2.60), in the quantum setting we have

$$H_{\tau} = \Theta_{\tau}(h) + \frac{1}{3} \Theta_{\tau}(W) \,,$$

where W is the 3-body operator with kernel (2.61). We note that, by Assumption 2.2.1, $\Theta_{\tau}(W)$ is a bounded operator on Fock space. For an operator $\mathcal{A}: \mathcal{F} \to \mathcal{F}$, we define the quantum time evolution

$$\Psi_{\tau}^{t}\mathcal{A} := e^{it\tau H_{\tau}}\mathcal{A} e^{-it\tau H_{\tau}} . \qquad (2.73)$$

2.2.3 Main results

We now state our main results in the quintic case. Theorems 2.2.6 and 2.2.7 are our results in the time-independent case, and Theorems 2.2.11 and 2.2.12 the corresponding time-dependent results. Throughout, we fix a cut-off function $f \in C_0^{\infty}(\mathbb{R})$ as in Assumption 2.2.4 above.

We first analyse bounded interaction potentials (as in Assumption 2.2.1).

Theorem 2.2.6 (Convergence for bounded interaction potentials). Let w be as in Assumption 2.2.1. Let $p \in \mathbb{N}^*$ be given. Consider γ_p and $\gamma_{\tau,p}$ defined as in (2.58) and (2.71) respectively. Then, we have

$$\lim_{\tau \to \infty} \|\gamma_{\tau,p} - \gamma_p\|_{\mathfrak{S}^1(\mathfrak{h}^{(p)})} = 0.$$
(2.74)

Moreover, for z and Z_{τ} defined as in (2.56) and (2.70) respectively, we have

$$\lim_{\tau \to \infty} \mathcal{Z}_{\tau} = z \,. \tag{2.75}$$

To obtain a result for w as in Assumption 2.2.2, we use an approximation argument. For a suitable approximation w^{ε} to w, as defined below, any object with a superscript ε will be the corresponding object defined using w^{ε} instead of w.

Theorem 2.2.7 (Convergence for $L^{\frac{3}{2}}$ interaction potentials). Let w be as in Assumption 2.2.2. Suppose that w^{ε} is a sequence of interaction potentials as in Assumption 2.2.1 converging to w in $L^{\frac{3}{2}}$. With objects defined analogously as for

Theorem 2.2.6, there exists a sequence $\varepsilon_{\tau} \to 0$ as $\tau \to \infty$ such that for any $p \in \mathbb{N}^*$, we have

$$\lim_{\tau \to \infty} \|\gamma_{\tau,p}^{\varepsilon_{\tau}} - \gamma_p\|_{\mathfrak{S}^1(\mathfrak{h}^{(p)})} = 0.$$
(2.76)

and such that

$$\lim_{\tau \to \infty} \mathcal{Z}_{\tau}^{\varepsilon_{\tau}} = z \,. \tag{2.77}$$

Before proceeding with the time-dependent results, let us make a few remarks.

Remark 2.2.8. As in the cubic problem, in the classical setting, our method relies on a good understanding of the nonlocal problem with bounded interaction potential (as in Theorem 2.2.6 above). We are then able to upgrade this to a result for the nonlocal problem with unbounded interaction potential (as in Theorem 2.2.7) above by a suitable approximation argument. Unlike in the cubic Hartree equation, the nonlinearity in (2.45) is not given by a convolution, and hence considering $w \in L^1(\Lambda)$ in Theorem 2.2.7 does not seem to be possible, as the nonlinear term in (2.45) becomes too singular. Note that formally taking $w = -\delta$, (2.45) yields the focusing quintic NLS. In this case, the nonlinearity simplifies due to the presence of delta functions and one can establish a suitable well-posedness theory, as in [8]. Due to the aforementioned lack of a convolution structure, we cannot recover the necessary estimates for $w \in L^1(\Lambda)$ from those given in the analysis of the local problem [8], as was possible in the cubic case. Instead, we work with w as in Assumption 2.2.2, which, in light of Section 4.1 below, we conjecture to be optimal.

Remark 2.2.9. We emphasise that the calculation (2.48) above does not rely on the evenness of w. This is in contrast to the cubic Hartree equation

$$\mathrm{i}\partial_t u + (\Delta - \kappa)u = \int dy \, w(x - y) \, |u(y)|^2 \, u(x) \,,$$

which is a Hamiltonian equation of motion associated with Hamiltonian

$$H(u) = \int dx \, \left(|\nabla u(x)|^2 + \kappa |u(x)|^2 \right) \, + \frac{1}{2} \int dx \, dy \, |u(x)|^2 \, w(x-y) \, |u(y)|^2 \, ,$$

and Poisson structure is given by (2.46) if and only if w is even. In fact, for the entire analysis of the classical problem in Section 4.1 below, we can omit the assumption that w is even in Assumption 2.2.2. However, the operator W with kernel (2.61) given above, acts (densely) on $\mathfrak{h}^{(3)}$ when w is even. Moreover, the operator $\Theta_{\tau}(W)$ is a well-defined operator on Fock space only when w is even. This is all necessary to study the quantum problem. Physically, when considering the *n*-body Hamiltonian given in (1.10), it is realistic to assume that w is even (in addition to being bounded) in the sense that the interaction between particle x_i and x_j is the same as the interaction between particle x_j and x_i ; see (1.10) above. Note that the approximation in Theorem 2.2.7 above is possible only when w is even. Hence, we will always consider w even in the analysis.

Remark 2.2.10. In Section 3.4.2, we prove that one can obtain an analogue of Theorem 2.1.7 with a cut-off function of the form of $f(x) = e^{-c|x|^2}$. This is not possible in the quintic case because there is no known corresponding analogue of Theorem 2.2.6 for non-negative interaction potentials without a cut-off function.

We now state our time-dependent results. Let us recall the definitions of ρ and ρ_{τ} in (2.57) and (2.68) respectively. Furthermore, we recall the definitions of $\Psi^t \Theta(\xi)$ and $\Psi^t_{\tau} \Theta_{\tau}(\xi)$ in (2.62) and (2.72)–(2.73) respectively. We first state the result for bounded interaction potentials.

Theorem 2.2.11 (Convergence for bounded potentials). Let w be as in Assumption 2.2.1. Let $m \in \mathbb{N}^*$, $p_1, \ldots, p_m \in \mathbb{N}^*$, $\xi_1 \in \mathcal{L}(\mathfrak{h}^{(p_1)}), \ldots, \xi_m \in \mathcal{L}(\mathfrak{h}^{(p_m)})$, and $t_1, \ldots, t_m \in \mathbb{R}$ be given. Then

$$\lim_{\tau \to \infty} \rho_{\tau}(\Psi_{\tau}^{t_1} \Theta_{\tau}(\xi_1) \dots \Psi_{\tau}^{t_m} \Theta_{\tau}(\xi_m)) = \rho(\Psi^{t_1} \Theta(\xi_1) \dots \Psi^{t_m} \Theta(\xi_m)) \,.$$

As in the time-independent problem, we can use an approximation result to prove a result for w as in Assumption 2.2.2.

Theorem 2.2.12 (Convergence for $L^{\frac{3}{2}}$ potentials). Let w be as in Assumption 2.2.2. Let w^{ε} be defined as in Theorem 2.2.7 above. Then there is a sequence $\varepsilon_{\tau} \to 0$ as $\tau \to \infty$ such that, for all $m \in \mathbb{N}^*$, $p_1, \ldots, p_m \in \mathbb{N}^*$, $\xi_1 \in \mathcal{L}(\mathfrak{h}^{(p_1)}), \ldots, \xi_m \in \mathcal{L}(\mathfrak{h}^{(p_m)})$, and $t_1, \ldots, t_m \in \mathbb{R}$, we have

$$\lim_{\tau \to \infty} \rho_{\tau}^{\varepsilon_{\tau}} (\Psi_{\tau}^{t_1} \Theta_{\tau}(\xi_1) \dots \Psi_{\tau}^{t_m} \Theta_{\tau}(\xi_m)) = \rho(\Psi^{t_1} \Theta(\xi_1) \dots \Psi^{t_m} \Theta(\xi_m)) \,.$$

We make the following remarks about the time-dependent results given in Theorems 2.2.11 and 2.2.12 above.

Remark 2.2.13. Theorems 2.2.11–2.2.12 give the first microscopic derivation of time-dependent correlation functions for a quintic NLS. From the PDE point of view, the study of time-dependent correlation functions is more relevant for quintic nonlinearities than for cubic ones. Namely, in the latter case, one can study the global well-posedness theory in \mathfrak{h} without using Gibbs measures [8]. When studying

the quintic problem, local well-posedness is known in $H^s(\mathbb{T})$ only for s > 0. Therefore, one needs to use the invariance of the Gibbs measure as the substitute for a conservation law that allows us to obtain (almost sure) global solutions [9].

Remark 2.2.14. By arguing as in [30, Remark 1.4], we can also recover the invariance of the Gibbs measure (2.55) for (2.45) from Theorem 2.2.12.

Remark 2.2.15. We note that Theorems 2.2.11 and 2.2.12 are both generalisations of Theorems 2.2.6 and 2.2.7 respectively. This can be seen by taking m = 1 and $t_1 = 0$, and arguing by duality as in Remarks 2.1.10 (5) and 2.1.14 above. Alternatively, see (4.109)–(4.111) below.

2.3 Previously known many body results

The results in Sections 2.1.3 and Section 2.2.3 are called microscopic derivations of Gibbs measures or time-dependent correlations from many-body quantum mechanics. These can be interpreted as the high-density limit where the mass or temperature of the system tends to infinity. One can also consider the parameter $\frac{1}{\tau}$ to be a semiclassical or mean field parameter, for a more detailed explanation of this interpretation, see [29, Section 1.1]. In the defocusing case, with f = 1, the first results of this kind were obtained by Lewin, Nam, and Rougerie in [49]. Here they considered the one dimensional problem with a positive translation-invariant interaction, which does not require any Wick ordering. This was based on the Gibbs variational principle and the quantum de Finetti theorem. These methods were used by the same authors to extend their results to one dimensional harmonic traps in [50].

In [29], using a perturbative series expansion of the quantum and classical states and Borel resummation, Fröhlich, Knowles, Schlein, and Sohinger gave an alternative proof of the one-dimensional result obtained in [49]. They also gave a proof for appropriately Wick-ordered Gibbs measures obtained from translation-invariant interaction potentials in two and three dimensions for a suitable modification of the grand canonical ensemble. The results in [29] in two and three dimensions were originally stated for $w \in L^{\infty}$, and were extended to optimal $w \in L^q$ using the methods of [29] by Sohinger in [80]. The optimal range of these q were originally observed in [11].

The full result for two and three dimensions was later shown simultaneously and independently using different methods by Lewin, Nam, and Rougerie in [53] and Fröhlich, Knowles, Schlein, and Sohinger in [31]. The result for two dimensions in [53] was previously announced in [51]. The method in [51,53] is a highly non-trivial extension of the one used in [49]. The proof in [31] is based on a functional integral representation. In [34], the two dimensional result in [31] is proved for the Φ_2^4 theory, which corresponds to taking an interaction potential $w = \delta$ in the cubic case.

The results from [29] were also used to study the time-dependent correlations in one dimension in [30]. Related problems for the lattice were studied in [33,45,71], and further details can be found in the expository works [32,52].

We emphasise that all of the results mentioned here are for the cubic defocusing case, and that ours are the first known results in the focusing case for the two body interaction, and the first known result for the three body interaction. Our methods rely crucially on the cut-off function f, which is a natural assumption when working with focusing potentials, see [9, 20, 47, 54, 65, 68, 87]. One would also expect the results to hold for defocusing three-body interactions without cut-off – we conjecture this is true, but do not address this problem in the thesis.

For works concerning the three-body interaction, we direct the reader to [21, 22,48,61–63,88]. We emphasise that the convention for the three-body interaction in (2.66) differs from the aforementioned works, since we require our effective equation (2.45) to be Hamiltonian; see Lemma 2.2.3 above. The effective evolution equations corresponding to the three-body problems in [21,88] are the local NLS, and the ones in [22,48] are nonlocal but have a different form to (2.45). We do not study the former nonlocal models in this thesis.

2.4 Outline of the proof

2.4.1 Cubic problem

We first analyse the time-independent problem for bounded interaction potentials $w \in L^{\infty}$, so the case of Theorem 2.1.7. The starting point is the perturbative expansion of the $e^{-H_{\#}}$, similarly to [29, Section 2.2] for the quantum setting and [29, Section 3.2] for the classical setting. Due to the presence of the truncation $f(\mathcal{N}_{\#})$, the series that result have infinite radius of convergence; see Propositions 3.2.4 and 3.2.7 below. The analyticity of the expansions means we avoid needing the Borel summation techniques used in [29].

To analyse the remainder term in the quantum setting, we apply the Feynman-Kac formula and use the support properties of the cut-off function from Assumption 2.1.4. This analysis is possible since we do not Wick order the interaction in one dimension; see Lemma 3.2.3. Similarly, the truncation is crucial to the analysis of the remainder term in the classical case; see Lemma 3.2.6.

When proving the convergence of the explicit terms for the obtained series, we use the Helffer-Sjöstrand formula from complex analysis to perform an expansion on the truncation $f(\mathcal{N}_{\#})$. This is the only stage where the technical assumption of the smoothness of the truncation is used. This reduces the analysis to the study of the problem with a shifted chemical potential, but with no truncation. It is important that when we no longer have the control from the truncation, the analysis does not depend on the sign of the interaction. The details of this step is found in Lemma 3.2.8 and Lemma 3.2.10.

The proofs of Theorems 2.1.8 and 2.1.9 are based on an application of Theorem 2.1.7 and a diagonal argument. Here we crucially use [9, Lemma 3.10], which we recall in Lemma 3.1.1 below. When working with L^1 interaction potentials, we apply the local version of this result, see Corollary 3.1.4. For full details, see the proof of Lemma 3.3.1, and in particular (3.62)–(3.63).

To treat the time-dependent case for bounded interaction potentials, we apply a Schwinger-Dyson expansion in both the quantum and classical cases, similarly to [30, Sections 3.2-3.3]. Precise statements can be found in Lemmas 3.5.1–3.5.2. This expansion allows us to deduce Theorem 2.1.11 from Theorem 2.1.7. Crucially, since we have the truncation $f(\mathcal{N}_{\tau})$, we do not have to consider the large particle regime as in [30, Section 4], which requires the non-negativity assumption of the interaction potential.

Finally, Theorems 2.1.12 and 2.1.13 are deduced from Theorem 2.1.11 by applying an approximation argument. More precisely, we estimate the flow map of the NLS with an interaction potential w with that of an interaction $w^{\varepsilon} \to w$. For precise statements, see Lemma 3.5.4 when $w \in L^1$ and Lemma 3.5.5 when $w = -\delta$. These results are proved within the framework of $X^{s,b}$ spaces.

2.4.2 Quintic problem

The primary difference in the quintic case is the analysis of the classical equation. We begin by proving the local well-posedness, exsitence of the Gibbs measure, and almost sure global well-posedness of (2.45), see Propositions 4.1.1 and 4.1.2 below for precise statements. The well-posedness results and construction of the Gibbs measure are analogous to the arguments in [8,9]. These results are proved using the multilinear estimates in Lemmas 4.1.3 and 4.1.4. The main difficulty in the analysis compared to [8,9] is that the nonlinearity in (2.45) no longer has a convolution structure.

On the many body side, the argument is analogous to the cubic case, requiring a perturbative expansion of the quantum and classical states. These are analysed using the same methods, meaning that the cut-off in $f(\mathcal{N}_{\tau})$ is again crucial in bounding both the explicit and remainder terms in the quantum and classical expansions. For precise statements, see Section 4.2.2. The same complex analysis argument from the cubic case implies the convergence of the quantum explicit terms to the classical explicit terms, see the proof of Proposition 4.2.13.

To prove the convergence of the power series, we require bounds and convergence of the untruncated terms, defined in (4.113). To prove this, we use a diagrammatic representation of the terms, similar to that of [29, Sections 2 and 4], but adapted to the normal ordered quintic interaction. We then consider interaction potentials in $L^{\frac{3}{2}}$ using a diagonilisation argument similar to the cubic case for unbounded interaction potentials.

For bounded interaction potentials, we treat bounded potentials using a Schwinger-Dyson expansion analogous to the cubic case, but proved for the quintic interaction. Finally, to prove Theorem 2.2.12, we need to prove an approximation result analogous to Lemma 3.5.4. To do this we need to explicitly use the almost sure global well-posedness of (2.45) and a suitable approximation lemma. See Lemma 4.4.4 for precise details.

Chapter 3

Microscopic Derivation of the Gibbs Measure for the Cubic NLS

In Section 3.1 we recall some results we will use in the chapters on the cubic and quintic NLS. Section 3.2 is an analysis of the time-independent case for bounded potentials, and Section 3.3 deals with unbounded potentials in the time-independent case. In Section 3.4, we deal with the cases of different cut-off functions, as mentioned in Remark 2.1.10 (1) and (3). Finally Section 3.5 deals with bounded and unbounded interaction potentials in the time-dependent case.

3.1 Preliminary results and basic estimates

3.1.1 Preliminary results

We recall the notation introduced in Section 1.6. We also note the following auxiliary results, which we will use throughout the thesis and state some basic estimates which will be used throughout the rest of the cubic case and the rest of thesis.

Gibbs measures for the focusing local NLS

When analysing Gibbs measures for the focusing cubic NLS with $w \in L^{\infty}(\mathbb{T})$, it is straightforward to make rigorous sense of (1.6) due to the presence of the truncation as in Assumption 2.1.4; see Lemma 3.1.6 (1) below.

For unbounded potentials, we will need to make use of the following result of Bourgain, found in [9, Lemma 3.10], whose proof is recalled in Appendix A. **Lemma 3.1.1.** Let $(\mathbb{C}^{\mathbb{N}}, \mathcal{G}, \mu)$ be the probability space defined in (2.11). For $\varphi \equiv \varphi^{\omega}$ as in 2.12, the quantity

$$e^{\frac{2}{p}\|\varphi\|_{L^{p}}^{p}}\chi_{\{\|\varphi\|_{L^{2}}\leq B\}}$$
(3.1)

is in $L^1(d\mu)$ for $p \in [4,6)$ for B > 0 arbitrary and p = 6 for B > 0 sufficiently small. Moreover, in (3.1), we can take the $\frac{2}{p}$ in the exponential to be any positive constant.

Remark 3.1.2. When p = 6, the optimal value of *B* in Lemma 3.1.1 was recently determined in [65, Theorem 1.1 (ii)]. We do not need to use this precise result since we work with p = 4 in the cubic case.

Remark 3.1.3. When p = 6, the maximum value of *B* depends on the constant in the exponential. For details, see (A.13) below. We do not use this in the cubic case.

Corollary 3.1.4. Let $(\mathbb{C}^{\mathbb{N}}, \mathcal{G}, \mu)$ be the probability space defined in (2.11), and let $w \in L^1(\mathbb{T})$. For $\varphi \equiv \varphi^{\omega}$,

$$e^{-\frac{1}{2}\int dx \, dy \, |\varphi(x)|^2 \, w(x-y) \, |\varphi(y)|^2} \chi_{\{\|\varphi\|_{L^2} \le B\}}$$

is in $L^1(d\mu)$ for B > 0 arbitrary.

We note that Corollary 3.1.4 follows from Lemma 3.1.1 with p = 4 by the same argument as estimate (3.3) below.

Hölder's inequality for Schatten spaces

We have the following version of Hölder's inequality for Schatten spaces (1.11), found in [76].

Lemma 3.1.5 (Hölder's Inequality). Given $p_1, p_2 \in [1, \infty]$ with $\frac{1}{p_1} + \frac{1}{p_2} = \frac{1}{p}$ and $\mathcal{A}_j \in \mathfrak{S}^{p_j}(\mathcal{F})$ we have

$$\|\mathcal{A}_1\mathcal{A}_2\|_{\mathfrak{S}^p(\mathcal{F})} \leq \|\mathcal{A}_1\|_{\mathfrak{S}^{p_1}(\mathcal{F})}\|\mathcal{A}_2\|_{\mathfrak{S}^{p_2}(\mathcal{F})}.$$

3.1.2 Basic estimates

Let us first note the following bound on the classical interaction.

Lemma 3.1.6. Suppose that $\mathcal{W} = \frac{1}{2} \int dx \, dy \, |\varphi(x)|^2 \, w(x-y) |\varphi(y)|^2$ is defined as in (2.15). The following estimates hold.

1. For $w \in L^{\infty}(\mathbb{T})$

$$|\mathcal{W}| \le \frac{1}{2} \|w\|_{L^{\infty}} \|\varphi\|_{L^{2}}^{4}.$$
(3.2)

2. For
$$w \in L^1(\mathbb{T})$$

 $|\mathcal{W}| \leq \frac{1}{2} ||w||_{L^1} ||\varphi||_{L^4}^4.$ (3.3)

Proof. For (1), we note that

$$\begin{aligned} |\mathcal{W}| &= \frac{1}{2} \left| \int dx \, dy \, |\varphi(x)|^2 w(x-y) |\varphi(y)|^2 \right| \\ &\leq \frac{1}{2} \|w\|_{L^{\infty}} \int dx \, dy \, |\varphi(x)|^2 |\varphi(y)|^2 = \frac{1}{2} \|w\|_{L^{\infty}} \|\varphi\|_{L^2}^4. \end{aligned}$$

For (2), we apply Cauchy-Schwarz and Young's inequality to get (3.3).

For the remainder of this section, we fix $p \in \mathbb{N}^*$. Unless otherwise specified, we consider $\xi \in \mathcal{L}(\mathfrak{h}^{(p)})$. Moreover, $\|\cdot\|$ denotes the operator norm. The following lemma follows from the definition of $\Theta(\xi)$ in (2.14).

Lemma 3.1.7. We have

$$|\Theta(\xi)| \le \|\varphi\|_{\mathfrak{h}}^{2p} \|\xi\|.$$

Let us note that with Θ_{τ} as in (2.28), we have

$$\Theta_{\tau}(\xi)\big|_{\mathfrak{h}^{(n)}} = \begin{cases} \frac{p!}{\tau^{p}} \binom{n}{p} P_{+} \left(\xi \otimes \mathbf{1}^{(n-p)}\right) P_{+} & \text{if } n \geq p\\ 0 & \text{otherwise} \,, \end{cases}$$
(3.4)

where $\mathbf{1}^{(q)}$ denotes the identity map on $\mathfrak{h}^{(q)}$ and P_+ is the orthogonal projection onto the subspace of symmetric tensors. More details of the above equality can be found in [45, (3.88)]. We also have the quantum analogue of Lemma 3.1.7, which follows from (3.4).

Lemma 3.1.8. For all $n \in \mathbb{N}^*$, we have

$$\left\|\Theta_{\tau}(\xi)\right|_{\mathfrak{h}^{(n)}}\right\| \leq \left(\frac{n}{\tau}\right)^{p} \|\xi\|.$$

3.2 The time-independent problem with bounded interaction potential. Proof of Theorem 2.1.7.

In this section, we study the time-independent problem with bounded interaction potential. In Section 3.2.1, we set up the Duhamel expansion in the quantum setting. For this expansion, bounds on the explicit term are shown in Section 3.2.2 and bounds on the remainder term are shown in Section 3.2.3. The analogous expansion in the classical setting is analysed in Section 3.2.4. In Section 3.2.4, we prove convergence of the explicit terms. The proof of Theorem 2.1.7 is given in Section 3.2.6.

3.2.1 Duhamel expansion

Throughout this section, we take $w \in L^{\infty}(\mathbb{T})$. Note that with ρ_{τ} defined as in (2.33), we have

$$\rho_{\tau}(\Theta_{\tau}(\xi)) = \frac{\tilde{\rho}_{\tau,1}(\Theta_{\tau}(\xi))}{\tilde{\rho}_{\tau,1}(\mathbf{I})},\tag{3.5}$$

where

$$\tilde{\rho}_{\tau,\zeta}(\mathcal{A}) := \frac{1}{Z_{\tau,0}} \operatorname{Tr} \left(\mathcal{A} e^{-H_{\tau,0} - \zeta \mathcal{W}_{\tau}} f(\mathcal{N}_{\tau}) \right),$$
(3.6)

and I denotes the identity operator on \mathcal{F} . Here, we recall the definition (2.34) of $Z_{\tau,0}$. With notation as above, we define

$$A^{\xi}_{\tau}(\zeta) := \tilde{\rho}_{\tau,\zeta}(\Theta_{\tau}(\xi)) \,.$$

Performing a Duhamel expansion by up to order $M \in \mathbb{N}$ by iterating the identity $e^{X+\zeta Y} = e^X + \zeta \int_0^1 dt \, e^{(1-t)X} Y e^{t(X+\zeta Y)}$ yields the following result.

Lemma 3.2.1. For $M \in \mathbb{N}$, we have $A^{\xi}_{\tau}(\zeta) = \sum_{m=0}^{M-1} a^{\xi}_{\tau,m} \zeta^m + R^{\xi}_{\tau,M}(\zeta)$, where

$$a_{\tau,m}^{\xi} := \frac{(-1)^m}{Z_{\tau,0}} \operatorname{Tr} \left(\int_0^1 dt_1 \int_0^{t_1} dt_2 \dots \int_0^{t_{m-1}} dt_m \,\Theta_\tau(\xi) \mathrm{e}^{-(1-t_1)H_{\tau,0}} \,\mathcal{W}_\tau \right. \\ \left. \times \,\mathrm{e}^{-(t_1-t_2)H_{\tau,0}} \,\mathcal{W}_\tau \,\mathrm{e}^{-(t_2-t_3)H_{\tau,0}} \dots \mathrm{e}^{-(t_{m-1}-t_m)H_{\tau,0}} \,\mathcal{W}_\tau \,\mathrm{e}^{-t_m H_{\tau,0}} f(\mathcal{N}_\tau) \right)$$

$$(3.7)$$

and

$$R_{\tau,M}^{\xi}(\zeta) := \frac{(-1)^{M} \zeta^{M}}{Z_{\tau,0}} \operatorname{Tr} \left(\int_{0}^{1} dt_{1} \int_{0}^{t_{1}} dt_{2} \dots \int_{0}^{t_{M-1}} dt_{M} \Theta_{\tau}(\xi) \mathrm{e}^{-(t_{-1})H_{\tau,0}} \mathcal{W}_{\tau} \right. \\ \left. \times \mathrm{e}^{-(t_{1}-t_{2})H_{\tau,0}} \dots \mathcal{W}_{\tau} \mathrm{e}^{-(t_{M-1}-t_{M})H_{\tau,0}} \mathcal{W}_{\tau} \right. \\ \left. \times \mathrm{e}^{-t_{M}(H_{\tau,0}+\zeta\mathcal{W}_{\tau})} f(\mathcal{N}_{\tau}) \right).$$

We also define

$$\mathfrak{A} := \{ \mathbf{t} \in \mathbb{R}^m : 0 < t_m < t_{m-1} \dots < t_1 < 1 \}.$$
(3.8)

3.2.2 Bounds on the explicit terms

Throughout the following proofs, we will use without mention that for any function $g: \mathbb{C} \to \mathbb{C}, g(\mathcal{N}_{\tau})$ commutes with all operators on \mathcal{F} that commute with \mathcal{N}_{τ} , which is clear from the definition of $g(\mathcal{N}_{\tau})$. Namely, $g(\mathcal{N}_{\tau})$ acts on the p^{th} sector of Fock space as multiplication by $g(n/\tau)$. In particular, all of the operators appearing in the integrands of $a_{\tau,m}^{\xi}$ and $R_{\tau,M}^{\xi}$ commute with $g(\mathcal{N}_{\tau})$.

Lemma 3.2.2. For $m \in \mathbb{N}$, we have

$$\left|a_{\tau,m}^{\xi}\right| \le \frac{K^{p} \|\xi\| \left(K^{2} \|w\|_{L^{\infty}}\right)^{m}}{2^{m} m!}.$$
(3.9)

Proof. Lemma 3.1.5 implies

$$\begin{aligned} \left| a_{\tau,m}^{\xi} \right| &\leq \frac{1}{Z_{\tau,0}} \int_{0}^{1} dt_{1} \int_{0}^{t_{1}} dt_{2} \dots \int_{0}^{t_{m-1}} dt_{m} \left\| \Theta_{\tau}(\xi) f^{\frac{1}{m+1}}(\mathcal{N}_{\tau}) \right\|_{\mathfrak{S}^{\infty}} \\ &\times \left\| e^{-(1-t_{1})H_{\tau,0}} \right\|_{\mathfrak{S}^{\frac{1}{1-t_{1}}}} \left\| \mathcal{W}_{\tau} f^{\frac{1}{m+1}}(\mathcal{N}_{\tau}) \right\|_{\mathfrak{S}^{\infty}} \left\| e^{-(t_{1}-t_{2})H_{\tau,0}} \right\|_{\mathfrak{S}^{\frac{1}{t_{1}-t_{2}}}} \\ &\times \dots \left\| \mathcal{W}_{\tau} f^{\frac{1}{m+1}}(\mathcal{N}_{\tau}) \right\|_{\mathfrak{S}^{\infty}} \left\| e^{-t_{m}H_{\tau,0}} \right\|_{\mathfrak{S}^{\frac{1}{t_{m}}}}. \end{aligned}$$
(3.10)

Since $e^{-sH_{\tau,0}}$ is a positive operator for $s \in [0,1]$, we have $\|e^{-sH_{\tau,0}}\|_{\mathfrak{S}^{1/s}} = (Z_{\tau,0})^s$. So it follows from (3.10) that

$$\left|a_{\tau,m}^{\xi}\right| \leq \frac{Z_{\tau,0}}{Z_{\tau,0}} \frac{1}{m!} \left\|\Theta_{\tau}(\xi) f^{\frac{1}{m+1}}(\mathcal{N}_{\tau})\right\|_{\mathfrak{S}^{\infty}} \left\|\mathcal{W}_{\tau} f^{\frac{1}{m+1}}(\mathcal{N}_{\tau})\right\|_{\mathfrak{S}^{\infty}}^{m}.$$
 (3.11)

From Lemma 3.1.8, for fixed n we have

$$\left\|\Theta_{\tau}(\xi)f^{\frac{1}{m+1}}(\mathcal{N}_{\tau})\right|_{\mathfrak{h}^{(n)}}\right\|_{\mathfrak{S}^{\infty}} \leq \left(\frac{n}{\tau}\right)^{p} \left|f^{\frac{1}{m+1}}\left(\frac{n}{\tau}\right)\right| \|\xi\| \leq K^{p} \|\xi\|,$$
(3.12)

where the final inequality follows from Assumption 2.1.4. It follows from (3.12) that, when viewed as an operator on \mathcal{F}

$$\left\|\Theta_{\tau}(\xi)f^{\frac{1}{m+1}}(\mathcal{N}_{\tau})\right\|_{\mathfrak{S}^{\infty}} \leq K^{p}\|\xi\|.$$
(3.13)

To bound $\left\| \mathcal{W}_{\tau} f^{\frac{1}{m+1}}(\mathcal{N}_{\tau}) \right\|_{\mathfrak{S}^{\infty}}$ we note that \mathcal{W}_{τ} acts on $\mathfrak{h}^{(n)}$ as multiplication by

$$\frac{1}{\tau^2} \sum_{1 \le i < j \le n} w(x_i - x_j).$$
(3.14)

In particular, arguing as in (3.13), it follows that

$$\left\| \mathcal{W}_{\tau} f^{\frac{1}{m+1}}(\mathcal{N}_{\tau}) \right\|_{\mathfrak{S}^{\infty}} \leq \frac{1}{2} K^2 \|w\|_{L^{\infty}}.$$
(3.15)

Combining (3.11) with (3.13) and (3.15), we have (3.9).

3.2.3 Bounds on the remainder term

The following bound holds on the remainder term.

Lemma 3.2.3. Let $M \in \mathbb{N}$, and $\mathbf{t} \in \mathfrak{A}$ (as in (3.8)) be given. Define

$$\mathcal{R}^{\xi}_{\tau,M}(\mathbf{t},\zeta) := \Theta_{\tau}(\xi) e^{-(1-t_1)H_{\tau,0}} \mathcal{W}_{\tau} e^{-(t_1-t_2)H_{\tau,0}} \dots \times \mathcal{W}_{\tau} e^{-(t_{M-1}-t_M)H_{\tau,0}} \mathcal{W}_{\tau} e^{-t_M(H_{\tau,0}+\zeta\mathcal{W}_{\tau})} f(\mathcal{N}_{\tau}) \,.$$

Then for any $\zeta \in \mathbb{C}$,

$$\frac{1}{Z_{\tau,0}} \left| \operatorname{Tr} \left(\mathcal{R}_{\tau,M}^{\xi}(\mathbf{t},\zeta) \right) \right| \le e^{|\operatorname{Re}(\zeta)|K^2||w||_{L^{\infty}}} \frac{K^p ||\xi|| \left(K^2 ||w||_{L^{\infty}} \right)^M}{2^M}.$$
(3.16)

Proof. Define

$$\mathcal{S}(\mathbf{t}) := \Theta_{\tau}(\xi) \mathrm{e}^{-(1-t_1)H_{\tau,0}} \mathcal{W}_{\tau} \mathrm{e}^{-(t_1-t_2)H_{\tau,0}} \dots \mathcal{W}_{\tau} \mathrm{e}^{-(t_{M-1}-t_M)H_{\tau,0}} \mathcal{W}_{\tau}.$$

Then

$$\operatorname{Tr}\left(\mathcal{R}^{\xi}_{\tau,M}(\mathbf{t},\zeta)\right) = \sum_{n\geq 0} \operatorname{Tr}\left(\left[\mathcal{S}(\mathbf{t})f^{\frac{1}{2}}(\mathcal{N}_{\tau})\right]\left[e^{-t_{M}(H_{\tau,0}+\zeta\mathcal{W}_{\tau})}f^{\frac{1}{2}}(\mathcal{N}_{\tau})\right]\right)^{(n)},\quad(3.17)$$

where the trace on the left hand side of (3.17) is taken over Fock space, whereas on the right hand side for each term it is taken over the n^{th} sector of Fock space. For $n \in N$, we have

$$\operatorname{Tr}\left(\left[\mathcal{S}(\mathbf{t})f^{\frac{1}{2}}(\mathcal{N}_{\tau})\right]\left[\mathrm{e}^{-t_{M}(H_{\tau,0}+\zeta\mathcal{W}_{\tau})}f^{\frac{1}{2}}(\mathcal{N}_{\tau})\right]\right)^{(n)}$$
$$=\int_{\mathbb{T}^{n}}d\mathbf{x}\int_{\mathbb{T}^{n}}d\mathbf{y}\left(\mathcal{S}(\mathbf{t})f^{\frac{1}{2}}(\mathcal{N}_{\tau})\right)^{(n)}(\mathbf{y};\mathbf{x})\left(\mathrm{e}^{-t_{M}(H_{\tau,0}+\zeta\mathcal{W}_{\tau})}f^{\frac{1}{2}}(\mathcal{N}_{\tau})\right)^{(n)}(\mathbf{x};\mathbf{y}).$$

We now rewrite $\left(e^{-t_M(H_{\tau,0}+\zeta W_{\tau})}f^{\frac{1}{2}}(\mathcal{N}_{\tau})\right)^{(n)}(\mathbf{x};\mathbf{y})$ using Proposition 1.7.12.

$$\begin{pmatrix} \mathrm{e}^{-t_M(H_{\tau,0}+\zeta \mathcal{W}_{\tau})} f^{\frac{1}{2}}(\mathcal{N}_{\tau}) \end{pmatrix}^{(n)}(\mathbf{x};\mathbf{y}) \\ = \int \mathbb{W}^{t_M}_{\mathbf{x},\mathbf{y}}(d\boldsymbol{\omega}) \mathrm{e}^{-\frac{\kappa n}{\tau}t_M} \mathrm{e}^{-\int_0^{t_M} ds \, \zeta \left(\frac{1}{\tau^2} \sum_{1 \le i < j \le n} w_{ij}(\omega(s))\right)} f^{\frac{1}{2}}\left(\frac{n}{\tau}\right),$$

where $\mathbb{W}_{\mathbf{x},\mathbf{y}}^t(d\boldsymbol{\omega}) := \prod_{i=1}^n \mathbb{W}_{x_i,y_i}^t(d\omega_i)$. Here we used that

$$(\mathcal{W}_{\tau})^{(n)}(\mathbf{u};\mathbf{v}) = \frac{1}{\tau^2} \sum_{1 \le i < j \le n} w(u_i - u_j) \prod_{k=1}^n \delta(u_k - v_k)$$

and defined $w_{ij}(\mathbf{u}) := w(u_i - u_j)$ for $\mathbf{u} = (u_1, \dots, u_n) \in \mathbb{T}^n$. Then

$$\left| \left(\mathrm{e}^{-t_{M}(H_{\tau,0}+\zeta \mathcal{W}_{\tau})} f^{\frac{1}{2}}(\mathcal{N}_{\tau}) \right)^{(n)}(\mathbf{x};\mathbf{y}) \right| \\
\leq \int \mathbb{W}_{\mathbf{x},\mathbf{y}}^{t_{M}}(d\boldsymbol{\omega}) \mathrm{e}^{-\frac{\kappa n}{\tau}t_{M}} \left| \mathrm{e}^{-\int_{0}^{t_{M}} ds \, \zeta \left(\frac{1}{\tau^{2}} \sum_{1 \leq i < j \leq n} w_{ij}(\boldsymbol{\omega}(s))\right)} \times f^{\frac{1}{2}}\left(\frac{n}{\tau}\right) \right| \\
\leq \sup_{\boldsymbol{\omega}} \left| \mathrm{e}^{-\int_{0}^{t_{M}} ds \, \zeta \left(\frac{1}{\tau^{2}} \sum_{1 \leq i < j \leq n} w_{ij}(\boldsymbol{\omega}(s))\right)} f^{\frac{1}{2}}\left(\frac{n}{\tau}\right) \right| \left(\mathrm{e}^{-t_{M}H_{\tau,0}} \right)^{(n)}(\mathbf{x};\mathbf{y}), \quad (3.18)$$

where we have used Proposition 1.7.12 in the second line. We have

$$\sup_{\boldsymbol{\omega}} \left| \mathrm{e}^{-\int_0^{t_M} ds \, \zeta \left(\frac{1}{\tau^2} \sum_{1 \le i < j \le n} w_{ij}(\boldsymbol{\omega}(s))\right)} f^{\frac{1}{2}}\left(\frac{n}{\tau}\right) \right| \le \mathrm{e}^{|\mathrm{Re}(\zeta)|t_M\left(\frac{n}{\tau}\right)^2 ||\boldsymbol{w}||_{L^{\infty}}} \left| f^{\frac{1}{2}}\left(\frac{n}{\tau}\right) \right|.$$
(3.19)

It follows from (3.19) that

$$\sup_{\boldsymbol{\omega}} \left| \mathrm{e}^{-\int_0^{t_M} ds \, \zeta \left(\frac{1}{\tau^2} \sum_{1 \le i < j \le n} w_{ij}(\boldsymbol{\omega}(s)) \right)} f^{\frac{1}{2}} \left(\frac{n}{\tau} \right) \right| \le \mathrm{e}^{|\mathrm{Re}(\zeta)|K^2| \|\boldsymbol{\omega}\|_{L^{\infty}}}.$$
 (3.20)

Combining (3.18) with (3.20) and the triangle inequality, we have shown

$$\left| \left(\mathrm{e}^{-t_M(H_{\tau,0} + \zeta \mathcal{W}_{\tau})} f^{\frac{1}{2}}(\mathcal{N}_{\tau}) \right)^{(n)}(\mathbf{x};\mathbf{y}) \right| \leq \mathrm{e}^{|\mathrm{Re}(\zeta)|K^2 ||w||_{L^{\infty}}} \left(\mathrm{e}^{-t_M H_{\tau,0}} \right)^{(n)}(\mathbf{x};\mathbf{y}).$$
(3.21)

Combining (3.17) with (3.21), it follows that

$$\left|\operatorname{Tr}\left(\mathcal{R}^{\xi}(\mathbf{t},\zeta)\right)\right| \leq \mathrm{e}^{|\operatorname{Re}(\zeta)|K^{2}\|w\|_{L^{\infty}}}\operatorname{Tr}\left(\int_{0}^{1}dt_{1}\int_{0}^{t_{1}}dt_{2}\dots\int_{0}^{t_{M-1}}dt_{M}\,\Theta_{\tau}(\tilde{\xi})\right)$$
$$\times \mathrm{e}^{-(1-t_{1})H_{\tau,0}}\,\widetilde{\mathcal{W}_{\tau}}\mathrm{e}^{-(t_{1}-t_{2})H_{\tau,0}}\widetilde{\mathcal{W}_{\tau}}\mathrm{e}^{-(t_{2}-t_{3})H_{\tau,0}}\dots\widetilde{\mathcal{W}_{\tau}}\mathrm{e}^{-t_{M}H_{\tau,0}}f^{\frac{1}{2}}(\mathcal{N}_{\tau})\right),$$

where $\tilde{\xi}$ is the operator with kernel $|\xi|$ and

$$\widetilde{\mathcal{W}_{\tau}} := \frac{1}{2} \int dx \, dy \, \varphi_{\tau}^*(x) \varphi_{\tau}^*(y) |w(x-y)| \varphi_{\tau}(x) \varphi_{\tau}(y) \,. \tag{3.22}$$

Then (3.16) follows by arguing as in the proof of Lemma 3.2.2.

Integrating (3.16) in the variables $\mathbf{t} \in \mathfrak{A}$, as defined in (3.8), implies

$$\left| R_{\tau,M}^{\xi}(\zeta) \right| \le e^{|\operatorname{Re}(\zeta)|K^2||w||_{L^{\infty}}} \frac{K^p ||\xi|| \left(K^2 ||w||_{L^{\infty}} \right)^M}{2^M M!} |\zeta|^M.$$
(3.23)

We note that this converges to 0 as $M \to \infty$ for any fixed $\zeta \in \mathbb{C}$. Moreover, since the radius of convergence of $a_{\tau,m}^{\xi}$ is infinite by Lemma 3.2.2, we conclude the following proposition.

Proposition 3.2.4. The function $A^{\xi}_{\tau}(\zeta) = \sum_{m=0}^{\infty} a^{\xi}_{\tau,m} \zeta^m$ is analytic on \mathbb{C} .

3.2.4 The classical setting

We now analyse the analogous expansion in the classical setting. Let us note that

$$\rho(\Theta(\xi)) = \frac{\tilde{\rho}_1(\Theta(\xi))}{\tilde{\rho}_1(\mathbf{I})}, \qquad (3.24)$$

where

$$\tilde{\rho}_{\zeta}(X) := \int d\mu \, X \mathrm{e}^{-\zeta \mathcal{W}} f(\mathcal{N}) \,.$$

Define

$$A^{\xi}(\zeta) := \tilde{\rho}_{\zeta}(\Theta(\xi)) \,.$$

Then, for $M \in \mathbb{N}$

$$A^{\xi}(\zeta) = \sum_{m=0}^{M-1} a_m^{\xi} \zeta^m + R_M^{\xi}(\zeta) \,,$$

where

$$a_m^{\xi} := \frac{(-1)^m}{m!} \int d\mu \,\Theta(\xi) \mathcal{W}^m f(\mathcal{N}) \tag{3.25}$$

$$R_{M}^{\xi}(\zeta) = \frac{(-1)^{M} \zeta^{M}}{M!} \int d\mu \,\Theta(\xi) \mathcal{W}^{M} f(\mathcal{N}) \mathrm{e}^{-\tilde{\zeta}\mathcal{W}} \quad \text{for some} \quad \tilde{\zeta} \in [0, \zeta].$$
(3.26)

Lemma 3.2.5. For each $m \in \mathbb{N}$, we have

$$\left|a_{m}^{\xi}\right| \leq \frac{K^{p} \|\xi\| \left(K^{2} \|w\|_{L^{\infty}}\right)^{m}}{2^{m} m!}$$
(3.27)

Proof. We have

$$\left|a_{m}^{\xi}\right| \leq \frac{1}{m!} \int d\mu \left|\Theta(\xi) f^{\frac{1}{m+1}}(\mathcal{N})\right| \left|\mathcal{W} f^{\frac{1}{m+1}}(\mathcal{N})\right|^{m}.$$
(3.28)

From Lemma 3.1.7 and Assumption 2.1.4, we have

$$\left|\Theta(\xi)f^{\frac{1}{m+1}}(\mathcal{N})\right| \le \left\|f^{\frac{1}{m+1}}\right\|_{L^{\infty}} K^p \|\xi\|.$$
(3.29)

Moreover, Lemma 3.1.6 (1) and Assumption 2.1.4 imply

$$\left| \mathcal{W}f^{\frac{1}{m+1}}(\mathcal{N}) \right| \le \frac{1}{2} \|w\|_{L^{\infty}} \left\| f^{\frac{1}{m+1}} \right\|_{L^{\infty}} K^{2}.$$
(3.30)

Recalling $||f||_{L^{\infty}} \leq 1$, (3.27) follows from (3.28) combined with (3.29) and (3.30). \Box

Note that Lemma 3.1.6 implies that

$$\left| \mathrm{e}^{-\tilde{\zeta}\mathcal{W}} f^{\frac{1}{M+2}}(\mathcal{N}) \right| \le \mathrm{e}^{\frac{1}{2}|\mathrm{Re}(\zeta)|K^2||w||_{L^{\infty}}}$$

for $\tilde{\zeta} \in [0, \zeta]$. Applying the same arguments as the proof of Lemma 3.2.5, we have the following lemma.

Lemma 3.2.6. For any $M \in \mathbb{N}$, we have

$$\left| R_{M}^{\xi}(\zeta) \right| \le e^{\frac{1}{2} |\operatorname{Re}(\zeta)| K^{2} ||w||_{L^{\infty}}} \frac{K^{p} ||\xi|| \left(K^{2} ||w||_{L^{\infty}} \right)^{M}}{M! \, 2^{M}} |\zeta|^{M}.$$
(3.31)

Like in the quantum case, for each fixed $\zeta \in \mathbb{C}$, $R_M^{\xi}(\zeta)$ converges to 0 as $M \to \infty$ and a_m^{ξ} has infinite radius of convergence, so we have the following result.

Proposition 3.2.7. The function $A^{\xi}(\zeta) = \sum_{m=0}^{\infty} a_m^{\xi} \zeta^m$ is analytic in \mathbb{C} .

3.2.5 Convergence of the explicit terms

When analysing the convergence of the explicit terms, we argue similarly as in [30, Section 3.1] and rewrite $f(\mathcal{N}_{\#})$ as an integral of the form

$$f(\mathcal{N}_{\#}) = \int_{\mathbb{C}} d\zeta \, \frac{\psi(\zeta)}{\mathcal{N}_{\#} - \zeta} \,, \qquad (3.32)$$

for suitable $\psi \in C_c^{\infty}(\mathbb{C})$. For the precise setup, see (3.46)–(3.47) below. Using (3.32), we use that

$$\frac{1}{\mathcal{N}_{\#} - \zeta} = \int_0^\infty d\nu \,\mathrm{e}^{-\nu(\mathcal{N}_{\#} - \zeta)}\,,\tag{3.33}$$

for $\operatorname{Re} \zeta < 0$, which leads us to analyse analogues of (3.7) and (3.25) without the truncation $f(\mathcal{N})$ and with chemical potential shifted by $\nu > 0$. More precisely, we note the following boundedness and convergence result. We recall that here we are always considering $w \in L^{\infty}$.

Lemma 3.2.8. Fix $\nu > 0$. We recall C_p given by (1.13) and consider $\xi \in C_p$. Let

$$b_{\tau,m}^{\xi,\nu} := \frac{(-1)^m}{Z_{\tau,0}} \operatorname{Tr} \left(\int_0^1 dt_1 \int_0^{t_1} dt_2 \dots \int_0^{t_{m-1}} dt_m \,\Theta_\tau(\xi) \mathrm{e}^{-(1-t_1)(H_{\tau,0}+\nu\mathcal{N}_\tau)} \right) \\ \times \mathcal{W}_\tau \mathrm{e}^{-(t_1-t_2)(H_{\tau,0}+\nu\mathcal{N}_\tau)} \mathcal{W}_\tau \mathrm{e}^{-(t_2-t_3)(H_{\tau,0}+\nu\mathcal{N}_\tau)} \dots \\ \times \mathrm{e}^{-(t_{m-1}-t_m)(H_{\tau,0}+\nu\mathcal{N}_\tau)} \mathcal{W}_\tau \mathrm{e}^{-t_m(H_{\tau,0}+\nu\mathcal{N}_\tau)} \right) \\ b_m^{\xi,\nu} := \frac{(-1)^m}{m!} \int d\mu \,\Theta(\xi) \mathcal{W}^m \mathrm{e}^{-\nu\mathcal{N}}.$$

Then, the following results hold

1.
$$\left| b_{\#,m}^{\xi,\nu} \right| \le C(m,p,\nu).$$

2. $b_{\tau,m}^{\xi,\nu} \to b_m^{\xi,\nu}$ as $\tau \to \infty$ uniformly in $\xi \in \mathcal{C}_p.$

Proof. Let us first consider the case when $\xi \in \mathfrak{B}_p$. We define

$$h^{\nu} := h + \nu = \sum_{k \in \mathbb{N}} (\lambda_k + \nu) u_k u_k^* \,.$$

Then the deformed classical state defined by

$$\widetilde{\rho}_0^{\nu}(X) := \frac{\int d\mu \, X \mathrm{e}^{-\nu \mathcal{N}}}{\int d\mu \, \mathrm{e}^{-\nu \mathcal{N}}} \tag{3.34}$$

satisfies a Wick theorem with Green function given by $G^{\nu} := \frac{1}{h^{\nu}}$. This is the same as in Proposition 2.1.2, since all we have done is shift the chemical potential by ν .

Moreover, the deformed quasi-free state defined by

$$\widetilde{\rho}_{\tau,0}^{\nu}(\mathcal{A}) := \frac{\operatorname{Tr}\left(\mathcal{A} e^{-H_{\tau,0}-\nu\mathcal{N}_{\tau}}\right)}{\operatorname{Tr}\left(e^{-H_{\tau,0}-\nu\mathcal{N}_{\tau}}\right)}$$
(3.35)

satisfies a quantum Wick theorem similar to [29, Lemma B.1] (see Lemma 4.2.19)

with quantum Green function $G_{\tau} = \frac{1}{\tau(e^{h/\tau}-1)}$ replaced by

$$G_\tau^\nu := \frac{1}{\tau(\mathrm{e}^{h^\nu/\tau} - 1)}$$

In particular, we have that $\|G_{\#}^{\nu}\|_{\mathfrak{S}^2} \leq \|G_{\#}\|_{\mathfrak{S}^2} < \infty$. Let us define

$$\tilde{b}_{\tau,m}^{\xi,\nu} := \frac{(-1)^m}{\mathrm{Tr}\left(\mathrm{e}^{-H_{\tau,0}-\nu\mathcal{N}_{\tau}}\right)} \mathrm{Tr}\left(\int_0^1 dt_1 \int_0^{t_1} dt_2 \dots \int_0^{t_{m-1}} dt_m \,\Theta_{\tau}(\xi) \mathrm{e}^{-(1-t_1)(H_{\tau,0}+\nu\mathcal{N}_{\tau})} \times \mathcal{W}_{\tau} \mathrm{e}^{-(t_1-t_2)(H_{\tau,0}+\nu\mathcal{N}_{\tau})} \mathcal{W}_{\tau} \mathrm{e}^{-(t_2-t_3)(H_{\tau,0}+\nu\mathcal{N}_{\tau})} \dots \times \mathrm{e}^{-(t_{m-1}-t_m)(H_{\tau,0}+\nu\mathcal{N}_{\tau})} \mathcal{W}_{\tau} \mathrm{e}^{-t_m(H_{\tau,0}+\nu\mathcal{N}_{\tau})}\right)$$

and

$$\tilde{b}_m^{\xi,\nu} := \frac{1}{m!} \frac{(-1)^m}{\int d\mu \, e^{-\nu \mathcal{N}}} \int d\mu \, \Theta(\xi) \mathcal{W}^m e^{-\nu \mathcal{N}}$$

Noting that noting that the arguments in [29, Sections 2.3–2.6] concerning explicit terms do not use any positivity properties of w, we hence obtain that the following properties hold.

(1') $\left| \tilde{b}_{\#,m}^{\xi,\nu} \right| \leq C(m,p,\nu).$ (2') $\tilde{b}_{\tau,m}^{\xi,\nu} \to \tilde{b}_m^{\xi,\nu}$ as $\tau \to \infty$ uniformly in $\xi \in \mathcal{B}_p$.

More precisely, (1') and (2') correspond to the 1 dimensional versions¹ of [29, Corollary 2.21, Proposition 2.26] proved in [29, Section 4.1], as well as [29, Lemma 3.1].

When $\xi \in \mathfrak{B}_p$, we deduce the claim from (1') and (2') by noting that by [30, Lemma 3.4], we have

$$\lim_{\tau \to \infty} \frac{\operatorname{Tr} \left(\mathcal{A} e^{-H_{\tau,0} - \nu \mathcal{N}_{\tau}} \right)}{\operatorname{Tr} \left(\mathcal{A} e^{-H_{\tau,0}} \right)} = \int d\mu \, e^{-\nu \mathcal{N}}$$

It remains to consider the case when $\xi = \mathbf{1}_p$ is the identity operator on $\mathfrak{h}^{(p)}$. We then have

$$\xi(x_1, \dots, x_p; y_1, \dots, y_p) = \prod_{j=1}^p \delta(x_j - y_j).$$
(3.36)

Since $\tilde{\rho}_{\tau,0}$ satisfies the quantum Wick theorem, we can argue analogously as in [29, Section 4.2] to get the required bounds and convergence as before. We omit the details.

¹Throughout the thesis, when referring to [29, Corollary 2.21, Proposition 2.26], we mean these 1 dimensional versions.

We also need the following result.

Lemma 3.2.9. Let $\mathcal{A} : \mathcal{F} \to \mathcal{F}$ and $g \in L^{\infty}(\mathbb{R})$. Then $|\operatorname{Tr}(\mathcal{A} g(\mathcal{N}_{\tau}))| \leq ||g||_{L^{\infty}} \operatorname{Tr}(\hat{\mathcal{A}})$, where $\hat{\mathcal{A}}^{(n)}$ has kernel $|\mathcal{A}^{(n)}(x;y)|$.

Proof. For an operator $\mathcal{A} : \mathcal{F} \to \mathcal{F}$, we define $\mathcal{A}^{(n)} := P^{(n)} \mathcal{A} P^{(n)}$, where $P^{(n)}$ is the projection of an operator on Fock space to the n^{th} component of Fock space. We also define $\hat{\mathcal{A}} := \bigoplus_{n>0} \hat{\mathcal{A}}^{(n)}$. We have

$$\begin{aligned} |\mathrm{Tr}(\mathcal{A}\,g(\mathcal{N}_{\tau}))| &= \left|\sum_{n\geq 0}\int_{\mathbb{T}^n} d\mathbf{x}\,\mathcal{A}^{(n)}(\mathbf{x};\mathbf{x})g\left(\frac{n}{\tau}\right)\right| \\ &\leq \sup_{n\geq 0}\left\{\left|g\left(\frac{n}{\tau}\right)\right|\right\}\sum_{n\geq 0}\int_{\mathbb{T}^n} d\mathbf{x}\,|A^{(n)}(\mathbf{x};\mathbf{x})| \\ &\leq \|g\|_{L^{\infty}}\mathrm{Tr}(\hat{\mathcal{A}})\,. \end{aligned}$$

Lemma 3.2.10. We recall the definitions (3.7) and (3.25). For each $m \in \mathbb{N}$, we have

$$\lim_{\tau \to \infty} a_{\tau,m}^{\xi} = a_m^{\xi} \tag{3.37}$$

uniformly in $\xi \in C_p$, defined in (1.13).

Proof. For $\zeta \in \mathbb{C} \setminus [0, \infty)$, we define

$$\alpha_{\tau,m}^{\xi}(\zeta) := \frac{1}{Z_{\tau,0}} \operatorname{Tr} \left(\int_{0}^{1} dt_{1} \int_{0}^{t_{1}} dt_{2} \dots \int_{0}^{t_{m-1}} dt_{m} \Theta_{\tau}(\xi) \mathrm{e}^{-(1-t_{1})H_{\tau,0}} \mathcal{W}_{\tau} \mathrm{e}^{-(t_{1}-t_{2})H_{\tau,0}} \right)$$
$$\times \mathcal{W}_{\tau} \dots \mathrm{e}^{-(t_{m-1}-t_{m})H_{\tau,0}} \mathcal{W}_{\tau} \mathrm{e}^{-t_{m}H_{\tau,0}} \frac{1}{\mathcal{N}_{\tau}-\zeta} \right)$$

and

$$\alpha_m^{\xi}(\zeta) := \frac{1}{m!} \int d\mu \,\Theta(\xi) \mathcal{W}^m \frac{1}{\mathcal{N} - \zeta} \,. \tag{3.38}$$

We prove that $\alpha_{\tau,m}^{\xi}$ and α_m^{ξ} are analytic in $\zeta \in \mathbb{C} \setminus [0, \infty)$. We first deal with α_m^{ξ} . Note that

$$\left|\alpha_m^{\xi}(\zeta)\right| \leq \frac{1}{m!} \int d\mu \left|\Theta(\xi)\mathcal{W}^m\right| \left|\frac{1}{\mathcal{N}-\zeta}\right|$$

Using Lemma 3.1.7, Lemma 3.1.6 (1), and that $\int d\mu \|\varphi\|_{\mathfrak{h}}^{2p} \leq C(p)$ by Remark 3.2.11, we have

$$|\alpha_m^{\xi}(\zeta)| \le \frac{C(m,p)}{\max\{-\operatorname{Re}\zeta, |\operatorname{Im}\zeta|\}}.$$
(3.39)

Arguing similarly to (3.39), it follows that

$$\frac{1}{m!} \int d\mu \left| \Theta(\xi) \mathcal{W}^m \right| \left| \frac{1}{\left(\mathcal{N} - \zeta \right)^2} \right| \le \frac{C(m, p)}{\max\{-\operatorname{Re} \zeta, |\operatorname{Im} \zeta|\}^2},$$

so by the dominated convergence theorem, we can differentiate under the integral sign in (3.38) and conclude that α_m^{ξ} is analytic in $\mathbb{C} \setminus [0, \infty)$.

To show $\alpha_{\tau,m}^{\xi}$ is analytic in $\mathbb{C}\setminus[0,\infty)$, we first note that $\frac{1}{N_{\tau}-\zeta}$ acts as multiplication by $\frac{1}{(n/\tau)-\zeta}$ on the n^{th} sector of Fock space. By using Lemma 3.2.9 we get

$$\begin{aligned} |\alpha_{\tau,m}(\zeta)| &\leq \frac{1}{Z_{\tau,0}} \mathrm{Tr} \bigg(\bigg[\int_0^1 dt_1 \int_0^{t_1} dt_2 \dots \int_0^{t_{m-1}} dt_m \,\Theta_{\tau}(\xi) \mathrm{e}^{-(1-t_1)H_{\tau,0}} \mathcal{W}_{\tau} \mathrm{e}^{-(t_1-t_2)H_{\tau,0}} \\ &\times \mathcal{W}_{\tau} \dots \mathrm{e}^{-(t_{m-1}-t_m)H_{\tau,0}} \mathcal{W}_{\tau} \mathrm{e}^{-t_m H_{\tau,0}} \bigg]^{\wedge} \bigg) \frac{1}{\max\{-\mathrm{Re}\,\zeta, |\mathrm{Im}\,\zeta|\}} \\ &\leq \frac{1}{Z_{\tau,0}} \mathrm{Tr} \bigg(\int_0^1 dt_1 \int_0^{t_1} dt_2 \dots \int_0^{t_{m-1}} dt_m \,\Theta_{\tau}(\tilde{\xi}) \mathrm{e}^{-(1-t_1)H_{\tau,0}} \widetilde{\mathcal{W}_{\tau}} \mathrm{e}^{-(t_1-t_2)H_{\tau,0}} \\ &\times \widetilde{\mathcal{W}_{\tau}} \dots \mathrm{e}^{-(t_{m-1}-t_m)H_{\tau,0}} \widetilde{\mathcal{W}_{\tau}} \mathrm{e}^{-t_m H_{\tau,0}} \bigg) \frac{1}{\max\{-\mathrm{Re}\,\zeta, |\mathrm{Im}\,\zeta|\}}, \end{aligned}$$

where we recall $\tilde{\xi}$ is the operator with kernel $|\xi|$, and $\widetilde{W_{\tau}}$ is as in (3.22). Applying [29, Corollary 2.21], we have

$$|\alpha_{\tau,m}^{\xi}(\zeta)| \le \frac{C(m,p)}{\max\{-\operatorname{Re}\zeta, |\operatorname{Im}\zeta|\}}.$$
(3.40)

Define

$$\mathfrak{h}^{(\leq p)} := \bigoplus_{n=0}^{p} \mathfrak{h}^{(n)} , \qquad (3.41)$$

and

$$P^{(\leq p)}: \mathcal{F} \to \mathfrak{h}^{(\leq p)}$$

as the orthogonal projection. Define

$$\alpha_{\tau,m,n}(\zeta) := \frac{1}{Z_{\tau,0}} \operatorname{Tr} \left(\int_0^1 dt_1 \int_0^{t_1} dt_2 \dots \int_0^{t_{m-1}} dt_m P^{(\leq n)} \Theta_{\tau}(\xi) e^{-(1-t_1)H_{\tau,0}} \mathcal{W}_{\tau} \right)$$
$$\times e^{-(t_1-t_2)H_{\tau,0}} \mathcal{W}_{\tau} \dots e^{-(t_{m-1}-t_m)H_{\tau,0}} \mathcal{W}_{\tau} e^{-t_m H_{\tau,0}} \frac{1}{\mathcal{N}_{\tau} - \zeta} \right).$$

Since $P^{(\leq n)}$ commutes with $\Theta_{\tau}(\xi)$, $H_{\tau,0}$, and \mathcal{W}_{τ} , it follows that $\alpha_{\tau,m,n}$ is analytic in $\mathbb{C}\setminus[0,\infty)$. By construction we have $\lim_{n\to\infty} \alpha_{\tau,m,n}(\zeta) = \alpha_{\tau,m}(\zeta)$ for all $\zeta \in \mathbb{C}\setminus[0,\infty)$

and by the same argument as (3.40), we have

$$|\alpha_{\tau,m,n}^{\xi}(\zeta)| \le \frac{C(m,p)}{\max\{-\operatorname{Re}\zeta, |\operatorname{Im}\zeta|\}}.$$
(3.42)

The pointwise convergence, (3.42) and the dominated convergence theorem imply that

$$\lim_{n \to \infty} \int_{\partial T} d\zeta \, \alpha_{\tau,m,n}(\zeta) = \int_{\partial T} d\zeta \, \alpha_{\tau,m}(\zeta)$$

for any triangle T contained in $\mathbb{C}\setminus[0,\infty)$. Morera's theorem implies that $\alpha_{\tau,m}$ is analytic in $\mathbb{C}\setminus[0,\infty)$.

We now prove that $\alpha_{\tau,m}^{\xi}(\zeta) \to \alpha_m^{\xi}(\zeta)$ as $\tau \to \infty$ for all $\zeta \in \mathbb{C} \setminus [0, \infty)$. First, for $\operatorname{Re} \zeta < 0$, we recall (3.33). Therefore,

$$\begin{aligned} \left| \alpha_{\tau,m}^{\xi}(\zeta) - \alpha_{m}^{\xi}(\zeta) \right| &= \left| \frac{1}{Z_{\tau,0}} \operatorname{Tr} \left(\int_{0}^{1} dt_{1} \int_{0}^{t_{1}} dt_{2} \dots \int_{0}^{t_{m-1}} dt_{m} \Theta_{\tau}(\xi) \mathrm{e}^{-(1-t_{1})H_{\tau,0}} \mathcal{W}_{\tau} \right. \\ &\quad \times \mathrm{e}^{-(t_{1}-t_{2})H_{\tau,0}} \mathcal{W}_{\tau} \dots \mathrm{e}^{-(t_{m-1}-t_{m})H_{\tau,0}} \mathcal{W}_{\tau} \mathrm{e}^{-t_{m}H_{\tau,0}} \frac{1}{\mathcal{N}_{\tau}-\zeta} \right) \\ &\quad - \frac{1}{m!} \int d\mu \, \mathcal{W}^{m} \frac{1}{\mathcal{N}_{-\zeta}} \\ &\leq \int_{0}^{\infty} d\zeta \, \mathrm{e}^{\nu\zeta} \left| \frac{1}{Z_{\tau,0}} \operatorname{Tr} \left(\int_{0}^{1} dt_{1} \int_{0}^{t_{1}} dt_{2} \dots \int_{0}^{t_{m-1}} dt_{m} \, \Theta_{\tau}(\xi) \mathrm{e}^{-(1-t_{1})(H_{\tau,0}+\nu\mathcal{N}_{\tau})} \right) \\ &\quad \times \mathcal{W}_{\tau} \mathrm{e}^{-(t_{1}-t_{2})(H_{\tau,0}+\nu\mathcal{N}_{\tau})} \mathcal{W}_{\tau} \dots \mathrm{e}^{-(t_{m-1}-t_{m})(H_{\tau,0}+\nu\mathcal{N}_{\tau})} \mathcal{W}_{\tau} \\ &\quad \times \mathrm{e}^{-t_{m}(H_{\tau,0}+\nu\mathcal{N}_{\tau})} \right) - \frac{1}{m!} \int d\mu \, \mathrm{e}^{-\nu\mathcal{N}} \mathcal{W}^{m} \bigg|, \end{aligned} \tag{3.43}$$

where we have used part (1) of Lemma 3.2.8 and Re $\zeta < 0$ to apply Fubini's theorem. Lemma 3.2.8, (3.43), and the dominated convergence theorem give

$$\lim_{\tau \to \infty} \alpha_{\tau,m}^{\xi}(\zeta) = \alpha_m^{\xi}(\zeta) \tag{3.44}$$

uniformly in $\xi \in \mathcal{C}_p$ for $\operatorname{Re} \zeta < 0$.

We define $\beta_{\tau,m}^{\xi} := \alpha_{\tau,m}^{\xi} - \alpha_m^{\xi}$. We follow the argument in [30, Proposition 3.3] to prove

$$\lim_{\tau \to \infty} \sup_{\xi \in \mathcal{C}_p} |\beta_{\tau,m}^{\xi}(\zeta)| = 0 \text{ for all } \zeta \in \mathbb{C} \setminus [0,\infty).$$
(3.45)

From the analyticity of $\alpha_{\#,m}^{\xi}$ on $\mathbb{C}\setminus[0,\infty)$, (3.39) and (3.40), and (3.44), we know that $\beta_{\tau,m}^{\xi}$ satisfy the following properties.

1. $\beta_{\tau,m}^{\xi}$ is analytic on $\mathbb{C} \setminus [0,\infty)$.

2. $\lim_{\tau \to \infty} \sup_{\xi \in \mathcal{C}_p} |\beta_{\tau,m}^{\xi}(\zeta)| = 0$ for all $\operatorname{Re} \zeta < 0$.

3. $\sup_{\xi \in \mathcal{C}_p} |\beta_{\tau,m}^{\xi}(\zeta)| \leq \frac{C(m,p)}{|\mathrm{Im}\,\zeta|} \text{ for all } \zeta \in \mathbb{C} \setminus [0,\infty).$

Given $\varepsilon > 0$, define

$$\mathcal{D}_{\varepsilon} := \{\zeta : \operatorname{Im} \zeta > \varepsilon\}$$

and

$$\mathcal{T}_{\varepsilon} := \left\{ \zeta_0 \in \mathcal{D}_{\varepsilon} : \lim_{\tau \to \infty} \sup_{\xi \in \mathcal{C}_p} \left| \partial_{\zeta}^n \beta_{\tau,m}^{\xi}(\zeta_0) = 0 \right| \text{ for all } n \in \mathbb{N} \,.$$

So $\mathcal{T}_{\varepsilon}$ is the set of points in $\mathcal{D}_{\varepsilon}$ at which all ζ -derivatives of $\beta_{\tau,m}^{\xi}$ converge to 0 as $\tau \to \infty$ uniformly in $\xi \in \mathcal{C}_p$. Using properties (1) - (3) of β , Cauchy's integral formula, and the dominated convergence theorem, we have $\mathcal{D}_{\varepsilon} \cap \{\zeta : \operatorname{Re} \zeta < 0\} \subset \mathcal{T}_{\varepsilon}$. In particular, $\mathcal{T}_{\varepsilon}$ is not empty.

So to prove (3.45) on $\mathcal{D}_{\varepsilon}$, it suffices to show that $\mathcal{T}_{\varepsilon} = \mathcal{D}_{\varepsilon}$. Since $\mathcal{D}_{\varepsilon}$ is connected, the latter claim follows from showing that $\mathcal{T}_{\varepsilon}$ is both open and closed in $\mathcal{D}_{\varepsilon}$. We first show that $\mathcal{T}_{\varepsilon}$ is open in $\mathcal{D}_{\varepsilon}$. Given $\zeta_0 \in \mathcal{T}_{\varepsilon}$, note that $B_{\varepsilon/2}(\zeta_0) \subset \mathcal{D}_{\varepsilon/2}$. So by property (3), $\left|\beta_{\tau,m}^{\zeta}\right| \leq C(\varepsilon)$ on $B_{\varepsilon/2}(\zeta_0)$. Analyticity and Cauchy's integral formula imply that the Taylor series of $\beta_{\tau,m}^{\xi}$ at ζ_0 converges on $B_{\varepsilon/2}(\zeta_0)$. So we can differentiate term by term and use the dominated convergence theorem and $\zeta_0 \in \mathcal{T}_{\varepsilon}$ to get that $B_{\delta}(\zeta_0) \subset \mathcal{T}_{\varepsilon}$ for $\delta \in (0, \varepsilon/2)$ sufficiently small such that $B_{\delta}(\zeta_0) \subset \mathcal{D}_{\varepsilon}$. So $\mathcal{T}_{\varepsilon}$ is open in $\mathcal{D}_{\varepsilon}$.

To show that $\mathcal{T}_{\varepsilon}$ is closed in $\mathcal{D}_{\varepsilon}$, let (ζ_n) be a sequence in $\mathcal{T}_{\varepsilon}$ which converges to some $\zeta \in \mathcal{D}_{\varepsilon}$. Since $\zeta \in \mathcal{D}_{\varepsilon}$ which is open, there is $\varepsilon' \in (0, \varepsilon/2)$ such that $B_{\varepsilon'}(\zeta) \subset \mathcal{D}_{\varepsilon}$. Since $(\zeta_n) \to \zeta$, for *n* sufficiently large, $\zeta \in B_{\varepsilon'/2}(\zeta_n)$. Since $B_{\varepsilon'/2}(\zeta_n) \subset B_{\varepsilon'}(\zeta) \subset \mathcal{D}_{\varepsilon}$, the argument that $\mathcal{T}_{\varepsilon}$ is open in $\mathcal{D}_{\varepsilon}$ implies that $B_{\varepsilon'/2}(\zeta_n) \subset \mathcal{T}_{\varepsilon}$. In particular, $\zeta \in \mathcal{T}_{\varepsilon}$, so $\mathcal{T}_{\varepsilon}$ is closed in $\mathcal{D}_{\varepsilon}$. By symmetry, the same argument shows that (3.45) holds on $\widetilde{\mathcal{D}}_{\varepsilon} := \{\zeta : \operatorname{Im} \zeta < -\varepsilon\}$. Then (3.45) holds on $\mathbb{C} \setminus [0, \infty)$ by letting $\varepsilon \to 0$ and recalling that (3.45) holds for $\zeta < 0$ by property (2) above.

Applying the Helffer-Sjöstrand formula and arguing as in [30, (3.29)-(3.33)], we can find $\psi \in C_c^{\infty}(\mathbb{C})$ satisfying

$$|\psi(\zeta)| \le C |\mathrm{Im}\,\zeta| \tag{3.46}$$

such that

$$f(\mathcal{N}_{\#}) = \int_{\mathbb{C}} d\zeta \, \frac{\psi(\zeta)}{\mathcal{N}_{\#} - \zeta} \,. \tag{3.47}$$

Then (3.39), (3.40), (3.46), and $\psi \in C_c^{\infty}(\mathbb{C})$ imply that

$$|\alpha_{\#,m}^{\xi}(\zeta)\psi(\zeta)| \le F(\zeta) \tag{3.48}$$

for some $F \in L^1(\mathbb{C})$. By (3.48), we can use Fubini's theorem to write

$$|a_{\tau,m}^{\xi} - a_m^{\xi}| \le \int_{\mathbb{C}} |\psi(\zeta)| \left| \alpha_{\tau,m}^{\xi}(\zeta) - \alpha_m^{\xi}(\zeta) \right|.$$

Using $\beta_{\tau,m}^{\xi} \to 0$ as $\tau \to \infty$ almost everywhere in \mathbb{C} uniformly in ξ and (3.48), the dominated convergence theorem implies (3.37).

Remark 3.2.11. In the proof of Lemma 3.2.10, we used that $\int d\mu \|\varphi\|_{\mathfrak{h}}^{2p} \leq C(p) < \infty$. To see this, recall (2.12) implies

$$\int d\mu \, \|\varphi\|_{\mathfrak{h}}^{2p} = \mathbb{E}_{\omega} \left[\left(\sum_{n \in \mathbb{N}} \frac{|\omega_n|^2}{\lambda_n} \right)^p \right]$$
$$= \mathbb{E}_{\omega} \left[\sum_{n_i \in \mathbb{N}} \frac{\omega_{n_1} \overline{\omega_{n_1}} \dots \omega_{n_p} \overline{\omega_{n_p}}}{\lambda_{n_1} \dots \lambda_{n_p}} \right]$$
$$\leq C(p) \left(\sum_n \frac{1}{\lambda_n} \right)^p \leq C(p) < \infty$$

The final line follows from Proposition 2.1.2.

3.2.6 Convergence of correlation functions. Proof of Theorem 2.1.7

Lemma 3.2.12. $A^{\xi}_{\tau}(\zeta) \to A^{\xi}(\zeta)$ as $\tau \to \infty$ uniformly in $\xi \in C_p$.

Proof. Since $A_{\#}^{\xi}$ are analytic in \mathbb{C} , for all $\zeta \in \mathbb{C}$

$$\sup_{\xi \in \mathfrak{B}_p} \left| A_{\tau}^{\xi}(\zeta) - A^{\xi}(\zeta) \right| \le \sum_m \sup_{\xi} \left| a_{\tau,m}^{\xi} - a_m^{\xi} \right| |\zeta|^m \to 0$$

as $\tau \to \infty$. Here we have used Lemma 3.2.10, Lemma 3.2.2, and the dominated convergence theorem. We also recall the notation (1.12).

Recalling (3.5) and (3.24) and taking $\zeta = 1$, we have the following result.

Corollary 3.2.13. $\rho_{\tau}(\Theta_{\tau}(\xi)) \to \rho(\Theta(\xi))$ as $\tau \to \infty$ uniformly in $\xi \in C_p$.

Before proceeding to the proof of Theorem 2.1.7, we first need to prove the following technical lemma.

Lemma 3.2.14. Recalling (2.22) and (2.36), we have $\gamma_{\#,p} \geq 0$ in the sense of operators.

Proof. For $\eta \in \mathfrak{h}^{(p)}$, define the orthogonal projection $\Pi_{\eta}(\cdot) := \langle \eta, \cdot \rangle \eta$. Let us first note that

$$\langle \eta, \gamma_{\#,p} \eta \rangle_{\mathfrak{h}^{(p)}} = \rho_{\#}(\Theta_{\#}(\Pi_{\eta})) \,. \tag{3.49}$$

In the quantum setting, we use (2.36) and linearity to compute

$$\langle \eta, \gamma_{\tau,p} \eta \rangle_{\mathfrak{h}^{(p)}} = \int dx_1 \cdots dx_p \, dy_1 \cdots dy_p \,\overline{\eta}(x_1, \dots, x_p) \,\eta(y_1, \dots, y_p) \\ \times \rho_\tau \big(\varphi_\tau^*(y_1) \cdots \varphi_\tau^*(y_p) \varphi_\tau(x_1) \cdots \varphi_\tau(x_p) \big) \\ = \rho_\tau \bigg(\int dx_1 \cdots dx_p \, dy_1 \cdots dy_p \,\overline{\eta}(x_1, \dots, x_p) \,\eta(y_1, \dots, y_p) \\ \times \varphi_\tau^*(y_1) \cdots \varphi_\tau^*(y_p) \varphi_\tau(x_1) \cdots \varphi_\tau(x_p) \bigg).$$
(3.50)

By (2.28) and the definition of Π_{η} we deduce that that the expression in (3.50) equals $\rho_{\tau}(\Theta_{\tau}(\Pi_{\eta}))$, thus showing (3.49) in the quantum setting. Similarly in the classical setting, we use (2.22) and (2.14) to compute

$$\langle \eta, \gamma_p \eta \rangle_{\mathfrak{h}^{(p)}} = \rho \left(\int dx_1 \cdots dx_p \, dy_1 \cdots dy_p \, \overline{\eta}(x_1, \dots, x_p) \, \eta(y_1, \dots, y_p) \right. \\ \left. \times \, \overline{\varphi}(y_1) \cdots \overline{\varphi}(y_p) \varphi(x_1) \cdots \varphi(x_p) \right) = \rho(\Theta(\Pi_\eta)) \,,$$

as was claimed.

We now show that the expression on the right-hand side of (3.49) is nonnegative. Let us first show this in the quantum setting. By (2.28), we note that $\Theta_{\tau}(\Pi_{\eta})$ is a positive operator. Furthermore $f(\mathcal{N}_{\tau})$ is a positive operator which commutes with $\Theta_{\tau}(\Pi_{\eta})$. In particular, their composition is a positive operator. Recalling (2.32), we know that

$$\mathcal{A} \mapsto \frac{\operatorname{Tr}(\mathcal{A}P_{\tau})}{\operatorname{Tr}(P_{\tau})}$$

is a quantum state. In particular, when applied to positive operators it is nonnegative, so we obtain that

$$\frac{\operatorname{Tr}(\Theta_{\tau}(\Pi_{\eta})f(\mathcal{N}_{\tau})P_{\tau})}{\operatorname{Tr}(P_{\tau})} \ge 0.$$
(3.51)

Since P_{τ} and $f(\mathcal{N}_{\tau})$ commute, by using (3.51), and recalling (2.32) as well as Assumption 2.1.4, it follows that

$$\rho_{\tau}(\Theta_{\tau}(\Pi_{\eta})f(\mathcal{N}_{\tau})) = \frac{\operatorname{Tr}(\Theta_{\tau}(\Pi_{\eta})f(\mathcal{N}_{\tau})P_{\tau})}{\operatorname{Tr}(P_{\tau}f(\mathcal{N}_{\tau}))} \ge 0.$$
(3.52)

We deduce the claim in the quantum setting from (3.49) and (3.52).

In the classical setting, we use (2.14) to write

$$\rho(\Theta(\Pi_{\eta})) = \rho\left(\int dx_1 \cdots dx_p \, dy_1 \cdots dy_p \overline{\eta}(x_1), \dots, x_p) \, \eta(y_1, \dots, y_p) \\ \times \varphi(x_1) \cdots \varphi(x_p) \, \overline{\varphi}(y_1) \cdots \overline{\varphi}(y_p)\right) \\ = \rho\left(\left|\int dx_1 \cdots dx_p \overline{\eta}(x_1, \dots, x_p) \, \varphi(x_1) \cdots \varphi(x_p)\right|^2\right) \ge 0. \quad (3.53)$$

For the last inequality in (3.53), we recalled (2.20). We deduce the claim in the classical setting from (3.49) and (3.53).

Remark 3.2.15. By following the same duality argument as [30, Proposition 3.3 (ii)], we can deduce from Lemma 3.2.14 that Corollary 3.2.13 holds for all $\xi \in \mathcal{L}(\mathfrak{h}^{(p)})$.

To prove Theorem 2.1.7, we use the following result, proved in [29, Lemma 4.10].

Lemma 3.2.16. Let $p \in \mathbb{N}$ be fixed. Suppose that for all $\tau > 0$, $\gamma_{\tau} \in \mathfrak{S}^{1}(\mathfrak{h}^{(p)})$ is positive and that $\gamma \in \mathfrak{S}^{1}(\mathfrak{h}^{(p)})$ is positive. Suppose further that

$$\lim_{\tau \to \infty} \|\gamma_{\tau} - \gamma\|_{\mathfrak{S}^2(\mathfrak{h}^{(p)})} = 0, \quad and \quad \lim_{\tau \to \infty} \mathrm{Tr}\gamma_{\tau} = \mathrm{Tr}\gamma.$$
(3.54)

Then $\lim_{\tau \to \infty} \|\gamma_{\tau} - \gamma\|_{\mathfrak{S}^1(\mathfrak{h}^{(p)})} = 0.$

Proof of Theorem 2.1.7. We first prove (2.37). Let $p \in \mathbb{N}^*$ be given. We verify the conditions of Lemma 3.2.16. Using the fact that $\mathfrak{S}^2(\mathfrak{h}^{(p)}) \cong \mathfrak{S}^2(\mathfrak{h}^{(p)})^*$ and recalling that \mathfrak{B}_p is the unit ball of $\mathfrak{S}^2(\mathfrak{h}^{(p)})$, we have

$$\|\gamma_{\tau,p} - \gamma_p\|_{\mathfrak{S}^2(\mathfrak{h}^{(p)})} = \sup_{\xi \in \mathfrak{B}_p} |\operatorname{Tr} \left(\gamma_{\tau,p}\xi - \gamma_p\xi\right)| = \sup_{\xi \in \mathfrak{B}_p} |\rho_\tau(\Theta_\tau(\xi)) - \rho(\Theta(\xi))| \to 0$$
(3.55)

as $\tau \to \infty$ by Corollary 3.2.13. We also note that $\text{Tr}\gamma_{\#,p} = \rho_{\#}(\Theta_{\#}(I))$. So Corollary 3.2.13 implies

$$\lim_{\tau \to \infty} \operatorname{Tr} \gamma_{\tau,p} = \operatorname{Tr} \gamma_p. \tag{3.56}$$

Combining Lemma 3.2.14, (3.55), and (3.56), Lemma 3.2.16 implies (2.37). The proof of (2.38) is similar. Namely we start from (3.6) with $\mathcal{A} = I$ and repeat the previous argument (in which we formally set p = 0).

3.3 The time-independent problem with unbounded interaction potentials. Proofs of Theorems 2.1.8 and 2.1.9.

In this section, we analyse the time-independent problem for general w as in Assumption 2.1.1. In particular, we no longer assume that w is bounded, as in Section 3.2. In Section 3.3.1, we consider w satisfying Assumption 2.1.1 (i) and prove Theorem 2.1.8. In Section 3.3.2, we consider w satisfying Assumption 2.1.1 (ii) and prove Theorem 2.1.9. As before, we fix $p \in \mathbb{N}^*$ throughout the section.

3.3.1 L^1 interaction potentials. Proof of Theorem 2.1.8.

We first consider the case where w satisfies Assumption 2.1.1 (i), i.e. when it is taken to be an even and real-valued function in $L^1(\mathbb{T})$. To do this, we approximate w with bounded potentials w^{ε} , which are even and real-valued. For instance, we can take $w^{\varepsilon} := w\chi_{\{|w| \leq 1/\varepsilon\}}$. We then use the results of the previous section combined with a diagonal argument.

Let us first note the following result.

Lemma 3.3.1. Let w be as in Assumption 2.1.1 (i), and suppose $w^{\varepsilon} \in L^{\infty}$ is a sequence of even, real-valued interaction potentials satisfying $w^{\varepsilon} \to w$ in $L^{1}(\mathbb{T})$ as $\varepsilon \to 0$. Then there exists a sequence (ε_{τ}) converging to 0 as $\tau \to \infty$ such that for all $p \in \mathbb{N}^{*}$

$$\lim_{\tau \to \infty} \rho_{\tau}^{\varepsilon_{\tau}} \left(\Theta_{\tau}(\xi) \right) = \rho \left(\Theta \left(\xi \right) \right) \,, \tag{3.57}$$

uniformly in $\xi \in C_p$. We recall that C_p is given by (1.13).

Proof. Using a standard diagonal argument, it suffices to prove that for each fixed $\varepsilon > 0$

$$\lim_{\tau \to \infty} \rho_{\tau}^{\varepsilon} \left(\Theta_{\tau}(\xi) \right) \to \rho^{\varepsilon} \left(\Theta\left(\xi\right) \right) \tag{3.58}$$

uniformly in $\xi \in \mathcal{C}_p$, and

$$\lim_{\varepsilon \to 0} \rho^{\varepsilon} \left(\Theta \left(\xi \right) \right) \to \rho \left(\Theta \left(\xi \right) \right) \tag{3.59}$$

uniformly in $\xi \in C_p$. The convergence in (3.58) holds by Corollary 3.2.13 because $w^{\varepsilon} \in L^{\infty}(\mathbb{T})$. To show (3.59), we first note that by Lemma 3.1.6 (2) and the Sobolev embedding theorem

$$|\mathcal{W}^{\varepsilon} - \mathcal{W}| \lesssim \|w^{\varepsilon} - w\|_{L^1} \|\varphi\|_{H^{\frac{1}{2}-}}^4.$$
(3.60)

Since $\varphi \in H^{\frac{1}{2}-}$ almost surely, it follows that

$$\lim_{\varepsilon \to 0} \mathcal{W}^{\varepsilon} = \mathcal{W} \tag{3.61}$$

almost surely. Continuity of the exponential implies that

$$\lim_{\varepsilon \to 0} e^{-\mathcal{W}^{\varepsilon}} = e^{-\mathcal{W}}$$

almost surely. By Lemma 3.1.6 (2), we have

$$|\mathcal{W}| \le \frac{1}{2} ||w||_{L^1} ||\varphi||_{L^4}^4$$

and for ε sufficiently small

$$|\mathcal{W}^{\varepsilon}| \leq \frac{1}{2} \|w^{\varepsilon}\|_{L^{1}} \|\varphi\|_{L^{4}}^{4} \leq \|w\|_{L^{1}} \|\varphi\|_{L^{4}}^{4}$$

It follows that

$$|\mathrm{e}^{-\mathcal{W}^{\varepsilon}} - \mathrm{e}^{-\mathcal{W}}| \le 2\mathrm{e}^{\|w\|_{L^{1}}\|\varphi\|_{L^{4}}^{4}}.$$
 (3.62)

By Lemma 3.1.1 and Assumption 2.1.4, we know that

$$e^{\|w\|_{L^1}\|\varphi\|_{L^4}^4} f^{\frac{1}{2}}(\mathcal{N}) \in L^1(d\mu).$$
(3.63)

By Lemma 3.1.7, we have that

$$\Theta(\xi) f^{\frac{1}{2}}(\mathcal{N}) \in L^{\infty}(d\mu).$$
(3.64)

Using (3.62)–(3.64) and the dominated convergence theorem, it follows that

$$\lim_{\varepsilon \to 0} \int d\mu \, |\Theta(\xi)| \, \left| e^{-\mathcal{W}^{\varepsilon}} - e^{-\mathcal{W}} \right| f(\mathcal{N}) = 0.$$
(3.65)

The same argument implies

$$\lim_{\varepsilon \to 0} z^{\varepsilon} = z \,. \tag{3.66}$$

Noting that

$$\rho^{\varepsilon}\left(\Theta\left(\xi\right)\right) - \rho\left(\Theta\left(\xi\right)\right) = \frac{1}{z} \int d\mu \,\Theta(\xi) f(\mathcal{N}) \left(\frac{z}{z^{\varepsilon}} \mathrm{e}^{-\mathcal{W}^{\varepsilon}} - \mathrm{e}^{-\mathcal{W}}\right),$$

(3.59) follows from (3.65) and (3.66).

We can now prove Theorem 2.1.8.

Proof of Theorem 2.1.8. We deduce (2.39) from Lemma 3.3.1 by arguing analogously as in the proof of (2.37). The proof of (2.40) is similar to that of (2.39). Instead of (3.58), we use

$$\lim_{\tau \to \infty} \mathcal{Z}_{\tau}^{\varepsilon} = z^{\varepsilon}$$

for fixed $\varepsilon > 0$, which follows from (2.38). Instead of (3.59), we use (3.66).

3.3.2 The delta function. Proof of Theorem 2.1.9

We now deal with the case $w = -\delta$. Let us first recall the definition (2.42) of w^{ε} . Let us note that since U is even, it is not necessary to take U to be non-positive, since we can argue as in [30, (5.33)] using |U| (note that in [30], one writes \tilde{w} for U). In what follows, we again denote objects corresponding to the interaction potential w^{ε} by using a superscript ε . Again, by following Section 3.2.6, to prove Theorem 2.1.9, it suffices to prove the following proposition.

Lemma 3.3.2. Let $w := -\delta$, and let w^{ε} be defined as in (2.42). Then there is a sequence (ε_{τ}) satisfying ε_{τ} converging to 0 as $\tau \to \infty$ such that

$$\lim_{\tau \to \infty} \rho_{\tau}^{\varepsilon_{\tau}} \left(\Theta_{\tau}(\xi) \right) = \rho \left(\Theta(\xi) \right) \,, \tag{3.67}$$

uniformly in $\xi \in C_p$, where C_p is given by (1.13).

Proof. As in the proof of Lemma 3.3.1, it suffices to prove for fixed ε that

$$\lim_{\tau \to \infty} \rho_{\tau}^{\varepsilon} \left(\Theta_{\tau}(\xi) \right) \to \rho^{\varepsilon} \left(\Theta\left(\xi\right) \right) \,, \tag{3.68}$$

uniformly in $\xi \in \mathcal{C}_p$, and

$$\lim_{\varepsilon \to 0} \rho^{\varepsilon} \left(\Theta \left(\xi \right) \right) \to \rho \left(\Theta \left(\xi \right) \right) \tag{3.69}$$

uniformly in $\xi \in C_p$. Since $w^{\varepsilon} \in L^{\infty}(\mathbb{T})$, (3.68) follows from Lemma 3.3.1. To prove (3.69), we note that Lemma 3.1.6 (2) now implies

$$|\mathcal{W}| \leq \frac{1}{2} \|U\|_{L^1} \|\varphi\|_{L^4}^4,$$

$$|\mathcal{W}^{\varepsilon}| \leq \frac{1}{2} \|U\|_{L^1} \|\varphi\|_{L^4}^4.$$
 (3.70)

Since $\int dx U = -1$ and U is even,

$$\mathcal{W}^{\varepsilon} - \mathcal{W} = \frac{1}{2} \int dx \, dy \, w^{\varepsilon}(x - y) \left(|\varphi(x)|^2 |\varphi(y)|^2 - |\varphi(x)|^4 \right) \,.$$

$$|\mathcal{W}^{\varepsilon} - \mathcal{W}| \le \frac{1}{2} \int dx \, dy \, |w^{\varepsilon}(x-y)| |\varphi(x)|^2 |\varphi(x) - \varphi(y)| \left(|\varphi(x)| + |\varphi(y)|\right). \quad (3.71)$$

We can then follow the argument in [30, (5.49) in the proof of Theorem 1.6.] to conclude $\mathcal{W}^{\varepsilon} \to \mathcal{W}$. We omit the details. Arguing as in the proof of Lemma 3.3.1, we obtain (3.69), and thus (3.67). We emphasise that, in order to apply the dominated convergence theorem as in the proof of Lemma 3.3.1, it is important that the upper bound (3.70) is uniform in ε .

Proof of Theorem 2.1.9. We obtain (2.43) by arguing analogously as for (2.39). Here, instead of Lemma 3.3.1, we use Lemma 3.3.2. The proof of (2.44) is analogous to that of (2.40).

3.4 Remarks about the cut-off function f

In this section, we expand on Remark 2.1.10(1) and (3).

3.4.1 Interaction potentials of positive type

For a bounded, real-valued, even interaction potential w of positive type (i.e. $\hat{w} \geq 0$ pointwise almost everywhere), we claim we can apply the methods used in the proof of [29, Theorem 1.8] to get the result of Theorem 2.1.7 for $\rho_{\#}$ defined without a truncation in \mathcal{N}_{\sharp} . To do this, we follow the convention from the two and three dimensional cases from [29] and consider a non-normal ordered quantum interaction, namely

$$\mathcal{W}_{\tau}' := \frac{1}{2} \int dx \, dy \, \varphi_{\tau}^*(x) \varphi_{\tau}(x) w(x-y) \varphi_{\tau}^*(y) \varphi_{\tau}(y) \,, \qquad (3.72)$$

which we note is different to the convention adopted in the rest of the cubic case. We also define $H'_{\tau} := H_{\tau,0} + W'_{\tau}$ in contrast to (2.30). Applying (2.26), we have

$$\mathcal{W}_{\tau}' = \mathcal{W}_{\tau} + \frac{1}{2\tau} w(0) \mathcal{N}_{\tau}^2 \,,$$

where we recall (2.29) and (2.31). We consider this non-normal ordered interaction since \mathcal{W}'_{τ} acts on the n^{th} sector of Fock space as multiplication by

$$\frac{1}{2\tau^2} \sum_{i,j=1}^n w(x_i - x_j).$$
(3.73)

 So

The key difference from (3.14) is (3.73) includes the diagonal terms of the sum. The remark follows from showing that if w is of positive type, (3.73) ≥ 0 almost everywhere, since we can apply Proposition 1.7.12 as in the proof of [29, Proposition 4.5]. We can further reduce this to showing (3.73) ≥ 0 for $w \in C^{\infty}$ of positive type by taking $w^{\varepsilon} := w * \varphi^{\varepsilon}$ for a standard approximation to the identity φ^{ε} of positive type, since then $w^{\varepsilon} \to w$ pointwise almost everywhere.

To see this, recall that for $g \in L^2$, Parseval's theorem implies

$$\langle g, w * g \rangle \sim \sum_{k \in \mathbb{Z}} |\hat{g}(k)|^2 \hat{w}(k) \ge 0$$
 (3.74)

since w is of positive type. Taking $g^{\varepsilon} \in C^{\infty}$ with $g^{\varepsilon} \to \sum_{j=1}^{n} \delta(\cdot - x_j)$ weakly with respect to continuous functions, for $w \in C^{\infty}$ we have, by (3.74)

$$0 \le \langle g^{\varepsilon}, w * g^{\varepsilon} \rangle \to \sum_{i,j=1}^{n} w(x_i - x_j).$$

Letting $\varepsilon \to 0$ then yields (3.73) ≥ 0 for w smooth of positive type.

3.4.2 General L^{∞} interaction potentials

For a general bounded, even, real-valued interaction potential w we show we could have used a Gaussian cut-off rather than a compactly supported one. Notice that since $w \in L^{\infty}$, there is some c such that $w^c := w + c \ge 0$ pointwise. Throughout this section, for an object $X_{\#}$, we use $X_{\#}^c$ to denote $X_{\#}$ defined using w^c rather than w. Notice that

$$\mathcal{W}_{\tau}^{\prime,c} = \mathcal{W}_{\tau}^{\prime} + \frac{c}{2} \mathcal{N}_{\tau}^2 \,, \qquad (3.75)$$

$$\mathcal{W}^c = \mathcal{W} + \frac{c}{2} \,\mathcal{N}^2 \,. \tag{3.76}$$

Applying an adapted form of [29, Theorem 1.8] for non-normal ordered interactions, we have

$$\lim_{\tau \to \infty} \frac{\operatorname{Tr} \left(\Theta_{\tau}(\xi) \mathrm{e}^{-H_{\tau}^{\prime,c}} \right)}{\operatorname{Tr} \left(\mathrm{e}^{-H_{\tau}^{\prime,c}} \right)} = \frac{\int d\mu \,\Theta(\xi) \mathrm{e}^{-H^{c}}}{\int d\mu \,\mathrm{e}^{-H^{c}}} \,. \tag{3.77}$$

We note that the adapted form of [29, Theorem 1.8] holds by applying the same proof, but using $\sum_{i,j=1}^{n} w^c(x_i - x_j) \ge 0$ instead of $\sum_{i,j=1,i\neq j}^{n} w^c(x_i - x_j) \ge 0$ in the proof of [29, Proposition 4.5].

Rewriting (3.77) using (3.75) and (3.76) gives

$$\lim_{\tau \to \infty} \frac{\operatorname{Tr}\left(\Theta_{\tau}(\xi) \mathrm{e}^{-H_{\tau}'} \mathrm{e}^{-\frac{c}{2}\mathcal{N}_{\tau}^{2}}\right)}{\operatorname{Tr}\left(\mathrm{e}^{-H_{\tau}'} \mathrm{e}^{-\frac{c}{2}\mathcal{N}_{\tau}^{2}}\right)} = \frac{\int d\mu \,\Theta(\xi) \mathrm{e}^{-H} \mathrm{e}^{-\frac{c}{2}\mathcal{N}^{2}}}{\int d\mu \,\mathrm{e}^{-H} \mathrm{e}^{-\frac{c}{2}\mathcal{N}^{2}}}$$

3.5 The time-dependent problem

In this section, we consider the time-dependent problem. The analysis for bounded w and the proof of Theorem 2.1.11 are given in Section 3.5.1. The case when w is unbounded is analysed in Section 3.5.2. Here, we prove Theorems 2.1.12 and 2.1.13. Throughout the section, we fix $p \in \mathbb{N}^*$ and $\xi \in \mathcal{L}(\mathfrak{h}^{(p)})$. In particular, we have the following two lemmas.

3.5.1 Bounded interaction potentials. Proof of Theorem 2.1.11.

In order to deal with bounded interaction potentials, we recall the *Schwinger-Dyson* expansion outlined in [30, Sections 3.2 and 3.3].

Lemma 3.5.1. Given $\mathcal{K} > 0$, $\varepsilon > 0$, and $t \in \mathbb{R}$, there exists $L = L(\mathcal{K}, \varepsilon, t, ||\xi||, p) \in \mathbb{N}$, a finite sequence $(e^l)_{l=0}^L$, with $e^l = e^l(\xi, t) \in \mathcal{L}(\mathfrak{h}^{(p)})$ and $\tau_0 = \tau_0(\mathcal{K}, \varepsilon, t, ||\xi||) > 0$ such that

$$\left\| \left(\Psi_{\tau}^{t} \Theta_{\tau}(\xi) - \sum_{l=0}^{L} \Theta_{\tau}(e^{l}) \right) \right\|_{\mathfrak{h}^{(\leq \mathcal{K}\tau)}} \right\| < \varepsilon \,,$$

for all $\tau \geq \tau_0$. Here we recall the definition of $\mathfrak{h}^{(\leq p)}$ from (3.41).

In other words, for large τ and restricted numbers of particles, we can approximate the evolution of the lift of an arbitrary operator with finitely many unevolved lifts. We also have the corresponding classical result.

Lemma 3.5.2. Given $\mathcal{K} > 0$, $\varepsilon > 0$, and $t \in \mathbb{R}$, then there exist $L = L(\mathcal{K}, \varepsilon, t, ||\xi||, p) \in \mathbb{N}$, $\tau_0 = \tau_0(\mathcal{K}, \varepsilon, t, ||\xi||) > 0$ both possibly larger than in Lemma 3.5.1, and for the same choice of $e^l = e^l(\xi, t)$ as in Lemma 3.5.1, we have

$$\left| \left(\Psi^t \Theta(\xi) - \sum_{l=0}^L \Theta(e^l) \right) \chi_{\{\mathcal{N} \le \mathcal{K}\}} \right| < \varepsilon \,,$$

for all $\tau \geq \tau_0$.

We note that the proofs of Lemmas 3.5.1 and 3.5.2, respectively [30, Lemmas 3.9 and 3.12], do not use the sign of the interaction potential, so still hold in our

case. The proofs of both results also require a compactly supported cut-off function, demonstrating the cut-off function of the form $f(x) = e^{-cx^2}$ discussed in Remark 2.1.10 (3) would not suffice here.

Proof of Theorem 2.1.11. By using Theorem 2.1.7, Lemmas 3.5.1-3.5.2, and following the proof of [30, Proposition 2.1], we obtain Theorem 2.1.11.

Remark 3.5.3. Recalling the proof of [30, Proposition 2.1], it follows that the convergence in Theorem 2.1.11 is uniform in the set

$$\{w \in L^{\infty}, m \in \mathbb{N}, t_i \in \mathbb{R}, p_i \in \mathbb{N} : \|w\|_{L^{\infty}}, |t_i|, p_i, \|\xi^i\|, m \le M\},\$$

for $i \in [1, \ldots, m]$ and for any fixed choice of M > 0.

3.5.2 Unbounded interaction potentials. Proofs of Theorems 2.1.12 and 2.1.13

Before proceeding, we need to prove a technical result concerning the flow of the NLS.

Lemma 3.5.4. Let $w \in L^1(\mathbb{T})$ and $s \geq \frac{3}{8}$ be given, and suppose $\varphi \in H^s$. Consider the Cauchy problem on \mathbb{T} given by

$$\begin{cases} i\partial_t u + (\Delta - \kappa)u = \left(w * |u|^2\right)u\\ u_0 = \varphi. \end{cases}$$
(3.78)

In addition, given $\varepsilon > 0$ and letting $w^{\varepsilon} \in L^{\infty}$ be a sequence satisfying $w^{\varepsilon} \to w$ in L^1 , we consider

$$\begin{cases} i\partial_t u^{\varepsilon} + (\Delta - \kappa)u^{\varepsilon} = \left(w^{\varepsilon} * |u^{\varepsilon}|^2\right)u^{\varepsilon} \\ u_0^{\varepsilon} = \varphi. \end{cases}$$
(3.79)

Since $s > 3/8 \ge 0$, the flow map defined in (2.5) is globally well defined. Denote by u and u^{ε} the solutions of (3.78) and (3.79) respectively. Then for T > 0

$$\lim_{\varepsilon \to 0} \|u - u^{\varepsilon}\|_{L^{\infty}_{[-T,T]}\mathfrak{h}} = 0.$$

In the following, we always take $b = \frac{1}{2} + \nu$, for $\nu > 0$ small.

Proof of Lemma 3.5.4. We recall the details of proof of [30, Proposition 5.1]. Firstly, we can take $\kappa = 0$ by considering $\tilde{u} := e^{i\kappa t}u$. We construct global mild solutions to (3.78) and (3.79) in the following way.
Let $\zeta, \psi : \mathbb{R} \to \mathbb{R}$ be smooth functions with

$$\zeta(t) = \begin{cases} 1 & \text{if } |t| \le 1\\ 0 & \text{if } |t| > 2. \end{cases}$$
(3.80)

$$\psi(t) = \begin{cases} 1 & \text{if } |t| \le 2\\ 0 & \text{if } |t| > 4. \end{cases}$$
(3.81)

We also define $\zeta_{\delta}(t) := \zeta(t/\delta)$ and $\psi_{\delta}(t) := \psi(t/\delta)$. We consider

$$(Lv)(\cdot,t) := \zeta_{\delta}(t) \mathrm{e}^{it\Delta} \varphi_0 - i\zeta_{\delta}(t) \int_0^t dt' \, \mathrm{e}^{i(t-t')\Delta} \left(w * |v_{\delta}|^2\right) v_{\delta}(t'), \tag{3.82}$$

$$(L^{\varepsilon}v)(\cdot,t) := \zeta_{\delta}(t)e^{it\Delta}\varphi_0 - i\zeta_{\delta}(t)\int_0^t dt' e^{i(t-t')\Delta} \left(w^{\varepsilon} * |v_{\delta}|^2\right)v_{\delta}(t'), \qquad (3.83)$$

where $v_{\delta}(x,t) := \psi_{\delta}(t)v(x,t)$. By proving L and L^{ε} are both contractions on appropriate function spaces for $\delta > 0$ sufficiently small, we are able to find local mild solutions to (3.78) and (3.79). The arguments used to prove (3.82) and (3.83) are contractions in [30, Proposition 5.1] still hold if we can show that

$$\left\| \left(w * |v_{\delta}|^{2} \right) v_{\delta} \right\|_{X^{0,b-1}} \lesssim \|w\|_{L^{1}} \|v_{\delta}\|_{X^{0,b}}^{3}.$$
(3.84)

To show (3.84), we define \mathcal{V}_{δ} as the function satisfying $\widetilde{\mathcal{V}}_{\delta} = |\tilde{v}_{\delta}|$. Note that by construction, $\|\mathcal{V}_{\delta}\|_{X^{0,b}} = \|v_{\delta}\|_{X^{0,b}}$. Then

To prove (3.84), it remains to show

$$\left\| |\mathcal{V}_{\delta}|^{2} \mathcal{V} \right\|_{X^{0,b-1}} \lesssim \|\mathcal{V}_{\delta}\|_{X^{0,b}}^{3} = \|v_{\delta}\|_{X^{0,b}}.$$
(3.85)

To show (3.85) we argue as in [79, (2.147)-(2.153)], where similar bounds are proved for the quintic case, and use a duality argument. Choose $c : \mathbb{Z} \times \mathbb{R} \to \mathbb{C}$ such that $\sum_{k} \int d\eta \, |c(k,\eta)|^2 = 1.$ We consider

$$I := \sum_{k} \int d\eta \, \left(1 + |\eta + k^2| \right)^{b-1} \left(|\mathcal{V}_{\delta}|^2 \mathcal{V}_{\delta} \right)^{\sim} (k, \eta) \, c(k, \eta) \, .$$

We have

$$|I| \leq \sum_{k} \sum_{k_1+k_2+k_3=k} \int_{\eta_1+\eta_2+\eta_3=\eta} d\eta_1 \, d\eta_2 \, d\eta_3 \, d\eta \, \frac{|c(k,\eta)|}{(1+|\eta+k^2|)^{1-b}} \\ \times |\tilde{v}_{\delta}(k_1,\eta_1)| |\tilde{v}_{\delta}(-k_2,-\eta_2)| |\tilde{v}_{\delta}(k_3,\eta_3)|.$$

Define

$$F(x,t) := \sum_{k} \int d\eta \, \frac{|c(k,\eta)|}{(1+|\eta+k^{2}|)^{1-b}} e^{2\pi i k x + 2\pi i t \eta},$$
$$G(x,t) := \sum_{k} \int d\eta \, |\tilde{v}_{\delta}(k,\eta)| e^{2\pi i k x + 2\pi i t \eta}.$$

Parseval's identity implies

$$I \lesssim \int \int dx \, dt \, F \overline{G} G \overline{G} = \left| \int \int dx \, dt \, F \overline{G} G \overline{G} \right|$$

$$\leq \|F\|_{L^4_{t,x}} \|G\|^3_{L^4_{t,x}}.$$
(3.86)

Since b > 3/8, we have the estimate $\|\phi\|_{L^4_{t,x}} \lesssim \|\phi\|_{X^{0,b}}$ (see [8, Proposition 2.6], [38, Lemma 2.1 (i)], and [83, Proposition 2.13]). So

$$\|F\|_{L^4_{t,x}} \lesssim \|F\|_{X^{0,3/8}} \le \|F\|_{X^{0,b-1}} = \|c\|_{\ell^2_k L^2_\eta} = 1.$$
(3.87)

Moreover

$$\|G\|_{L^4_{t,x}} \lesssim \|v_\delta\|_{X^{0,3/8}} \lesssim \delta^{\theta} \|v_\delta\|_{X^{0,b}}, \qquad (3.88)$$

where $\theta > 0$. Here the final inequality follows from [30, Lemma 5.3 (iv)]. Combining (3.86) with (3.87) and (3.88) yields (3.85).

So, for a time of existence δ that depends only on the L^2 norm of the initial data, we are able to construct local mild solutions, $v_{(n)}$ and $v_{(n)}^{\varepsilon}$ on $[n\delta, (n+1)\delta]$. We then piece these solutions together to create mild solutions u and u^{ε} to (3.78)

and (3.79) respectively. Using v and v^{ε} to denote $v_{(0)}$ and $v_{(0)}^{\varepsilon}$ respectively, we have

$$\begin{aligned} \|u - u^{\varepsilon}\|_{L^{\infty}_{[0,\delta]}L^{2}_{x}} &= \|v - v^{\varepsilon}\|_{L^{\infty}_{[0,\delta]}L^{2}_{x}} \\ &\leq \left\|\zeta_{\delta}(t)\int_{0}^{t}dt'\,\mathrm{e}^{i(t-t')\Delta}\left[(w - w^{\varepsilon})*|v_{\delta}(t')|^{2}\right]v_{\delta}(t')\right\|_{X^{0,b}} \\ &+ \left\|\zeta_{\delta}(t)\int_{0}^{t}dt'\,\mathrm{e}^{i(t-t')\Delta}\left[w^{\varepsilon}*\left(|v_{\delta}(t')|^{2} - |v^{\varepsilon}_{\delta}(t')|^{2}\right)\right]v_{\delta}(t')\right\|_{X^{0,b}} \\ &+ \left\|\zeta_{\delta}(t)\int_{0}^{t}dt'\,\mathrm{e}^{i(t-t')\Delta}\left[w^{\varepsilon}*|v^{\varepsilon}_{\delta}(t')|^{2}\right]\left(v_{\delta}(t') - v^{\varepsilon}_{\delta}(t')\right)\right\|_{X^{0,b}}. (3.89) \end{aligned}$$

For the first term of (3.89), we have

$$\begin{split} \left\| \zeta_{\delta}(t) \int_{0}^{t} dt' \, \mathrm{e}^{i(t-t')\Delta} \left[(w-w^{\varepsilon}) * |v_{\delta}(t')|^{2} \right] v_{\delta}(t') \right\|_{X^{0,b}} \\ & \leq C \delta^{\frac{1-2b}{2}} \left\| \left[(w-w^{\varepsilon}) * |v_{\delta}|^{2} \right] v_{\delta} \right\|_{X^{0,b-1}}, \end{split}$$

where the $\delta^{\frac{1-2b}{2}}$ comes from the estimates for local $X^{s,b}$ spaces proved in [43] and [44]. For a summary of these local $X^{s,b}$ spaces, we direct the reader to [30, Appendix A] and Lemma 1.7.6.

Arguing as in (3.84), we have

$$\left\| \left[(w - w^{\varepsilon}) * |v_{\delta}|^2 \right] v_{\delta} \right\|_{X^{0,b-1}} \le \|w - w^{\varepsilon}\|_{L^1} \|v_{\delta}\|_{X^{0,b}}^3 \to 0.$$

The bound on the second term in (3.89) follows by the same argument as in the proof of [30, Proposition 5.1], although we note that since $||w^{\varepsilon}||_{L^1}$ is only bounded rather than equal to 1, we may get a larger constant times a positive power of ε , which is not a problem. The third term in (3.89) then follows for the same reasons combined with [30, Proposition 5.1].

Following the remainder of the argument from [30, Proposition 5.1] and noting that there we gain no negative powers of ε , we have

$$\|u-u^{\varepsilon}\|_{L^{\infty}_{[0,T]}}\mathfrak{h}\to 0\,.$$

The corresponding negative time estimates follow from an analogous argument. \Box

We also have the corresponding result for the focusing local NLS.

Lemma 3.5.5. Let $s \geq \frac{3}{8}$ be given, and suppose $\varphi \in H^s(\mathbb{T})$. Consider the Cauchy

problem on \mathbb{T} given by

$$\begin{cases} i\partial_t u + (\Delta - \kappa)u = -|u|^2 u \\ u_0 = \varphi \,. \end{cases}$$
(3.90)

In addition, given $\varepsilon > 0$, let w^{ε} be as in (2.42). We consider

$$\begin{cases} i\partial_t u^{\varepsilon} + (\Delta - \kappa)u^{\varepsilon} = \left(w^{\varepsilon} * |u^{\varepsilon}|^2\right)u^{\varepsilon} \\ u_0^{\varepsilon} = \varphi. \end{cases}$$
(3.91)

Since $s > 3/8 \ge 0$, the flow map defined in (2.5) is globally well defined. Denote by u and u^{ε} the solutions of (3.90) and (3.91) respectively. Then for T > 0

$$\lim_{\varepsilon \to 0} \|u - u^{\varepsilon}\|_{L^{\infty}_{[-T,T]}\mathfrak{h}} = 0.$$

Proof. We can follow exactly the proof of [30, Proposition 5.1], recalling (2.41), and noting that the function w^{ε} defined in (2.42) is even and to deduce

$$\left||v_{\delta}(x)|^{2}-(w^{\varepsilon}*|v^{\delta}|^{2})(x)\right| \leq \int dy \left|w^{\varepsilon}(x-y)\right| \left|v_{\delta}(x)-v_{\delta}(y)\right| \left(|v_{\delta}(x)|+|v_{\delta}(y)|\right),$$

similarly as in [30, (5.27)]. We also have the same point about $||w^{\varepsilon}||_{L^1}$ not necessarily equal to 1 as in the proof of Lemma 3.5.4, which does not affect the argument. \Box

Before proving Theorem 2.1.12, we recall the following diagonalisation result, proved in [30, Lemma 5.5].

Proposition 3.5.6. Let $(Z_k)_{k\in\mathbb{N}}$ be an increasing sequence of sets in the sense that $Z_k \subset Z_{k+1}$ and put $Z := \bigcup_{k\in\mathbb{N}}Z_k$. For $\varepsilon, \tau > 0$, suppose that $g, g^{\varepsilon}, g^{\varepsilon}_{\tau} : Z \to \mathbb{C}$ are functions with the following properties.

- 1. For each fixed $k \in \mathbb{N}$ and $\varepsilon > 0$, $\lim_{\tau \to \infty} g_{\tau}^{\varepsilon}(\zeta) = g^{\varepsilon}(\zeta)$ uniformly in $\zeta \in Z_k$.
- 2. For each fixed $k \in \mathbb{N}$, $\lim_{\varepsilon \to 0} g^{\varepsilon}(\zeta) = g(\zeta)$ uniformly in $\zeta \in Z_k$.

Then there is a sequence (ε_{τ}) such that $\lim_{\tau\to\infty} = 0$ and

$$\lim_{\tau \to \infty} g_{\tau}^{\varepsilon_{\tau}}(\zeta) = g(\zeta)$$

for any $\zeta \in Z$.

Proof of Theorem 2.1.12. Throughout this proof we use X^{ε} or X to denote an object

defined using w^{ε} or w respectively. Define

$$Z := \{ (m, t_i, p_i, \xi^i) : m \in \mathbb{N}, |t_i| \in \mathbb{R}, p_i \in \mathbb{N}, \xi^i \in \mathcal{L}(\mathfrak{h}^{(p_i)}) \}$$
$$Z_k := \{ (m, t_i, p_i, \xi^i) : m \le k, |t_i| \le k, p_i \le k, \|\xi^i\| \le k \},$$

where $i \in [1, \ldots, m]$. We also define

$$g_{\#}^{\varepsilon}(\zeta) := \rho_{\#}^{\varepsilon} \left(\Psi_{\#}^{t_1,\varepsilon}(\Theta_{\#}(\xi^1)) \dots \Psi_{\#}^{t_m,\varepsilon}(\Theta_{\#}(\xi^m)) \right),$$
$$g(\zeta) := \rho \left(\Psi^{t_1}(\Theta(\xi^1)) \dots \Psi^{t_m}(\Theta(\xi^m)) \right).$$

Applying Theorem 2.1.11 and recalling Remark 3.5.3, Proposition 3.5.6 implies it suffices to show that for a fixed $k \in \mathbb{N}$

$$\lim_{\varepsilon \to 0} \rho^{\varepsilon} \left(\Psi^{t_1,\varepsilon} \Theta(\xi^1) \dots \Psi^{t_m,\varepsilon} \Theta(\xi^m) \right) = \rho \left(\Psi^{t_1} \Theta(\xi^1) \dots \Psi^{t_m} \Theta(\xi^m) \right), \tag{3.92}$$

uniformly in Z_k . Recalling (3.61), we have

$$\lim_{\varepsilon\to 0}\mathcal{W}^\varepsilon=\mathcal{W},$$

almost surely. Using Corollary 3.1.4 and the dominated convergence theorem, we have

$$\lim_{\varepsilon \to 0} \tilde{\rho}_1^{\varepsilon}(I) = \tilde{\rho}_1(I). \tag{3.93}$$

Here we recall

$$\tilde{\rho}_{\zeta}(X) := \int X e^{-\zeta \mathcal{W}} d\mu.$$

So by (3.24) and (3.93), to prove (3.92), it suffices to show

$$\lim_{\varepsilon \to 0} \tilde{\rho}_1^{\varepsilon} (\left(\Psi^{t_1,\varepsilon} \Theta(\xi^1) \dots \Psi^{t_m,\varepsilon} \Theta(\xi^m) \right)) = \tilde{\rho}_1 \left(\Psi^{t_1} \Theta(\xi^1) \dots \Psi^{t_m} \Theta(\xi^m) \right)$$
(3.94)

uniformly in Z_k .

Let S_t and S_t^{ε} be the flow maps for equations (3.78) and (3.79) respectively. Let $\varphi_0 \in H^{\frac{1}{2}-} \subset \mathfrak{h}$ be the classical free field defined in (2.12). Then for $\xi \in \mathcal{L}(\mathfrak{h}^{(k)})$, we can write

$$\Psi^{t,\varepsilon}\Theta(\xi) = \left\langle (S_t^{\varepsilon}\varphi_0)^{\otimes_k}, \xi (S_t^{\varepsilon}\varphi_0)^{\otimes_k} \right\rangle_{\mathfrak{h}^{\otimes_k}}, \quad \Psi^t\Theta(\xi) = \left\langle (S_t\varphi_0)^{\otimes_k}, \xi (S_t\varphi_0)^{\otimes_k} \right\rangle_{\mathfrak{h}^{\otimes_k}}.$$

Lemma 3.5.4 implies $(S_t^{\varepsilon})^{\otimes_k} \varphi_0 \to (S_t)^{\otimes_k} \varphi_0$ as $\varepsilon \to 0$. Moreover, $\xi \in \mathcal{L}(\mathfrak{h}^{(k)})$ implies

$$\lim_{\varepsilon \to 0} \Psi^{\varepsilon,t} \Theta(\xi) = \Psi^t \Theta(\xi),$$

uniformly in Z_k . Since $\mathcal{W}^{\varepsilon} \to \mathcal{W}$ as $\varepsilon \to 0$ almost surely, we have

$$\lim_{\varepsilon \to 0} \Psi^{t_1,\varepsilon} \Theta(\xi^1) \dots \Psi^{t_m,\varepsilon} \Theta(\xi^m) e^{-\mathcal{W}^{\varepsilon}} = \Psi^{t_1} \Theta(\xi^1) \dots \Psi^{t_m} \Theta(\xi^m) e^{-\mathcal{W}},$$
(3.95)

almost surely. Using conservation of mass for (3.78) - (3.79) and Lemma 3.1.6 (2), we have

$$\begin{aligned} \left| \Psi^{t_{1},\varepsilon} \Theta(\xi^{1}) \dots \Psi^{t_{m},\varepsilon} \Theta(\xi^{m}) \mathrm{e}^{-\mathcal{W}^{\varepsilon}} f(\mathcal{N}) \right| &\leq \prod_{j=1}^{m} \|\xi^{j}\| \|\varphi_{0}\|_{\mathfrak{h}}^{2p_{j}} \mathrm{e}^{\frac{1}{2} \|w^{\varepsilon}\|_{L^{1}} \|\varphi_{0}\|_{L^{4}}^{4}} f(\mathcal{N}), \\ \left| \Psi^{t_{1}} \Theta(\xi^{1}) \dots \Psi^{t_{m}} \Theta(\xi^{m}) \mathrm{e}^{-\mathcal{W}} f(\mathcal{N}) \right| &\leq \prod_{j=1}^{m} \|\xi^{j}\| \|\varphi_{0}\|_{\mathfrak{h}}^{2p_{j}} \mathrm{e}^{\frac{1}{2} \|w\|_{L^{1}} \|\varphi_{0}\|_{L^{4}}^{4}} f(\mathcal{N}). \end{aligned}$$

$$(3.96)$$

Using Lemma 3.1.1 and Assumption 2.1.4, both of the bounding functions in (3.96) are $L^1(d\mu)$. So (3.94) follows from (3.95) and the dominated convergence theorem.

Proof of Theorem 2.1.13. We follow the proof of Theorem 2.1.12, with the same definitions of $Z, Z_k, g_{\tau}^{\varepsilon}, g^{\varepsilon}, g$. Applying Theorem 2.1.12 and using Proposition 3.5.6, it suffices to show

$$\lim_{\varepsilon \to 0} \rho^{\varepsilon} \left(\Psi^{t_1,\varepsilon} \Theta(\xi^1) \dots \Psi^{t_m,\varepsilon} \Theta(\xi^m) \right) = \rho \left(\Psi^{t_1} \Theta(\xi^1) \dots \Psi^{t_m} \Theta(\xi^m) \right),$$

This follows by arguing analogously to the proof of Theorem 2.1.12, recalling that the fact that $\mathcal{W}^{\varepsilon} \to \mathcal{W}$ almost surely was shown in (3.71).

Chapter 4

Microscopic Derivation of the Gibbs Measure for the Quintic NLS

4.1 Analysis of the quintic Hartree equation (2.45)

In this section, we consider w as in Assumption 2.2.2 and study the Cauchy problem for the quintic Hartree equation (2.45).

$$\begin{cases} i\partial_t u + (\Delta - \kappa)u = \int dy \, dz \, w(x - y) \, w(y - z) \, w(z - x) \, |u(y)|^2 \, |u(z)|^2 \, u(x) \\ u|_{t=0} = u_0 \in H^s(\Lambda) \,. \end{cases}$$

$$(4.1)$$

4.1.1 Deterministic local well-posedness and invariance of the Gibbs measure

We first prove the following deterministic local well-posedness result, which should be viewed as an analogue of [8, Theorem 1] for the local quintic NLS.

Proposition 4.1.1 (Deterministic local existence for (4.1)). The Cauchy problem (4.1) is locally well-posed in $H^{s}(\Lambda)$ for s > 0.

We then prove the following probabilistic result, which should be viewed as an analogue of the result proved in [9] for the focusing local quintic NLS.

Proposition 4.1.2 (Invariance of truncated Gibbs measure and almost sure global existence for (4.1)). *The following claims hold.*

(i) Recall the probability space $(\mathbb{C}, \mathcal{G}, \mu)$ defined in (2.11), $\varphi \equiv \varphi^{\omega}$ defined in (2.12), and \mathcal{N} defined in (2.18). For B > 0 sufficiently small, we have

$$e^{-\frac{1}{3}\int dx\,dy\,dz\,w(x-y)\,w(y-z)\,w(x-z)\,|\varphi(x)|^2|\varphi(y)|^2|\varphi(z)|^2}\chi_{(\mathcal{N}\leq B)}\in L^1(d\mu)\,.$$

In particular, taking K = B in (2.54), we get that the probability measure $\mathbb{P}^{f}_{\text{Gibbs}}$ in (2.55) is well-defined.

(ii) Consider $s \in (0, \frac{1}{2})$. The measure $\mathbb{P}^{f}_{\text{Gibbs}}$ is invariant under the flow of (4.1). Furthermore, (4.1) admits global solutions for $\mathbb{P}^{f}_{\text{Gibbs}}$ -almost every $u_{0} \in H^{s}(\Lambda)$.

Before proving Propositions 4.1.1 and 4.1.2, we note several multilinear estimates.

Lemma 4.1.3. Consider $w_1, w_2, w_3 \in L^{\frac{3}{2}}(\Lambda)$. Given $q \in H^{\frac{1}{3}}(\Lambda)$, let

$$\mathcal{N}_1(q) := \int dx \, dy \, dz \, w_1(x-y) \, w_2(y-z) \, w_3(z-x) \, |q(x)|^2 \, |q(y)|^2 \, |q(z)|^2 \, .$$

Then, we have

$$|\mathcal{N}_1(q)| \lesssim ||w_1||_{L^{\frac{3}{2}}} ||w_2||_{L^{\frac{3}{2}}} ||w_3||_{L^{\frac{3}{2}}} ||q||_{H^{\frac{1}{3}}}^6.$$

Proof. By a density argument, we may assume without loss of generality that w_1, w_2, w_3, q are smooth functions, thus making the calculations that follow rigorous. Writing each integrand as a Fourier series, we compute

$$\mathcal{N}_{1}(q) = \sum_{k_{1},\dots,k_{6}} \sum_{\zeta_{1},\zeta_{2},\zeta_{3}} \int dx \, dy \, dz \, \widehat{w}_{1}(\zeta_{1}) \, \widehat{w}_{2}(\zeta_{2}) \, \widehat{w}_{3}(\zeta_{3}) \, \widehat{q}(k_{1}) \, \overline{\widehat{q}(k_{2})} \, \widehat{q}(k_{3}) \, \overline{\widehat{q}(k_{4})} \, \widehat{q}(k_{5}) \, \overline{\widehat{q}(k_{6})} \\ \times e^{2\pi i x(k_{1}-k_{2}+\zeta_{1}-\zeta_{3})} e^{2\pi i y(k_{3}-k_{4}-\zeta_{1}+\zeta_{2})} e^{2\pi i z(k_{5}-k_{6}-\zeta_{2}+\zeta_{3})}$$
(4.2)

The summations in (4.2) and in the sequel are taken over \mathbb{Z} . By integrating in x, y, z in (4.2) and taking absolute values, we deduce that

$$\begin{aligned} |\mathcal{N}_{1}(q)| &\leq \\ \sum_{k_{1},\dots,k_{6}} \sum_{\zeta_{1},\zeta_{2},\zeta_{3}} |\widehat{w}_{1}(\zeta_{1})| \, |\widehat{w}_{2}(\zeta_{2})| \, |\widehat{w}_{3}(\zeta_{3})| \, |\widehat{q}(k_{1})| \, |\widehat{q}(k_{2})| \, |\widehat{q}(k_{3})| \, |\widehat{q}(k_{4})| \, |\widehat{q}(k_{5})| \, |\widehat{q}(k_{6})| \\ &\times \, \delta(k_{1}-k_{2}+\zeta_{1}-\zeta_{3}) \, \delta(k_{3}-k_{4}-\zeta_{1}+\zeta_{2}) \, \delta(k_{5}-k_{6}-\zeta_{2}+\zeta_{3}) \,. \end{aligned}$$
(4.3)

By the constraints on the summands, we can rewrite the expression on the right-

hand side of (4.3) as

$$\sum_{k_1,\dots,k_6} \left\{ \sum_{\zeta_1} |\widehat{w}_1(\zeta_1)| |\widehat{w}_2(\zeta_1 - k_3 + k_4)| |\widehat{w}_3(\zeta_1 + k_1 - k_2)| \right\} \\ \times |\widehat{q}(k_1)| |\widehat{q}(k_2)| |\widehat{q}(k_3)| |\widehat{q}(k_4)| |\widehat{q}(k_5)| |\widehat{q}(k_6)| \\ \times \delta(k_1 - k_2 + k_3 - k_4 + k_5 - k_6), \quad (4.4)$$

which by applying Hölder's inequality in ζ_1 in the curly brackets in (4.4) is

$$\leq \|\widehat{w}_{1}\|_{L^{3}} \|\widehat{w}_{2}\|_{L^{3}} \|\widehat{w}_{3}\|_{L^{3}} \sum_{k_{1},\dots,k_{6}} |\widehat{q}(k_{1})| |\widehat{q}(k_{2})| |\widehat{q}(k_{3})| |\widehat{q}(k_{4})| |\widehat{q}(k_{5})| |\widehat{q}(k_{6})| \times \delta(k_{1}-k_{2}+k_{3}-k_{4}+k_{5}-k_{6}).$$
 (4.5)

By the Hausdorff-Young inequality and Parseval's identity, we have that

$$(4.5) \lesssim \|w_1\|_{L^{\frac{3}{2}}} \|w_2\|_{L^{\frac{3}{2}}} \|w_3\|_{L^{\frac{3}{2}}} \int dx \, |F(x)|^6 \,, \tag{4.6}$$

where F is chosen such that

$$\widehat{F} = |\widehat{q}|. \tag{4.7}$$

By applying Sobolev embedding, we deduce that

$$(4.6) = \|w_1\|_{L^{\frac{3}{2}}} \|w_2\|_{L^{\frac{3}{2}}} \|w_3\|_{L^{\frac{3}{2}}} \|F\|_{L^6}^6 \lesssim \|w_1\|_{L^{\frac{3}{2}}} \|w_2\|_{L^{\frac{3}{2}}} \|w_3\|_{L^{\frac{3}{2}}} \|F\|_{H^{\frac{1}{3}}}^6$$
$$= \|w_1\|_{L^{\frac{3}{2}}} \|w_2\|_{L^{\frac{3}{2}}} \|w_3\|_{L^{\frac{3}{2}}} \|q\|_{H^{\frac{1}{3}}}^6. \quad (4.8)$$

For the last equality in (4.8), we used that $||F||_{H^{\frac{1}{3}}} = ||q||_{H^{\frac{1}{3}}}$, which follows by (4.7).

We recall the definition of the local $X^{s,b}$ spaces from Definition 1.7.5.

Lemma 4.1.4. Consider $w_1, w_2, w_3 \in L^{\frac{3}{2}}(\Lambda)$. Given $s, \varepsilon > 0$ and $v_j \in X^{s,1/2-\varepsilon}$, for $j = 1, \ldots, 5$, we let

$$\mathcal{N}_{2}(v_{1}, v_{2}, v_{3}, v_{4}, v_{5})(x, t) := \int dy \, dz \, w_{1}(x - y) \, w_{2}(y - z) \, w_{3}(z - x) \, v_{1}(y, t) \, \overline{v_{2}(y, t)} \, v_{3}(z, t) \, \overline{v_{4}(z, t)} \, v_{5}(x, t) \,. \tag{4.9}$$

For $\varepsilon > 0$ sufficiently small, we have that for all $t_0 \in \mathbb{R}$ and $\delta > 0$ small

$$\left\|\mathcal{N}_{2}(v_{1}, v_{2}, v_{3}, v_{4}, v_{5})\right\|_{X_{[t_{0}, t_{0}+\delta]}^{s, -\frac{1}{2}+\varepsilon}} \lesssim_{s, \varepsilon} \delta^{\varepsilon} \left\|w_{1}\right\|_{L^{\frac{3}{2}}} \left\|w_{2}\right\|_{L^{\frac{3}{2}}} \left\|w_{3}\right\|_{L^{\frac{3}{2}}} \prod_{j=1}^{5} \left\|v_{j}\right\|_{X_{[t_{0}, t_{0}+\delta]}^{s, \frac{1}{2}+\varepsilon}}.$$

$$(4.10)$$

Proof. We first prove the following global version of the estimate (4.10).

$$\left\|\mathcal{N}_{2}(v_{1}, v_{2}, v_{3}, v_{4}, v_{5})\right\|_{X^{s, -\frac{1}{2}+\varepsilon}} \lesssim_{s, \varepsilon} \left\|w_{1}\right\|_{L^{\frac{3}{2}}} \left\|w_{2}\right\|_{L^{\frac{3}{2}}} \left\|w_{3}\right\|_{L^{\frac{3}{2}}} \prod_{j=1}^{5} \left\|v_{j}\right\|_{X^{s, \frac{1}{2}-\varepsilon}}.$$
 (4.11)

As in the proof of Lemma 4.1.3, it suffices to show (4.11) when the w_i , i = 1, 2, 3and v_j , j = 1, ..., 5 are smooth. Given $k \in \mathbb{Z}$ and $\eta \in \mathbb{R}$, we show that

$$\begin{aligned} \left| \left(\mathcal{N}_{2}(v_{1}, v_{2}, v_{3}, v_{4}, v_{5}) \right)^{\sim}(k, \eta) \right| &\lesssim \left\| w_{1} \right\|_{L^{\frac{3}{2}}} \left\| w_{2} \right\|_{L^{\frac{3}{2}}} \left\| w_{3} \right\|_{L^{\frac{3}{2}}} \\ &\times \sum_{k_{1}, \dots, k_{5}} \int d\eta_{1} \cdots d\eta_{5} \, \delta(k_{1} - k_{2} + k_{3} - k_{4} + k_{5} - k) \, \delta(\eta_{1} - \eta_{2} + \eta_{3} - \eta_{4} + \eta_{5} - \eta) \\ &\times \left| \widetilde{v}_{1}(k_{1}, \eta_{1}) \right| \left| \widetilde{v}_{2}(k_{2}, \eta_{2}) \right| \left| \widetilde{v}_{3}(k_{3}, \eta_{3}) \right| \left| \widetilde{v}_{4}(k_{4}, \eta_{4}) \right| \left| \widetilde{v}_{5}(k_{5}, \eta_{5}) \right|. \end{aligned}$$
(4.12)

Let us assume (4.12) for the moment. By Parseval's theorem for the spacetime Fourier transform, we note that the right-hand side of (4.12) can be written as

$$\|w_1\|_{L^{\frac{3}{2}}} \|w_2\|_{L^{\frac{3}{2}}} \|w_3\|_{L^{\frac{3}{2}}} (F_1F_2F_3F_4F_5)^{\sim}(k,\eta), \qquad (4.13)$$

where for j = 1, ..., 5, the function F_j is chosen such that

$$\widetilde{F}_j = |\widetilde{v}_j|. \tag{4.14}$$

Using Definition 1.7.5 we note that (4.12)-(4.13) imply that

$$\left\|\mathcal{N}_{2}(v_{1}, v_{2}, v_{3}, v_{4}, v_{5})\right\|_{X^{s, -\frac{1}{2}+\varepsilon}} \lesssim \left\|w_{1}\right\|_{L^{\frac{3}{2}}} \left\|w_{2}\right\|_{L^{\frac{3}{2}}} \left\|w_{3}\right\|_{L^{\frac{3}{2}}} \left\|F_{1}F_{2}F_{3}F_{4}F_{5}\right\|_{X^{s, -\frac{1}{2}+\varepsilon}}.$$
(4.15)

We now use the known quintic estimate

$$\|F_1 F_2 F_3 F_4 F_5\|_{X^{s,-\frac{1}{2}+\varepsilon}} \lesssim_{s,\varepsilon} \prod_{j=1}^5 \|F_j\|_{X^{s,\frac{1}{2}-\varepsilon}}, \qquad (4.16)$$

for $\varepsilon > 0$ sufficiently small. For a proof, see [26, Proof of (3.56)]. Strictly speaking, the claim [26, (3.56)] is stated for all F_j being equal, but the proof based on the

trilinear estimates [26, Lemma 3.29, Corollary 3.30] extends to the general case verbatim. We omit the details. By (4.14) and Definition 1.7.5, we have that for $j = 1, \ldots, 5$

$$\|F_{j}\|_{X^{s,-\frac{1}{2}+\varepsilon}} = \|v_{j}\|_{X^{s,-\frac{1}{2}+\varepsilon}}.$$
(4.17)

The claim (4.11) then follows from (4.15)-(4.17) (provided that we know (4.12)).

We now show (4.12). By expanding all of the integrands as a Fourier series and arguing analogously as in the proof of Lemma 4.1.3, we compute

$$\mathcal{N}_{2}(v_{1}, v_{2}, v_{3}, v_{4}, v_{5})(x, t) = \sum_{k_{1}, \dots, k_{5}} \sum_{\zeta_{1}, \zeta_{2}, \zeta_{3}} \int dy \, dz \, \int d\eta_{1} \cdots d\eta_{5} \, \widehat{w}_{1}(\zeta_{1}) \, \widehat{w}_{2}(\zeta_{2}) \, \widehat{w}_{3}(\zeta_{3}) \, \widetilde{v}_{1}(k_{1}, \eta_{1}) \, \overline{\widetilde{v}_{2}(k_{2}, \eta_{2})} \\ \times \, \widetilde{v}_{3}(k_{3}, \eta_{3}) \, \overline{\widetilde{v}_{4}(k_{4}, \eta_{4})} \, \widetilde{v}_{5}(k_{5}, \eta_{5}) \, \mathrm{e}^{2\pi \mathrm{i} x(\zeta_{1} - \zeta_{3} + k_{5})} \, \mathrm{e}^{2\pi \mathrm{i} y(-\zeta_{1} + \zeta_{2} + k_{1} - k_{2})} \\ \times \, \mathrm{e}^{2\pi \mathrm{i} z(-\zeta_{2} + \zeta_{3} + k_{3} - k_{4})} \, \mathrm{e}^{2\pi \mathrm{i} t(\eta_{1} - \eta_{2} + \eta_{3} - \eta_{4} + \eta_{5})} \,. \tag{4.18}$$

From (4.18), we hence deduce that

$$\begin{aligned} \left(\mathcal{N}_{2}(v_{1}, v_{2}, v_{3}, v_{4}, v_{5}) \right)^{\sim}(k, \eta) &= \\ & \sum_{k_{1}, \dots, k_{5}} \sum_{\zeta_{1}, \zeta_{2}, \zeta_{3}} \int d\eta_{1} \cdots d\eta_{5} \, \widehat{w}_{1}(\zeta_{1}) \, \widehat{w}_{2}(\zeta_{2}) \, \widehat{w}_{3}(\zeta_{3}) \, \widetilde{v}_{1}(k_{1}, \eta_{1}) \, \overline{\widetilde{v}_{2}(k_{2}, \eta_{2})} \\ & \times \, \widetilde{v}_{3}(k_{3}, \eta_{3}) \, \overline{\widetilde{v}_{4}(k_{4}, \eta_{4})} \, \widetilde{v}_{5}(k_{5}, \eta_{5}) \, \delta(\zeta_{1} - \zeta_{3} + k_{5} - k) \, \delta(-\zeta_{1} + \zeta_{2} + k_{1} - k_{2}) \\ & \times \, \delta(-\zeta_{2} + \zeta_{3} + k_{3} - k_{4}) \, \delta(\eta_{1} - \eta_{2} + \eta_{3} - \eta_{4} + \eta_{5} - \eta) \\ &= \sum_{k_{1}, \dots, k_{5}} \sum_{\zeta_{1}} \int d\eta_{1} \cdots d\eta_{5} \, \widehat{w}_{1}(\zeta_{1}) \, \widehat{w}_{2}(\zeta_{1} - k_{1} + k_{2}) \, \widehat{w}_{3}(\zeta_{1} + k_{5} - k) \\ & \times \, \widetilde{v}_{1}(k_{1}, \eta_{1}) \, \overline{\widetilde{v}_{2}(k_{2}, \eta_{2})} \, \widetilde{v}_{3}(k_{3}, \eta_{3}) \, \overline{\widetilde{v}_{4}(k_{4}, \eta_{4})} \, \widetilde{v}_{5}(k_{5}, \eta_{5}) \\ & \times \, \delta(k_{1} - k_{2} + k_{3} - k_{4} + k_{5} - k) \, \delta(\eta_{1} - \eta_{2} + \eta_{3} - \eta_{4} + \eta_{5} - \eta) \,. \end{aligned}$$

We now deduce (4.12) from (4.19) by arguing analogously as in the proof of Lemma 4.1.3. Hence, we obtain (4.11).

We now prove (4.10) when $t_0 = 0$. The general case follows by a suitable translation in time. We recall (1.21), use (4.11) for $V_j \in X^{s,b}$ such that $V_j|_{\mathbb{T}\times[0,\delta]} = v_j|_{\mathbb{T}\times[0,\delta]}$ for $j = 1, \ldots, 5$, take infima over V_j , and deduce the claim by using the

localisation property¹

$$\|v\|_{X^{s,b_1}_{[0,\delta]}} \lesssim_s \delta^{b_2 - b_1} \|v\|_{X^{s,b_2}_{[0,\delta]}}, \qquad (4.20)$$

for $-\frac{1}{2} < b_1 \le b_2 < \frac{1}{2}$. For a self-contained proof of (4.20), we refer the reader to [26, Lemma 3.11]; see also [83, Lemma 2.11]. The estimate (4.10) now follows.

Before the next proof, we recall the properties of the local $X^{s,b}$ spaces from Lemma 1.7.6.

Proof of Proposition 4.1.1. By replacing u with $e^{it\kappa}u$, we can reduce to considering the case² when $\kappa = 0$. We consider solutions for non-negative times. The argument for negative times is analogous. Throughout the proof, we consider $w_1 = w_2 = w_3 =$ w in (4.9). Therefore, the nonlinearity in (4.1) is equal to $\mathcal{N}_2(u, u, u, u, u)$. Let us fix $b = \frac{1}{2} + \varepsilon$ for $\varepsilon > 0$ sufficiently small. We consider the map

$$(Lv)(\cdot,t) := e^{it\Delta} u_0 - i \int_0^t dt' e^{i(t-t')\Delta} \mathcal{N}_2(v,v,v,v,v)(t').$$
(4.21)

In order to show the local existence of a solution, by the Banach fixed point theorem, it suffices to show that for suitable $\alpha > 0$ and $\delta \sim ||u_0||_{H^s}^{-\alpha}$ sufficiently small, the map L is a contraction on the ball

$$\mathcal{B} := \left\{ v \in X_{[0,\delta]}^{s,b}, \quad \|v\|_{X_{[0,\delta]}^{s,b}} \le \mathcal{M} \|u_0\|_{H^s} \right\}$$
(4.22)

in the Banach space $X_{[0,\delta]}^{s,b}$ for suitable $\mathcal{M} > 0$. By construction, each such fixed point u will be a mild solution of (4.1) on the time interval $[0,\delta]$, meaning that for all $t \in [0,\delta]$, we have

$$u(\cdot, t) = e^{it\Delta} u_0 - i \int_0^t dt' e^{i(t-t')\Delta} \mathcal{N}_2(u, u, u, u, u)(t').$$
 (4.23)

Furthermore, by Lemma 1.7.6 (i), it follows that $\|u\|_{L^{\infty}_{t\in[0,\delta]}H^s_x} \lesssim_{\mathcal{M},b} \|u_0\|_{H^s}$.

We now show that (4.21) is a contraction on (4.22). By using Lemma 1.7.6 (ii), (iii) and Lemma 4.1.4 with $v_1 = v_2 = v_3 = v_4 = v_5 = v$, we deduce that

$$\|Lv\|_{X^{s,b}_{[0,\delta]}} \le C_1 \|u_0\|_{H^s} + C_2 \,\delta^{\varepsilon} \|v\|^5_{X^{s,b}_{[0,\delta]}}, \qquad (4.24)$$

for suitable constants $C_1, C_2 > 0$. Note that C_1 is the implied constant in Lemma

¹In fact, this argument shows (4.10) with δ^{ε} replaced by $\delta^{5\varepsilon-}$, but we will not need this estimate in the sequel.

 $^{^2\}mathrm{Note}$ that this transformation does not change the H^s norm.

1.7.6 (ii). Similarly, we have

$$\|Lv^{(1)} - Lv^{(2)}\|_{X^{s,b}_{[0,\delta]}} \le C_3 \,\delta^{\varepsilon} (\|v^{(1)}\|^4_{X^{s,b}_{[0,\delta]}} + \|v^{(2)}\|^4_{X^{s,b}_{[0,\delta]}}) \|v^{(1)} - v^{(2)}\|_{X^{s,b}_{[0,\delta]}}, \quad (4.25)$$

for a suitable constant $C_3 > 0$. In order to deduce (4.25), we used Lemma 1.7.6 (iii), as well as the precise multilinear form of \mathcal{N}_2 given by (4.9), and Lemma 4.1.4 with v_j taking values $v^{(1)}, v^{(2)}$, or $v^{(1)} - v^{(2)}$. We have also used the elementary inequality $x^a y^b \leq C(x^{a+b} + y^{a+b})$ for x, y > 0 to control the cross terms. From (4.24)–(4.25), it follows that (4.21) is a contraction on (4.22) if we take $\mathcal{M} = 2C_1$ and $\delta \sim ||u_0||_{H^s}^{-\frac{4}{\varepsilon}}$ sufficiently small. More precisely, we choose $\delta > 0$ such that

$$C_2 \,\delta^{\varepsilon} \mathcal{M}^5 \|u_0\|_{H^s}^4 \le C_1 \,, \qquad 2C_3 \,\delta^{\varepsilon} \mathcal{M}^4 \|u_0\|_{H^s}^4 \le \frac{1}{2} \,.$$
 (4.26)

The above argument also shows the conditional uniqueness of mild solutions of (4.1) in (4.22). Namely, suppose that $u^{(1)}, u^{(2)} \in \mathcal{B}$ both satisfy (4.23) for $t \in [0, \delta]$. By repeating the earlier arguments, we deduce that

$$\begin{aligned} \|u^{(1)} - u^{(2)}\|_{X^{s,b}_{[0,\delta]}} &\leq C_3 \,\delta^{\varepsilon} (\|u^{(1)}\|^4_{X^{s,b}_{[0,\delta]}} + \|u^{(2)}\|^4_{X^{s,b}_{[0,\delta]}}) \|u^{(1)} - u^{(2)}\|_{X^{s,b}_{[0,\delta]}} \\ &\leq \frac{1}{2} \,\|u^{(1)} - u^{(2)}\|_{X^{s,b}_{[0,\delta]}} \,. \end{aligned}$$
(4.27)

For the second inequality in (4.27), we used the second condition in (4.26). From (4.27), it indeed follows that $u^{(1)} = u^{(2)}$. This concludes the proof.

Proof of Proposition 4.1.2. We first prove (i). The analysis is similar to that of [9, Lemma 3.10], which we recall in Appendix A.6. We follow the exposition in Appendix A.6 and explain the main differences. By using Proposition 4.1.3 with $w_1 = w_2 = w_3$ and $q = \varphi$, it follows that

$$\left| -\frac{1}{3} \int dx \, dy \, dz \, w(x-y) \, w(y-z) \, w(z-x) \, |\varphi(x)|^2 \, |\varphi(y)|^2 \, |\varphi(z)|^2 \right| \lesssim \|w\|_{L^{\frac{3}{2}}}^3 \, \|\varphi\|_{H^{\frac{1}{3}}}^6 \, .$$

$$(4.28)$$

The estimate (4.28) is the key reduction to the proof of [9, Lemma 3.10]. In particular, given $c_0 > 0$, it suffices to show that for B > 0 sufficiently small, depending on c_0 , we have

$$e^{c_0 \|\varphi\|_{H^{1/3}}^6} \chi(\mathcal{N} \le B) \in L^1(d\mu).$$
 (4.29)

By arguing analogously as for (A.11) (see also [9, (3.11)]), we reduce the proof of

(4.29) to showing that there exists c > 0 such that for large enough $\lambda > 0$, we have

$$\mu\left[\left\|\sum_{k\in\mathbb{Z}}\frac{\omega_k}{\sqrt{\lambda_k}}\,\mathrm{e}^{2\pi\mathrm{i}kx}\right\|_{H^{\frac{1}{3}}} > \lambda\,,\,\left(\sum_{k\in\mathbb{Z}}\frac{|\omega_k|^2}{\lambda_k}\right) \le B\right] \lesssim \exp\left(-cM_0^{4/3}\,\lambda^2\right),\qquad(4.30)$$

where

$$M_0 \sim \left(\frac{\lambda}{B}\right)^3. \tag{4.31}$$

Here, we recall (2.12). In other words, we reduce to the analysis from [9, Lemma 3.10] and Appendix A.6 with p = 6, with $\|\cdot\|_{L^6}$ norms replaced by $\|\cdot\|_{H^{\frac{1}{3}}}$ norms³ Since the norms are now defined in Fourier space, the analysis in fact simplifies. In particular, the analogue of (A.14), which is now given by

$$\left\| \sum_{|k| \sim M} a_k e^{2\pi i kx} \right\|_{H^{\frac{1}{3}}} \lesssim M^{\frac{1}{3}} \left\| \sum_{|k| \sim M} a_k e^{2\pi i kx} \right\|_{L^2},$$

follows by definition of $\|\cdot\|_{H^{\frac{1}{3}}}$. Here, M is a dyadic integer and $|k| \sim M$ means that $\frac{3M}{4} \leq |k| < \frac{3M}{2}$. Similarly, the analogue of (A.21) which is now given by

$$\left\|\sum_{|k|\sim M} \frac{\omega_k}{\sqrt{\lambda_k}} e^{2\pi i kx}\right\|_{H^{\frac{1}{3}}} \lesssim \frac{1}{M} \left\|\sum_{|k|\sim M} \omega_k e^{2\pi i kx}\right\|_{H^{\frac{1}{3}}},$$

follows by definition of $\|\cdot\|_{H^{\frac{1}{3}}}$.

With M_0 as in (4.31) and

$$\sigma_M \sim M^{-1/6} + (M_0/M)^{1/2}, \quad M > M_0,$$

as in (A.24), the above modifications allow us to deduce that there is $M > M_0$ a dyadic integer such that

$$\left\|\sum_{|k|\sim M} \omega_k \,\mathrm{e}^{2\pi \mathrm{i}kx}\right\|_{H^{\frac{1}{3}}} > \sigma_M M \lambda \,. \tag{4.32}$$

We now conclude the proof of (4.30) by using (4.32) and a union bound as in (A.25) – (A.30). In order to apply the above union bound argument, one needs to show

³These norms are linked by the Sobolev embedding $H^{\frac{1}{3}} \subset L^6$.

that, given a dyadic integer M, and

$$\mathcal{S} \equiv \mathcal{S}_M := \left\{ \sum_{|k| \sim M} a_k \, \mathrm{e}^{2\pi \mathrm{i} k x} \,, a_k \in \mathbb{C} \right\},\,$$

there exists a set $\Xi \equiv \Xi_M$ contained in the unit sphere of $H^{-\frac{1}{3}}$ satisfying the following properties.

- (1) $\max_{\phi \in \Xi} |\langle g, \phi \rangle| \ge \frac{1}{2} \|g\|_{H^{\frac{1}{3}}}$ for all $g \in \mathcal{S}$.
- (2) $\|\phi\|_{L^2} \lesssim M^{\frac{1}{3}}$ for all $\phi \in \Xi$.
- (3) $\log |\Xi| \lesssim M$.

The property (2) above follows from the frequency localisation of ϕ and the assumption that $\|\phi\|_{H^{-1/3}} = 1$. Properties (1) and (2) follow from the analogue of Lemma A.5.1 with the spaces L^p and $L^{p'}$ replaced with $H^{\frac{1}{3}}$ and $H^{-\frac{1}{3}}$ respectively. The modified claim follows from the same proof. We refer the reader to Appendix A.5 for the full details. Claim (i) now follows.

We now prove claim (ii). Once we have the local well-posedness given by Proposition 4.1.1 above and the tools used in the proof (most notably the multilinear estimate given by Lemma 4.1.4), the argument follows that of [9, Section 4], which corresponds to formally taking $w = \delta$ in (4.1). We outline only the main differences needed to consider the nonlocal problem (4.1). The main idea is to approximate (4.1) by a finite-dimensional system and to prove a suitable approximation result.

Step 1. Introducing the finite-dimensional system.

Given $N \in \mathbb{N}^*$, we denote by P_N the operator

$$P_N g(x) := \sum_{|k| \le N} \widehat{g}(k) e^{2\pi i k x},$$

i.e. the projection onto frequencies $|k| \leq N$. We then compare (4.33) with its finitedimensional truncation given by the following.

$$\begin{cases} i\partial_t u^N + (\Delta - \kappa)u^N = \\ P_N \left[\int dy \, dz \, w(x - y) \, w(y - z) \, w(z - x) \, |u^N(y)|^2 \, |u^N(z)|^2 \, u^N(x) \right] \\ u^N|_{t=0} = P_N u_0 \,. \end{cases}$$
(4.33)

Let us note that, in (4.33), we are applying $P_N(\cdot)$ in the x variable. We write the

solution u^N of the finite-dimensional system (4.33) as

$$u^{N}(x,t) = \sum_{|k| \le N} a_{k}(t) e^{2\pi i kx} .$$
(4.34)

In light of (4.34), we identity u^N with $a = (a_k)_{|k| \le N}$. With this identification, we can write (4.33) as a Hamiltonian system

$$\frac{da_k}{dt} = -i \frac{\partial H_N(a)}{\partial \bar{a}_k}, \quad |k| \le N, \qquad (4.35)$$

where the Hamiltonian is given by

$$H_N(a) = \sum_{|k| \le N} (4\pi^2 |k|^2 + \kappa) |a_k|^2 + \frac{1}{3} \int dx \, dy \, dz \, w(x - y) \, w(y - z) \, w(z - x)$$
$$\times \left| \sum_{|k| \le N} a_k \, \mathrm{e}^{2\pi \mathrm{i} k x} \right|^2 \left| \sum_{|k| \le N} a_k \, \mathrm{e}^{2\pi \mathrm{i} k y} \right|^2 \left| \sum_{|k| \le N} a_k \, \mathrm{e}^{2\pi \mathrm{i} k z} \right|^2 \quad (4.36)$$

and

$$\frac{\partial}{\partial \bar{a}_k} = \frac{1}{2} \left(\frac{\partial}{\partial \operatorname{Re} a_k} + \mathrm{i} \frac{\partial}{\partial \operatorname{Im} a_k} \right).$$
(4.37)

Let us note that (4.36) and (4.37) differ from the corresponding quantities used in [9]. This is because the convention of the Poisson structure in [9] amounts to taking $\{u(x), \bar{u}(y)\} = 2i\delta(x - y)$, which differs from (2.46) by a factor of 2. See Remark 4.1.5 below for more details.

Let us show (4.35) in detail. By (4.33) and (4.36), it suffices to show that for $|k| \leq N$, the k-th Fourier coefficient of

$$Q_N(a) := P_N \left[\int dy \, dz \, w(x-y) \, w(y-z) \, w(z-x) \right] \\ \times \left| \sum_{|k_2| \le N} a_{k_2} e^{2\pi i k_2 y} \right|^2 \left| \sum_{|k_3| \le N} a_{k_3} e^{2\pi i k_3 z} \right|^2 \sum_{|k_1| \le N} a_{k_1} e^{2\pi i k_1 x} \right]$$
(4.38)

equals $\frac{\partial W_N(a)}{\partial \bar{a}_k}$, where

$$W_N(a) := \frac{1}{3} \int dx \, dy \, dz \, w(x-y) \, w(y-z) \, w(z-x) \\ \times \left| \sum_{|k_1| \le N} a_{k_1} e^{2\pi i k_1 x} \right|^2 \left| \sum_{|k_2| \le N} a_{k_2} e^{2\pi i k_2 y} \right|^2 \left| \sum_{|k_3| \le N} a_{k_3} e^{2\pi i k_3 z} \right|^2.$$
(4.39)

(The terms corresponding to the kinetic energy are easily seen to be the same and we omit the proof).

Expanding the factors of w in (4.39) Fourier series, using $\frac{\partial}{\partial \bar{a}_k} \bar{a}_k = 1, \frac{\partial}{\partial \bar{a}_k} a_k = 0$, we compute for $|k| \leq N$

$$\frac{\partial W_N(a)}{\partial \bar{a}_k} = \sum_{\substack{k_1, \dots, k_5 \\ |k_j| \le N}} \sum_{\substack{\zeta_1, \zeta_2, \zeta_3}} \int dx \, dy \, dz \, \widehat{w}(\zeta_1) \, \widehat{w}(\zeta_2) \, \widehat{w}(\zeta_3) \, a_{k_1} \, a_{k_2} \, \bar{a}_{k_3} \, a_{k_4} \, \bar{a}_{k_5}
\times e^{2\pi i (\zeta_1 - \zeta_3 + k_1 - k)x} e^{2\pi i (-\zeta_1 + \zeta_2 + k_2 - k_3)y} e^{2\pi i (-\zeta_2 + \zeta_3 + k_4 - k_5)z} \chi_{|k| \le N}
= \sum_{\substack{k_1, \dots, k_5 \\ |k_j| \le N}} \sum_{\substack{\zeta_1, \zeta_2, \zeta_3 \\ |k_j| \le N}} \widehat{w}(\zeta_1) \, \widehat{w}(\zeta_2) \, \widehat{w}(\zeta_3) \, a_{k_1} a_{k_2} \bar{a}_{k_3} a_{k_4} \, \bar{a}_{k_5}
\times \, \delta(\zeta_1 - \zeta_3 + k_1 - k) \, \delta(-\zeta_1 + \zeta_2 + k_2 - k_3) \, \delta(-\zeta_2 + \zeta_3 + k_4 - k_5) \, \chi_{|k| \le N}. \quad (4.40)$$

By similar arguments, we can rewrite (4.38) as

$$Q_N(a) = P_N \left[\sum_{\substack{k_1, \dots, k_5 \\ |k_j| \le N}} \sum_{\substack{\zeta_1, \zeta_2, \zeta_3 \\ \zeta_1 \ }} \widehat{w}(\zeta_1) \, \widehat{w}(\zeta_2) \, \widehat{w}(\zeta_3) \, a_{k_1} a_{k_2} \bar{a}_{k_3} a_{k_4} \bar{a}_{k_5} \right] \\ \times \, \delta(-\zeta_1 + \zeta_2 + k_2 - k_3) \, \delta(-\zeta_2 + \zeta_3 + k_4 - k_5) \, \mathrm{e}^{2\pi \mathrm{i}(\zeta_1 - \zeta_3 + k_1)x} \left]. \quad (4.41)$$

We hence deduce (4.35) by noting that (4.40) is indeed equal to the k-th Fourier coefficient of (4.41).

By construction, the truncated Gibbs measure

$$d\mathbb{P}^{f}_{\text{Gibbs},N}(a) := \frac{1}{z^{f}_{\text{Gibbs},N}} e^{-H_{N}(a)} f\left(\sum_{|k| \le N} |a_{k}|^{2}\right) \prod_{|k| \le N} da_{k}, \qquad (4.42)$$

where

$$z_{\text{Gibbs},N}^{f} := \int \prod_{|k| \le N} da_{k} \, \mathrm{e}^{-H_{N}(a)} \, f\left(\sum_{|k| \le N} |a_{k}|^{2}\right) \tag{4.43}$$

is invariant under the finite-dimensional Hamiltonian flow (4.33). Here, we recall Assumption 2.2.4 and (4.36). The normalisation factor (4.43) is chosen in such that (4.42) is a probability measure. In order to deduce the invariance stated above, we used the fact that the flow (4.33) conserves mass. This is true because w is real-valued by Assumption 2.2.2.

Step 2. Approximation by the finite-dimensional system.

Having defined the finite-dimensional approximation (4.33) of (4.1), we want to compare the flow of the two. We prove an approximation result, which is an analogue of [9, Lemma 2.27] proved for the local quintic NLS. In order to state the claim, we need to introduce some notation. Suppose that $\psi \in C_c^{\infty}(\mathbb{R})$ is a function such that

$$\psi(\xi) = \begin{cases} 1 & \text{if } |\xi| \le \frac{1}{2}, \\ 0 & \text{if } |\xi| > 1. \end{cases}$$
(4.44)

With ψ as in (4.44), we define the following Fourier multiplier operators.

$$(R_N^- g)^{\hat{}}(k) := \psi\left(\frac{3k}{N}\right) \widehat{g}(k), \quad R_N^+ := \mathbf{1} - R_N^-.$$
 (4.45)

We note the following result, which corresponds to Bourgain's approximation lemma [9, Lemma 2.27].

Claim (*): Let A, T > 0 be given. Fix $u_0 \in H^s$ with

$$\|u_0\|_{H^s} \le A \,. \tag{4.46}$$

Consider for large N a solution u^N of (4.33) on [0,T] that satisfies

$$\sup_{t \in [0,T]} \|u^N(t)\|_{H^s} \le \mathcal{M}A, \qquad (4.47)$$

for some constant $\mathcal{M} > 0$ independent of N. Then the initial value problem (4.1) is well-posed on [0,T] and the following approximation bound holds for all $s_1 \in (0,s)$

$$\sup_{t \in [0,T]} \|u(t) - u^N(t)\|_{H^{s_1}} \le C(s, s_1, A, T, w, \mathcal{M}) \left(\|R_N^+ w\|_{L^{\frac{3}{2}}} + N^{s_1 - s} \right), \quad (4.48)$$

provided that the expression on the right-hand side of (4.48) is strictly less than 1.

The well-posedness of (4.1) stated above is interpreted in H^{s_1} for $s_1 \in (0, s)$. This claim follows immediately from (4.48) and the local well-posedness in H^{s_1} which we obtain from Proposition 4.1.1. We present the details of the proof of Claim (*) in Section 4.1.2 below.

Let us note that for $w \in L^{\frac{3}{2}}$, we have that

$$\lim_{N \to \infty} \|R_N^+ w\|_{L^{\frac{3}{2}}} = 0, \qquad (4.49)$$

hence (4.48) is an appropriate bound. In order to prove (4.49), we first recall (4.45)

and use Lemma 1.7.2 to deduce that

$$\|R_N^+\|_{L^{\frac{3}{2}} \to L^{\frac{3}{2}}} \lesssim 1, \qquad (4.50)$$

uniformly in N. Now, we note that (4.49) holds for all $w \in L^{\frac{3}{2}}$ whose Fourier transform is compactly supported. Namely, in this case, we have that $R_N^+w = 0$ for large enough N. Such w are dense in $L^{\frac{3}{2}}$, so we deduce that (4.49) holds on all of $L^{\frac{3}{2}}$ by using density and (4.50).

Step 3. Conclusion of the proof. Once we have the approximation result from Step 2, the proof follows identically to that given in [9, Sections 3–4]. We refer the reader to [9] for details. \Box

Remark 4.1.5. In our convention, we write the fields as

$$u(x) = \sum_{k} (p_k + \mathrm{i}q_k) \,\mathrm{e}^{2\pi \mathrm{i}kx}$$

where $p_k := \operatorname{Re} \widehat{u}(k), q_k := \operatorname{Im} \widehat{u}(k)$. We work with p_k and q_k as the canonical coordinates. Our convention for the Hamiltonian system is

$$\begin{cases} \dot{p}_k = \frac{1}{2} \frac{\partial H}{\partial q_k} \\ \dot{q}_k = -\frac{1}{2} \frac{\partial H}{\partial p_k} \end{cases}, \tag{4.51}$$

and our convention for the Poisson bracket is

$$\{g_1, g_2\} := \frac{1}{2} \sum_k \left(\frac{\partial g_1}{\partial q_k} \frac{\partial g_2}{\partial p_k} - \frac{\partial g_1}{\partial p_k} \frac{\partial g_2}{\partial q_k} \right).$$
(4.52)

We note that, with the Poisson bracket as in (4.52), we have

$$\{u(x), \bar{u}(y)\} = \frac{1}{2} \sum_{k} \left(i e^{2\pi i k x} e^{-2\pi i k y} - e^{2\pi i k x} (-i) e^{-2\pi i k y} \right) = i \sum_{k} e^{2\pi i k (x-y)} = i \delta(x-y).$$

Similarly, we have $\{u(x), u(y)\} = \{\bar{u}(x), \bar{u}(y)\} = 0$. Therefore, the Poisson bracket convention in (4.52) coincides with (2.46).

4.1.2 Proof of the approximation lemma from Step 2

In this section, we give the details of the proof of Claim (*) given in Step 2 of the proof of Proposition 4.1.2 (ii). In particular, we prove the estimate (4.48). Before

proceeding with the proof, we prove a multilinear estimate.

Lemma 4.1.6. Let us fix $b = \frac{1}{2} + \varepsilon$ for $\varepsilon > 0$ small and $s_1 > 0$. For $t_0 \in \mathbb{R}$, $\delta > 0$, and $v \in X_{[0,\delta]}^{s_1,b}$, and $\mathcal{N}_2(v,v,v,v,v,v)$ as in (4.9) with $w_1 = w_2 = w_3 = w$, we have

$$\begin{split} \|\mathcal{N}_{2}(v,v,v,v,v) - P_{N}\mathcal{N}_{2}(v,v,v,v,v)\|_{X^{s_{1},b-1}_{[t_{0},t_{0}+\delta]}} \\ &\lesssim \delta^{\varepsilon} \|w\|^{2}_{L^{\frac{3}{2}}} \|v\|^{4}_{X^{s_{1},b}_{[t_{0},t_{0}+\delta]}} \left(\|R^{+}_{N}w\|_{L^{\frac{3}{2}}} \|v\|_{X^{s_{1},b}_{[t_{0},t_{0}+\delta]}} + \|w\|_{L^{\frac{3}{2}}} \|R^{+}_{N}v\|_{X^{s_{1},b}_{[t_{0},t_{0}+\delta]}} \right) \end{split}$$

Here, we recall (4.45).

Proof. We write

$$\mathcal{N}_{2}(v, v, v, v, v) - P_{N}\mathcal{N}_{2}(v, v, v, v, v) = \int dy \, dz \left\{ w(x-y) \, w(x-z) \, v(x,t) - P_{N} \left[w(x-y) \, w(x-z) \, v(x,t) \right] \right\} \\ \times w(y-z) \, |v(y,t)|^{2} \, |v(z,t)|^{2} \,. \tag{4.53}$$

We write

$$w = R_N^+ w + R_N^- w, \quad v(\cdot, t) = R_N^+ v(\cdot, t) + R_N^- v(\cdot, t)$$
(4.54)

for all the terms appearing in the curly brackets in (4.53). By (4.44)–(4.45), it follows that

$$R_{N}^{-}w(x-y)R_{N}^{-}w(y-z)R_{N}^{-}v(x,t) - P_{N}\left[R_{N}^{-}w(x-y)R_{N}^{-}w(y-z)R_{N}^{-}v(x,t)\right] = 0.$$
(4.55)
The claim follows by combining (4.53)–(4.55) and Lemma 4.1.4.

The claim follows by combining (4.53)–(4.55) and Lemma 4.1.4.

Proof of Claim (*) from the proof of Proposition 4.1.2 (ii). By arguing analogously as in the proof of Proposition 4.1.1, we can reduce to the case when $\kappa = 0$. Let us note that the local well-posedness argument in the proof of Proposition 4.1.1 carries over immediately to the finite-dimensional approximation (4.33). This is because the operator P_N is a contraction on $X^{s,b}$. In particular, there exists $\delta \sim_A 1$ such that both (4.1) and (4.33) are well-posed on $[0, \delta]$ in H^{s_1} whenever the initial data is bounded in the H^{s_1} norm by $\mathcal{M}A + 1$. By construction, this is possible if we have

$$\delta^{\varepsilon} A^4 \ll 1. \tag{4.56}$$

By recalling (4.33), (4.46), and by using frequency localisation, we have that

$$||u(0) - u^{N}(0)||_{H^{s_{1}}} \lesssim N^{s_{1}-s} A.$$
(4.57)

Define $U := u - u^N$. Then, U solves the following difference equation.

$$\begin{cases} i\partial_t U + (\Delta - \kappa)U = \mathcal{N}_2(u, u, u, u, u) - P_N \mathcal{N}_2(u^N, u^N, u^N, u^N, u^N), \\ U|_{t=0} = u(0) - u^N(0). \end{cases}$$
(4.58)

Arguing analogously as in the proof of Proposition 4.1.1, but now in the context of (4.58) (instead of (4.1)), and using (4.47), it follows that there exists $T_0 \leq \delta$ depending only on A such that for all N, we have

$$\sup_{t \in [0,T_0]} \|u(t) - u^N(t)\|_{H^{s_1}} \lesssim \|u(0) - u^N(0)\|_{H^{s_1}}.$$
(4.59)

We note that the implied constant in (4.59) is independent of N. Let us fix $\sigma_0 \in (0, 1)$ arbitrarily small. From (4.57) and (4.59), we have that

$$\sup_{t \in [0,T_0]} \|u(t) - u^N(t)\|_{H^{s_1}} \le \sigma_0 \tag{4.60}$$

for all N large enough (depending on σ_0, A, s, s_1). For the remainder of the proof, we consider such N.

Combining (4.47) and (4.60), it follows that

$$\|u(T_0)\|_{H^{s_1}} \le \|u^N(T_0)\|_{H^{s_1}} + \sigma_0 \le \mathcal{M}A + 1.$$
(4.61)

We now introduce the following Cauchy problem.

$$\begin{cases} i\partial_t v^N + (\Delta - \kappa)v^N = \int dy \, dz \, w(x - y) \, w(y - z) \, w(z - x) \, |v^N(y)|^2 \, |v^N(z)|^2 \, v^N(x) \\ v^N|_{t=T_0} = u^N(T_0) \,. \end{cases}$$

$$(4.62)$$

Let us compare (4.62) with the flow of (4.1) started at $t = T_0$. Again arguing analogously as in the proof of Proposition 4.1.1, but now in the context of (4.62), it follows that we can take $\delta \sim_A 1$ possibly smaller satisfying (4.56) and obtain that (4.62) has a solution on the time interval $[T_0, T_0 + \delta]$ which satisfies

$$\sup_{t \in [T_0, T_0 + \delta]} \|u(t) - v^N(t)\|_{H^{s_1}} \le \mathcal{K} \|u(T_0) - v^N(T_0)\|_{H^{s_1}}$$
$$= \mathcal{K} \|u(T_0) - u^N(T_0)\|_{H^{s_1}} \le \mathcal{K}\sigma_0, \quad (4.63)$$

for some constant $\mathcal{K} > 0$. In (4.63), we recalled the initial condition in (4.62), as well as (4.60).

Let us now compare $v^N(t)$ and $u^N(t)$ for $t \in [T_0, T_0 + \delta]$. For the remainder of the proof, we always set $w_1 = w_2 = w_3 = w$ when working with the quantity \mathcal{N}_2 in (4.9). By (4.33), (4.62), and Duhamel's principle, we have that for all $t \in [T_0, T_0 + \delta]$

$$v^{N}(t) - u^{N}(t) = -i \int_{T_{0}}^{t} dt' e^{i(t-t')\Delta} \Gamma(\cdot, t'), \qquad (4.64)$$

where

$$\Gamma := \mathcal{N}_2(v^N, v^N, v^N, v^N, v^N) - P_N \mathcal{N}_2(u^N, u^N, u^N, u^N, u^N).$$
(4.65)

Using (4.64)–(4.65) and Lemma 1.7.6 (iii), we deduce that

$$\begin{aligned} \|v^{N} - u^{N}\|_{X_{[T_{0}, T_{0}+\delta]}^{s_{1}, b}} &\lesssim \|\Gamma\|_{X_{[T_{0}, T_{0}+\delta]}^{s_{1}, b-1}} \\ &\leq \left\|\mathcal{N}_{2}(v^{N}, v^{N}, v^{N}, v^{N}, v^{N}) - P_{N}\mathcal{N}_{2}(v^{N}, v^{N}, v^{N}, v^{N})\right\|_{X_{[T_{0}, T_{0}+\delta]}^{s_{1}, b-1}} \\ &+ \left\|P_{N}\mathcal{N}_{2}(v^{N}, v^{N}, v^{N}, v^{N}, v^{N}) - P_{N}\mathcal{N}_{2}(u^{N}, u^{N}, u^{N}, u^{N}, u^{N})\right\|_{X_{[T_{0}, T_{0}+\delta]}^{s_{1}, b-1}} \\ &=: I + II . \quad (4.66) \end{aligned}$$

We estimate the terms I and II in (4.66) separately.

Let us first estimate I. By Lemma 4.1.6, we have that

$$I \lesssim \delta^{\varepsilon} \|w\|_{L^{\frac{3}{2}}}^{2} \|v^{N}\|_{X^{s_{1},b}_{[T_{0},T_{0}+\delta]}}^{4} \left(\|R^{+}_{N}w\|_{L^{\frac{3}{2}}}\|v^{N}\|_{X^{s_{1},b}_{[T_{0},T_{0}+\delta]}} + \|w\|_{L^{\frac{3}{2}}}\|R^{+}_{N}v^{N}\|_{X^{s_{1},b}_{[T_{0},T_{0}+\delta]}}\right).$$

$$(4.67)$$

Let us note that

$$\|v^N\|_{X^{s,b}_{[T_0,T_0+\delta]}} \lesssim A$$
. (4.68)

In order to obtain (4.68), we note that for δ as in (4.56), we can argue as in the proof of Proposition 4.1.1 and obtain well-posedness of (4.62) on the time interval $[T_0, T_0 + \delta]$ since $\|v^N(T_0)\|_{H^s} = \|u^N(T_0)\|_{H^s} \leq \mathcal{M}A$ by (4.47). From (4.68), we obtain

$$\|v^N\|_{X^{s_1,b}_{[T_0,T_0+\delta]}} \lesssim A \tag{4.69}$$

and

$$\|R_N^+ v^N\|_{X^{s_1,b}_{[T_0,T_0+\delta]}} \lesssim N^{s_1-s} A.$$
(4.70)

In order to deduce (4.70) from (4.68), we recall (4.45) and use frequency localisation. Combining (4.56), (4.67), and (4.69)–(4.70), we obtain that

$$I \lesssim \|w\|_{L^{\frac{3}{2}}}^{2} A\left(\|R_{N}^{+}w\|_{L^{\frac{3}{2}}}^{2} + N^{s_{1}-s} \|w\|_{L^{\frac{3}{2}}}^{2}\right).$$

$$(4.71)$$

Let us now estimate II. Since P_N is a contraction on $X^{s_1,b-1}_{[T_0,T_0+\delta]}$, we have that

$$II \le \left\| \mathcal{N}_2(v^N, v^N, v^N, v^N, v^N) - \mathcal{N}_2(u^N, u^N, u^N, u^N, u^N) \right\|_{X^{s_1, b-1}_{[T_0, T_0 + \delta]}}.$$
 (4.72)

By recalling (4.9), using multilinearity, and using Lemma 4.1.4, it follows that the right-hand side of (4.72) is

$$\lesssim \delta^{\varepsilon} \|w\|_{L^{\frac{3}{2}}}^{3} \|u^{N} - v^{N}\|_{X^{s_{1},b}_{[T_{0},T_{0}+\delta]}} \left(\|u^{N}\|_{X^{s_{1},b}_{[T_{0},T_{0}+\delta]}}^{4} + \|v^{N}\|_{X^{s_{1},b}_{[T_{0},T_{0}+\delta]}}^{4} \right).$$
(4.73)

By arguing analogously as for (4.69), we have

$$\|u^N\|_{X^{s_1,b}_{[T_0,T_0+\delta]}} \lesssim A.$$
(4.74)

Using (4.69), and (4.74) in (4.73), it follows that

$$II \lesssim \delta^{\varepsilon} A^{4} \|w\|_{L^{\frac{3}{2}}}^{3} \|u^{N} - v^{N}\|_{X^{s_{1},b}_{[T_{0},T_{0}+\delta]}}.$$
(4.75)

We combine (4.66), (4.71), (4.75), and choose $\delta \sim_A 1$ possibly smaller satisfying (4.56) to deduce that

$$\|v^{N} - u^{N}\|_{X^{s_{1},b}_{[T_{0},T_{0}+\delta]}} \lesssim \|w\|^{2}_{L^{\frac{3}{2}}} A\Big(\|R^{+}_{N}w\|_{L^{\frac{3}{2}}} + N^{s_{1}-s} \|w\|_{L^{\frac{3}{2}}}\Big).$$
(4.76)

Combining (4.76) and Lemma 1.7.6 (i), it follows that

$$\sup_{t \in [T_0, T_0 + \delta]} \| v^N(t) - u^N(t) \|_{H^s} \lesssim \| w \|_{L^{\frac{3}{2}}}^2 A \Big(\| R_N^+ w \|_{L^{\frac{3}{2}}} + N^{s_1 - s} \| w \|_{L^{\frac{3}{2}}} \Big).$$
(4.77)

Using (4.63) and (4.77), it follows that

$$\sup_{t \in [T_0, T_0 + \delta]} \|u(t) - u^N(t)\|_{H^s} \le C_0 \|w\|_{L^{\frac{3}{2}}}^2 A\Big(\|R_N^+ w\|_{L^{\frac{3}{2}}} + N^{s_1 - s} \|w\|_{L^{\frac{3}{2}}}\Big) + \mathcal{K}\sigma_0 =: \sigma_1.$$
(4.78)

for some constant C_0 (which depends on s and s_1 , but we suppress this dependence here).

We now iterate this construction. Namely, we start from the time interval $[T_0 + (j-1)\delta, T_0 + j\delta]$ on which we have

$$\sup_{t \in [T_0 + (j-1)\delta, T_0 + j\delta]} \|u(t) - u^N(t)\|_{H^s} \le \sigma_j$$

and use the above arguments to deduce that

$$\sup_{t \in [T_0 + j\delta, T_0 + (j+1)\delta]} \|u(t) - u^N(t)\|_{H^s} \le \sigma_{j+1} \,,$$

where

$$\sigma_{j+1} := C_0 \|w\|_{L^{\frac{3}{2}}}^2 A\Big(\|R_N^+ w\|_{L^{\frac{3}{2}}} + N^{s_1 - s} \|w\|_{L^{\frac{3}{2}}}\Big) + \mathcal{K}\sigma_j.$$
(4.79)

The iteration step is possible provided that

$$\sigma_j \le 1. \tag{4.80}$$

(Recall (4.60) and (4.61) above).

From (4.79), by recalling (4.49), it follows that (4.80) holds for all $j \leq \lceil T/\delta \rceil$ provided that σ_0 is chosen sufficiently small and provided that N is chosen large enough. Note that

$$[0,T] \subset [0,T_0] \cup \bigcup_{j=1}^{\lceil T/\delta \rceil} [T_0 + (j-1)\delta, T_0 + j\delta] \,.$$

The claim then follows.

4.2 The time-independent problem with bounded interaction potentials. Proof of Theorem 2.2.6

In this section, we consider w as in Assumption 2.2.1 above. Our goal is to prove Theorem 2.2.6. As a preliminary step in the analysis, we expand the quantum and classical states into a power series. As in the cubic case, we note that, due to the presence of the cut-off, the resulting series are analytic in the complex plane. The precise series in the quantum and classical setting are respectively given in (4.83) and (4.88) below. In Section 4.2.1, we state several estimates that will be used in the analysis. In Section 4.2.2, we explicitly compute the expansion for the quantum and classical states mentioned above. In Sections 4.2.3 and 4.2.4, we prove bounds on the explicit and remainder terms of the resulting series. In Section 4.2.5, we comment on how to use this to prove Theorem 2.2.6, provided that we have convergence of the untruncated explicit terms given by Proposition 4.2.13. We prove Proposition 4.2.13 by using graphical methods in Section 4.2.6.

4.2.1 Basic estimates

Throughout this section, we fix $p \in \mathbb{N}^*$ and take $\xi \in \mathcal{L}(\mathfrak{h}^{(p)})$ unless stated otherwise. We have the following estimates on the quantities $\Theta_{\tau}(\xi)$ and $\Theta(\xi)$ defined in (2.72) and (2.59) respectively. We note that Lemmas 4.2.1 and 4.2.2 are also stated in the cubic case, however we state them again to be self-contained.

Lemma 4.2.1. For any $n \in \mathbb{N}^*$, we have

$$\left\|\Theta_{\tau}(\xi)\right|_{\mathfrak{h}^{(n)}}\right\| \leq \left(\frac{n}{\tau}\right)^{p} \|\xi\|.$$

Lemma 4.2.2. We have

$$|\Theta(\xi)| \le \|\varphi\|_{\mathfrak{h}}^{2p} \|\xi\|.$$

Lemma 4.2.1 follows from [45, (3.88)], and Lemma 4.2.2 is a consequence of (2.59). We also have the following estimates on the classical interaction, which follows immediately from Hölder's inequality.

Lemma 4.2.3. Suppose that the classical interaction \mathcal{W} is defined as in (2.51). Then for $w \in L^{\infty}(\Lambda)$, we have

$$|\mathcal{W}| \leq \frac{1}{3} \|w\|_{L^\infty}^3 \|\varphi\|_{L^2}^6$$

We also collect some estimates about Schatten space operators. The following result follows from the spectral decomposition of $|\mathcal{A}| = \sqrt{\mathcal{A}^* \mathcal{A}}$.

Lemma 4.2.4. Let \mathcal{H} be a separable Hilbert space. Suppose $1 \leq p_1 \leq p_2 \leq \infty$ and $\mathcal{A} \in \mathfrak{S}^{p_1}(\mathcal{H}) \cap \mathfrak{S}^{p_2}(\mathcal{H})$. Then

$$\|\mathcal{A}\|_{\mathfrak{S}^{p_2}(\mathcal{H})} \leq \|\mathcal{A}\|_{\mathfrak{S}^{p_1}(\mathcal{H})}.$$

We also recall Lemma 3.1.5, which is Hölder's inequality for Schatten spaces.

4.2.2 Power series expansions of the classical and quantum states

In this section, we compute the power series for the quantum and classical states. Let us recall (2.68) and (2.72). We note the following identities.

$$\rho_{\tau}(\Theta_{\tau}(\xi)) = \frac{\tilde{\rho}_{\tau,1}(\Theta_{\tau}(\xi))}{\tilde{\rho}_{\tau,1}(\mathbf{1})},\tag{4.81}$$

where for $z \in \mathbb{C}$, we let

$$\tilde{\rho}_{\tau,z}(\mathcal{A}) := \frac{1}{Z_{\tau,0}} \operatorname{Tr} \left(\mathcal{A} \mathrm{e}^{-H_{\tau,0} - z \mathcal{W}_{\tau}} f(\mathcal{N}_{\tau}) \right), \qquad (4.82)$$

and recall 1 is the identity operator on \mathcal{F} . Let us also define

$$F^{\xi}_{\tau}(z) := \tilde{\rho}_{\tau,z}(\Theta_{\tau}(\xi)). \tag{4.83}$$

Let us note the following result.

Lemma 4.2.5. For $M \in \mathbb{N}^*$, we have $F_{\tau}^{\xi}(z) = \sum_{m=0}^{M-1} a_{\tau,m}^{\xi} z^m + R_{\tau,M}^{\xi}(z)$. Here,

$$a_{\tau,m}^{\xi} := \frac{1}{Z_{\tau,0}} \operatorname{Tr} \left((-1)^m \int_0^1 dt_1 \int_0^{t_1} dt_2 \dots \int_0^{t_{m-1}} dt_m \,\Theta_\tau(\xi) \mathrm{e}^{-(1-t_1)H_{\tau,0}} \mathcal{W}_\tau \right. \\ \times \,\mathrm{e}^{-(t_1-t_2)H_{\tau,0}} \dots \mathcal{W}_\tau \mathrm{e}^{-(t_{m-1}-t_m)H_{\tau,0}} \mathcal{W}_\tau \mathrm{e}^{-t_m H_{\tau,0}} f(\mathcal{N}_\tau) \right), \quad (4.84)$$

and

$$R_{\tau,M}^{\xi}(z) := \frac{1}{Z_{\tau,0}} \operatorname{Tr}\left((-z)^{M} \int_{0}^{1} dt_{1} \int_{0}^{t_{1}} dt_{2} \dots \int_{0}^{t_{m-1}} dt_{M} \Theta_{\tau}(\xi) \mathrm{e}^{-(1-t_{1})H_{\tau,0}} \mathcal{W}_{\tau} \right. \\ \times \mathrm{e}^{-(t_{1}-t_{2})H_{\tau,0}} \dots \mathcal{W}_{\tau} \mathrm{e}^{-(t_{M-1}-t_{M})H_{\tau,0}} \mathcal{W}_{\tau} \mathrm{e}^{-t_{m}(H_{\tau,0}+z\mathcal{W}_{\tau})} f(\mathcal{N}_{\tau}) \right).$$
(4.85)

Proof. By performing a Duhamel expansion up to order M, we obtain the following result. More precisely, we use the identity

$$e^{X+Y} = e^X + \int_0^1 dt \, e^{(1-t)X} \, Y \, e^{X+tY}$$

M times in (4.82)–(4.83).

As before, given $g : \mathbb{C} \to \mathbb{C}$, and any operator $\mathcal{A} : \mathcal{F} \to \mathcal{F}$, that commutes with \mathcal{N}_{τ} , we note that \mathcal{A} also commutes with $g(\mathcal{N}_{\tau})$. This is because the operator $g(\mathcal{N}_{\tau})$ acts on the n^{th} sector of Fock space as multiplication by $\frac{n}{\tau}$. In particular, it follows that that, for every $\alpha > 0$, $f^{\alpha}(\mathcal{N}_{\tau})$ commutes with any operators of the form \mathcal{W}_{τ} , $e^{-tH_{\tau,0}}$, $e^{-t(H_{\tau,0}+z\mathcal{W}_{\tau})}$, occurring as factors in the integrands in (4.84)–(4.85) above. We use this fact without further mention in the sequel.

Let us recall (2.57) and (2.59). By analogy with (4.81), we rewrite the classical state as $\tilde{z}_{1}(Q(t))$

$$\rho(\Theta(\xi)) = \frac{\tilde{\rho}_1(\Theta(\xi))}{\tilde{\rho}_1(\mathbf{1})}, \qquad (4.86)$$

where for a random variable X and $z \in \mathbb{C}, \, \tilde{\rho}_z$ is defined as

$$\tilde{\rho}_z(X) := \int d\mu \, X \mathrm{e}^{-z\mathcal{W}} f(\mathcal{N}) \,. \tag{4.87}$$

For $z \in \mathbb{C}$, we define

$$F^{\xi}(z) := \tilde{\rho}_z(\Theta(\xi)) \,. \tag{4.88}$$

Then we have the following analogue of Lemma 4.2.5.

Lemma 4.2.6. For $M \in \mathbb{N}$, we have $F^{\xi}(z) = \sum_{m=0}^{M-1} a_m^{\xi} z^m + R_M^{\xi}(z)$. Here

$$a_m^{\xi} := \frac{(-1)^m}{m!} \int d\mu \,\Theta(\xi) \mathcal{W}^m f(\mathcal{N}) \,, \tag{4.89}$$

and

$$R_M^{\xi}(z) := \frac{(-z)^M}{M!} \int d\mu \,\Theta(\xi) \mathcal{W}^M \mathrm{e}^{-\tilde{z}\mathcal{W}} f(\mathcal{N}) \,, \tag{4.90}$$

for some $\tilde{z} \in [0, z]$.

4.2.3 Analysis of the quantum series (4.83)

In this section we prove that the explicit and remainder terms defined in (4.84) and (4.85) satisfy sufficient bounds for (4.83) to be analytic.

Lemma 4.2.7. For any $m \in \mathbb{N}$ and $a_{\tau,m}^{\xi}$ defined as in (4.84), we have

$$|a_{\tau,m}^{\xi}| \le \frac{(K^3 \|w\|_{L^{\infty}}^3)^m K^p \|\xi\|}{3^m m!}.$$
(4.91)

Proof. We argue similarly to the proof of Lemma 3.2.2. Lemma 3.1.5 implies

$$|a_{\tau,m}^{\xi}| \leq \frac{1}{Z_{\tau,0}} \int_{0}^{1} dt_{1} \int_{0}^{t_{1}} dt_{2} \dots \int_{0}^{t_{m-1}} dt_{m} \left\| \Theta_{\tau}(\xi) f^{\frac{1}{m+1}}(\mathcal{N}_{\tau}) \right\|_{\mathfrak{S}^{\infty}(\mathfrak{h})} \\ \times \left\| \mathcal{W}_{\tau} f^{\frac{1}{m+1}}(\mathcal{N}_{\tau}) \right\|_{\mathfrak{S}^{\infty}(\mathfrak{h})}^{m} \prod_{j=0}^{m} \left\| e^{-(t_{j}-t_{j+1})H_{\tau,0}} \right\|_{\mathfrak{S}^{\frac{1}{t_{j}-t_{j+1}}}(\mathfrak{h})}, \quad (4.92)$$

where we take the convention $t_0 := 1$ and $t_{m+1} := 0$. Noting that

$$||e^{-sH_{\tau,0}}||_{\mathfrak{S}^{1/s}(\mathfrak{h})} = (Z_{\tau,0})^s,$$

since $e^{-sH_{\tau,0}}$ is a positive operator, it follows from (4.92) that

$$|a_{\tau,m}^{\xi}| \leq \frac{1}{m!} \left\| \Theta_{\tau}(\xi) f^{\frac{1}{m+1}}(\mathcal{N}_{\tau}) \right\|_{\mathfrak{S}^{\infty}(\mathfrak{h})} \left\| \mathcal{W}_{\tau} f^{\frac{1}{m+1}}(\mathcal{N}_{\tau}) \right\|_{\mathfrak{S}^{\infty}(\mathfrak{h})}^{m} .$$
(4.93)

Noting that \mathcal{W}_{τ} acts on $\mathfrak{h}^{(n)}$ as multiplication by

$$\frac{1}{3\tau^3} \sum_{\substack{i,j,k\\i\neq j\neq k\neq i}}^n w(x_i - x_j) w(x_j - x_k) w(x_k - x_i) \,,$$

and recalling Lemma 4.2.1, we have

$$\left\| \mathcal{W}_{\tau} f^{\frac{1}{m+1}}(\mathcal{N}_{\tau}) \right\|_{\mathfrak{S}^{\infty}(\mathfrak{h}^{(n)})} \leq \frac{1}{3} \left(\frac{n}{\tau}\right)^{3} \|w\|_{L^{\infty}}^{3} \left| f\left(\frac{n}{\tau}\right) \right| \leq \frac{1}{3} K^{3} \|w\|_{L^{\infty}}^{3}.$$
(4.94)

Here we have also used the support properties of f as in Assumption 2.2.4, namely (2.54), and that $||f||_{L^{\infty}} \leq 1$. Using Lemma 4.2.1 and Assumption 2.2.4, we have

$$\left\|\Theta_{\tau}(\xi)f^{\frac{1}{m+1}}(\mathcal{N}_{\tau})\right\|_{\mathfrak{S}^{\infty}(\mathfrak{h})} \le K^{p}\|\xi\|.$$
(4.95)

Then (4.91) follows from (4.93), (4.94), and (4.95).

In order to estimate $R_{\tau,M}^{\xi}(z)$, we apply the Feynman-Kac formula, which we recall from Proposition 1.7.12.

Lemma 4.2.8. For any $M \in \mathbb{N}$ and $R_{\tau,M}^{\xi}(z)$ defined as in (4.85), we have

$$\left| R_{\tau,M}^{\xi}(z) \right| \le e^{\frac{1}{3}K^3 |\operatorname{Re}(z)| \|w\|_{L^{\infty}}^3} \frac{(K^3 \|w\|_{L^{\infty}}^3)^M K^p \|\xi\|}{3^M M!} |z|^M.$$
(4.96)

Proof. By arguing similarly as the proof of Lemma 3.2.3 and (3.23), it suffices to show that for $t \in [0, 1]$, we have

$$\left| \left(e^{-t(H_{\tau,0} + z\mathcal{W}_{\tau})} f^{\frac{1}{2}}(\mathcal{N}_{\tau}) \right)^{(n)}(\mathbf{x}; \mathbf{y}) \right| \le e^{\frac{1}{3}K^{3} |\operatorname{Re}(z)| \|\mathbf{w}\|_{L^{\infty}}^{3}} \left(e^{-tH_{\tau,0}} \right)^{(n)}(\mathbf{x}; \mathbf{y}), \quad (4.97)$$

where $\mathcal{A}^{(n)}$ denotes the kernel of \mathcal{A} restricted to the n^{th} sector of Fock space. Noting that

$$\left(\mathcal{W}_{\tau}\right)^{(n)}(\mathbf{x};\mathbf{y}) = \frac{1}{3\tau^3} \sum_{\substack{i,j,k\\i\neq j\neq k\neq i}}^n w(x_i - x_j)w(x_j - x_k)w(x_k - x_i) \prod_{l=1}^n \delta(x_l - y_l),$$

we can rewrite $\left(e^{-t(H_{\tau,0}+z\mathcal{W}_{\tau})}f^{\frac{1}{2}}(\mathcal{N}_{\tau})\right)^{(n)}(\mathbf{x};\mathbf{y})$ using Proposition 1.7.12 as

$$\int \mathbb{W}_{\mathbf{x},\mathbf{y}}^{t}(d\widetilde{\omega}) \,\mathrm{e}^{-\frac{t\kappa n}{\tau} - \int_{0}^{t} ds \, z \left(\frac{1}{3\tau^{3}} \sum_{i \neq j \neq k \neq i}^{n} w_{i,j}(\widetilde{\omega}(s)) w_{j,k}(\widetilde{\omega}(s)) w_{k,i}(\widetilde{\omega}(s))\right)} f^{\frac{1}{2}}\left(\frac{n}{\tau}\right) \,. \tag{4.98}$$

In (4.98), we have written $\widetilde{\omega} \equiv (\omega_1, \dots, \omega_n)$ and $\mathbb{W}_{\mathbf{x}, \mathbf{y}}^t(d\widetilde{\omega}) \equiv \prod_{l=1}^n \mathbb{W}_{x_l, y_l}^t(d\omega_l)$. Furthermore, we have abbreviated $w_{i,j}(\mathbf{x}) \equiv w(x_i - x_j)$ and

$$\sum_{\substack{i \neq j \neq k \neq i}}^{n} \equiv \sum_{\substack{i,j,k\\i \neq j \neq k \neq i}}^{n}$$

all of which we use in the sequel. Using the triangle inequality, it follows that (4.98) is in absolute value

$$\leq \int \mathbb{W}_{\mathbf{x},\mathbf{y}}^{t}(d\widetilde{\omega}) e^{-\frac{t\kappa n}{\tau}} \left| e^{-\int_{0}^{t} ds \, z \left(\frac{1}{3\tau^{3}} \sum_{i \neq j \neq k \neq i}^{n} w_{i,j}(\widetilde{\omega}(s)) w_{j,k}(\widetilde{\omega}(s)) w_{k,i}(\widetilde{\omega}(s))\right)} f^{\frac{1}{2}}\left(\frac{n}{\tau}\right) \right|. \tag{4.99}$$

Using Proposition 1.7.12 once more, we deduce that (4.99) is

$$\leq \sup_{\widetilde{\omega}} \left| e^{-\int_0^t ds \, z \left(\frac{1}{3\tau^3} \sum_{i \neq j \neq k \neq i}^n w_{i,j}(\widetilde{\omega}(s)) w_{j,k}(\widetilde{\omega}(s)) w_{k,i}(\widetilde{\omega}(s)) \right)} f^{\frac{1}{2}} \left(\frac{n}{\tau} \right) \right| \left(e^{-tH_{\tau,0}} \right)^{(n)} (\mathbf{x}; \mathbf{y}).$$

$$(4.100)$$

Arguing as in (4.94), we have

$$\sup_{\widetilde{\omega}} \left| e^{-\int_0^t ds \, z \left(\frac{1}{3\tau^3} \sum_{i \neq j \neq k \neq i}^n w_{i,j}(\widetilde{\omega}(s)) w_{j,k}(\widetilde{\omega}(s)) w_{k,i}(\widetilde{\omega}(s)) \right)} f^{\frac{1}{2}} \left(\frac{n}{\tau} \right) \right| \le e^{\frac{1}{3}K^3 |\operatorname{Re}(z)| \|w\|_{L^{\infty}}^3}.$$
(4.101)

Combining (4.100) and (4.101), we obtain (4.97), thus completing the proof. \Box

Combining the bounds proved in Lemmas 4.2.7 and 4.2.8 with Taylor's theorem yields the following corollary.

Corollary 4.2.9. $F^{\xi}_{\tau}(z) = \sum_{m=0}^{\infty} a^{\xi}_{\tau,m} z^m$ is analytic on the whole of \mathbb{C} .

4.2.4 Analysis of the classical series (4.88)

We now prove that the explicit and remainder terms defined in (4.89) and (4.90) satisfy sufficient bounds for the function defined in (4.88) to be analytic on \mathbb{C} .

Lemma 4.2.10. Let $m \in \mathbb{N}$ and a_m^{ξ} be defined as in (4.89). Then

$$\left|a_{m}^{\xi}\right| \leq \frac{(K^{3} \|w\|_{L^{\infty}}^{3})^{m} K^{p} \|\xi\|}{3^{m} m!} \,. \tag{4.102}$$

Proof. Using Lemma 4.2.2 as in the proof of Lemma 3.2.5, it is sufficient to prove that

$$\left| \mathcal{W}f^{\frac{1}{m+1}}(\mathcal{N}) \right| \le \frac{1}{3}K^3 \|w\|_{L^{\infty}}^3.$$
 (4.103)

Using Lemma 4.2.3 and (2.54), we have

$$\left| \mathcal{W}f^{\frac{1}{m+1}}(\mathcal{N}) \right| \le \frac{1}{3} K^3 \|w\|_{L^{\infty}}^3 \|f^{\frac{1}{m+1}}\|_{L^{\infty}}.$$

Here we recall that $\mathcal{N} = \|\varphi\|_{\mathfrak{h}}^2$. Noting that $\|f\|_{L^{\infty}} \leq 1$, (4.103) follows. \Box

Lemma 4.2.11. Let $M \in \mathbb{N}$ and $R_m^{\xi}(z)$ be defined as in (4.90). Then

$$\left| R_{M}^{\xi}(z) \right| \leq e^{\frac{1}{3}K^{3}|\operatorname{Re}(z)|\|w\|_{L^{\infty}}^{3}} \frac{\left(K^{3}\|w\|_{L^{\infty}}^{3} \right)^{M} K^{p}\|\xi\|}{M! \, 3^{M}} \, |z|^{M} \,. \tag{4.104}$$

Proof. We note that Lemma 4.2.3 and (2.54) imply that for any $\tilde{z} \in [0, z]$, we have

$$\left| \mathrm{e}^{-z\mathcal{W}} f^{\frac{1}{M+2}}(\mathcal{N}) \right| \leq \mathrm{e}^{\frac{1}{3}K^3 |\mathrm{Re}(z)| \|w\|_{L^{\infty}}^3}.$$

Recalling (4.90) and using Lemma 4.2.10, we obtain (4.104).

Combining Lemmas 4.2.10 and 4.2.11, we have the following corollary. Corollary 4.2.12. $F^{\xi}(z) = \sum_{m=0}^{\infty} a_m^{\xi} z^m$ is analytic on the whole of \mathbb{C} .

4.2.5 Proof of Theorem 2.2.6

We note the following result, whose proof we defer to Section 4.2.6 below.

Proposition 4.2.13. Let $\nu > 0$ be fixed. Let C_p be as in (1.13). Define

$$\alpha_{\tau,m}^{\xi,\nu} := \frac{(-1)^m}{Z_{\tau,0}} \operatorname{Tr} \left(\int_0^1 dt_1 \int_0^{t_1} dt_2 \dots \int_0^{t_{m-1}} dt_m \,\Theta_\tau(\xi) \mathrm{e}^{-(1-t_1)(H_{\tau,0}+\nu\mathcal{N}_\tau)} \mathcal{W}_\tau \right. \\ \times \,\mathrm{e}^{-(t_1-t_2)(H_{\tau,0}+\nu\mathcal{N}_\tau)} \mathcal{W}_\tau \mathrm{e}^{-(t_2-t_3)(H_{\tau,0}+\nu\mathcal{N}_\tau)} \dots \\ \times \,\mathrm{e}^{-(t_{m-1}-t_m)(H_{\tau,0}+\nu\mathcal{N}_\tau)} \mathcal{W}_\tau \mathrm{e}^{-t_M(H_{\tau,0}+\nu\mathcal{N}_\tau)} \right), \tag{4.105}$$

$$\alpha_m^{\xi,\nu} := \frac{(-1)^m}{m!} \int d\mu \,\Theta(\xi) \mathcal{W}^m \mathrm{e}^{-\nu\mathcal{N}}.$$
(4.106)

Then, the following results hold uniformly in $\xi \in C_p$.

(i)
$$\left| \alpha_{\#,m}^{\xi,\nu} \right| \le C(m,p,\nu).$$

(*ii*)
$$\lim_{\tau \to \infty} \alpha_{\tau,m}^{\varsigma,\nu} = \alpha_m^{\varsigma,\nu}$$
.

Proof of Theorem 2.2.6. By using Proposition 4.2.13 and by arguing analogously as for Lemma 3.2.10, we deduce that for $m \in \mathbb{N}$, we have

$$\lim_{\tau \to \infty} a_{\tau,m}^{\xi} = a_m^{\xi} \,, \tag{4.107}$$

uniformly in $\xi \in C_p$. We then combine (4.107) with Corollary 4.2.9, Corollary 4.2.12, Lemma 4.2.7, Lemma 4.2.10, and the dominated convergence theorem to deduce that for all $z \in \mathbb{C}$, we have

$$\lim_{\tau \to \infty} \sup_{\xi \in \mathcal{C}_p} \left| F_{\tau}^{\xi}(z) - F^{\xi}(z) \right| \le \lim_{\tau \to \infty} \sum_m \sup_{\xi \in \mathcal{C}_p} |a_{\tau,m}^{\xi} - a_m^{\xi}| |z|^m = 0.$$
(4.108)

The claim (2.75) follows from (4.108) by taking p = 0 and recalling (2.56), (2.69)–(2.70), (4.82)–(4.83), and (4.87)–(4.88) above. Note that by convention, when p = 0, there is no observable ξ in the analysis above. Moreover, we take $\mathcal{A} = \mathbf{1}$ in (4.82) and X = 1 in (4.87), respectively.

The claim (2.74) follows from (4.108) by a duality argument. More precisely, by using (2.68) and (2.71)–(2.72), we have that for all $\xi \in \mathcal{L}(\mathfrak{h}^{(p)})$

$$\rho_{\tau}(\Theta_{\tau}(\xi)) = \operatorname{Tr}\left(\gamma_{\tau,p}\,\xi\right). \tag{4.109}$$

Analogously, by using (2.57)-(2.59), we have

$$\rho(\Theta(\xi)) = \operatorname{Tr}(\gamma_p \xi). \tag{4.110}$$

Recalling (4.81) and (4.86), we see that (4.108) implies

$$\lim_{\tau \to \infty} \sup_{\xi \in \mathcal{C}_p} \left| \rho_\tau(\Theta_\tau(\xi)) - \rho(\Theta(\xi)) \right| = 0.$$
(4.111)

From (4.109)–(4.110) (and taking suprema over $\xi \in \mathfrak{B}_p$), we immediately deduce the weaker analogue of (2.74) given by

$$\lim_{\tau \to \infty} \|\gamma_{\tau,p} - \gamma_p\|_{\mathfrak{S}^2(\mathfrak{h}^{(p)})} = 0.$$
(4.112)

We upgrade (4.112) to (2.74) by noting that $\gamma_{\tau,p} \geq 0$ and $\gamma_p \geq 0$ in the sense of operators and that $\lim_{\tau\to\infty} \operatorname{Tr}\gamma_{\tau,p} = \operatorname{Tr}\gamma_p$. The former claim follows from Lemma 3.2.14, whose proof carries over directly to the quintic setting. The latter claim from (4.109)–(4.111) by taking $\xi = \mathbf{1}_p \in \mathcal{C}_p$. For the details of the last step, we refer the reader to [29, Lemma 4.10] and Lemma 3.2.16.

4.2.6 Graphical analysis of the untruncated explicit terms. Proof of Proposition 4.2.13

In this section, we prove Proposition 4.2.13 stated above. For the proof, we use a graphical argument similar to that used in [29, Sections 2.3-2.6 and 4.1]. The graphs will be different, due to the three-body interaction. The essence of the argument is quite similar. For completeness, we review the proof and refer the reader to [29] for more details and motivation. In what follows, we denote

$$b_{\#,m}^{\xi} := \alpha_{\#,m}^{\xi,0}, \qquad (4.113)$$

where we recall (4.105)–(4.106). Let us note that it suffices to show Proposition 4.2.13, when $\nu = 0$. The general claim follows from this one (with possibly different constants depending on ν) by replacing κ in (2.6) with $\kappa + \nu$. In Subsection 4.2.6, we show that

$$\left| b_{\tau,m}^{\xi} \right| \le C(m,p) \,, \tag{4.114}$$

uniformly in $\xi \in \mathfrak{B}_p$. In Subsection 4.2.6, we show that

$$\lim_{\tau \to \infty} b_{\tau,m}^{\xi} = b_m^{\xi} \,, \tag{4.115}$$

uniformly in $\xi \in \mathfrak{B}_p$. In Subsection 4.2.6, we show that (4.114)–(4.115) hold for $\xi = \mathbf{1}_p$. The latter requires a slightly modified graphical structure. Putting these steps together, we complete the proof of Proposition 4.2.13.

Proof of (4.114) **uniformly in** $\xi \in \mathfrak{B}_p$

We begin by recalling a number of definitions and results from [29]. Throughout, we abbreviate $\varphi_{\tau,k} := \varphi_{\tau}(u_k)$, where the u_k are defined as in (2.9) above.

Definition 4.2.14. For $t \in \mathbb{R}$, we define the operator valued distributions $(e^{th/\tau}\varphi_{\tau})(x)$ and $(e^{th/\tau}\varphi_{\tau}^*)(x)$ as

$$\left(\mathrm{e}^{th/\tau} \varphi_{\tau} \right)(x) := \sum_{k \in \mathbb{N}} \mathrm{e}^{t\lambda_{k}/\tau} u_{k}(x) \varphi_{\tau,k} \,,$$
$$\left(\mathrm{e}^{th/\tau} \varphi_{\tau}^{*} \right)(x) := \sum_{k \in \mathbb{N}} \mathrm{e}^{t\lambda_{k}/\tau} \overline{u}_{k}(x) \varphi_{\tau,k}^{*} \,.$$

Here, we recall (2.8).

In what follows, we use the result of [29, Lemma 2.3].

Lemma 4.2.15. For $t \in \mathbb{R}$ we have,

$$\mathrm{e}^{tH_{\tau,0}}\varphi_{\tau}^{*}(x)\mathrm{e}^{-tH_{\tau,0}} = \left(\mathrm{e}^{th/\tau}\varphi_{\tau}^{*}\right)(x), \quad \mathrm{e}^{tH_{\tau,0}}\varphi_{\tau}(x)\mathrm{e}^{-tH_{\tau,0}} = \left(\mathrm{e}^{-th/\tau}\varphi_{\tau}\right)(x).$$

Here, we recall the definition (2.64) of $H_{\tau,0}$.

The following result follows from Lemma 4.2.15 and the definition (2.63) of W_{τ} ; see also [29, Corollary 2.4].

Lemma 4.2.16. For $t \in \mathbb{R}$, we have

$$e^{tH_{\tau,0}} \mathcal{W}_{\tau} e^{-tH_{\tau,0}} = \frac{1}{3} \int dx \, dy \, dz \, \left(e^{th/\tau} \varphi_{\tau}^* \right) (x) \left(e^{th/\tau} \varphi_{\tau}^* \right) (y) \left(e^{th/\tau} \varphi_{\tau}^* \right) (z) \times w(x-y)w(y-z)w(z-x) \left(e^{-th/\tau} \varphi_{\tau} \right) (x) \left(e^{-th/\tau} \varphi_{\tau} \right) (y) \left(e^{-th/\tau} \varphi_{\tau} \right) (z) .$$

$$(4.116)$$

Given a closed operator \mathcal{A} on \mathcal{F} , we define

$$\rho_{\tau,0}(\mathcal{A}) := \frac{\operatorname{Tr}(\mathcal{A}e^{-H_{\tau,0}})}{\operatorname{Tr}(e^{-H_{\tau,0}})}.$$
(4.117)

Furthermore, let us recall (4.113) and (4.105). Using Lemma 4.2.16 and the cyclicity of the trace, it follows that for all $m, p \in \mathbb{N}$ and $\xi \in \mathfrak{B}_p$, we have

$$b_{\tau,m}^{\xi} = \frac{(-1)^m}{3^m} \int_0^1 dt_1 \int_0^{t_1} dt_2 \dots \int_0^{t_{m-1}} dt_m \, g_{\tau,m}^{\xi}(\mathbf{t}) \,, \tag{4.118}$$

where $\mathbf{t} := (t_1, \ldots, t_m)$ and

$$g_{\tau,m}^{\xi}(\mathbf{t}) := \int dx_1 \dots dx_{m+p} \, dy_1 \dots dy_{m+p} \, dz_1 \dots dz_m$$

$$\times \left(\prod_{i=1}^m w(x_i - y_i) w(y_i - z_i) w(z_i - x_i) \right) \xi(x_{m+1}, \dots, x_{m+p}; y_{m+1}, \dots, y_{m+p})$$

$$\times \rho_{\tau,0} \left(\prod_{i=1}^m \left[\left(e^{t_i h_{\tau}/\tau} \varphi_{\tau}^* \right) (x_i) \left(e^{t_i h_{\tau}/\tau} \varphi_{\tau}^* \right) (y_i) \left(e^{t_i h_{\tau}/\tau} \varphi_{\tau}^* \right) (z_i) \right. \\ \left. \times \left(e^{-t_i h_{\tau}/\tau} \varphi_{\tau} \right) (x_i) \left(e^{-t_i h_{\tau}/\tau} \varphi_{\tau} \right) (y_i) \left(e^{-t_i h_{\tau}/\tau} \varphi_{\tau} \right) (z_i) \right]$$

$$\times \prod_{i=1}^p \varphi_{\tau}^*(x_{m+i}) \prod_{i=1}^p \varphi_{\tau}(y_{m+i}) \right). \quad (4.119)$$

Here and throughout all of our products are taken in the order of increasing indices. We now fix $m, p \in \mathbb{N}$ and define an abstract vertex set Σ containing (6m+2p)

elements as follows.

Definition 4.2.17. Given $m, p \in \mathbb{N}$, we define $\Sigma \equiv \Sigma(m, p)$ to be the set of triples (i, r, δ) with $i \in \{1, \ldots, m+1\}$. If $i \in \{1, \ldots, m\}$, we consider $r \in \{1, 2, 3\}$ and if i = m + 1, we take $r \in \{1, \ldots, p\}$. Finally, we take $\delta \in \{\pm 1\}$. We write⁴ $\alpha = (i, r, \delta)$ and write the components of α as $i_{\alpha}, r_{\alpha}, \delta_{\alpha}$. We use the lexicographical ordering on Σ to order the vertices, which we denote by \leq . If $\alpha \leq \beta$ and $\alpha \neq \beta$, we write $\alpha < \beta$. To each vertex $\alpha = (i, r, \delta) \in \Sigma$, we assign a spatial integration variable x_{α} . Moreover, to each $i \in \{1, \ldots, m\}$, we assign a time t_i , and take $t_{m+1} := 0$ by convention. Where convenient, we write $x_{i,r,\delta}$ or $t_{i,r,\delta}$ instead of x_{α} or t_{α} respectively. Let us write

$$\mathbf{x} := (x_{\alpha})_{\alpha \in \Sigma} \in \Lambda^{\Sigma}, \qquad \mathbf{t} := (t_{\alpha})_{\alpha \in \Sigma} \in \mathbb{R}^{\Sigma}.$$

We only consider (t_1, \ldots, t_m) to be in the support of the integral from (4.118). In other words, we always take $\mathbf{t} \in \mathfrak{V} \equiv \mathfrak{V}(m)$, where

$$\mathfrak{V} := \left\{ \mathbf{t} \in \mathbb{R}^{\Sigma} : t_{i,r,\delta} = t_i \text{ with } 0 = t_{m+1} < t_m < \dots < t_1 < 1 \right\}.$$
(4.120)

In the discussion that follows, we fix $m, p \in \mathbb{N}$ as well as $\xi \in \mathfrak{B}_p$.

Remark 4.2.18. We interpret the integrand in (4.119) in terms of the set Σ in Definition 4.2.17 as follows. Each occurrence of $\varphi_{\tau}^*(\cdot)$ or $\varphi_{\tau}^*(\cdot)$ corresponds to an element of Σ . For $i \in \{1, \ldots, m\}$, *i* denotes that we are working with the *i*th interaction, and hence that we are considering a factor of the form $(e^{t_i h_{\tau}/\tau} \varphi_{\tau}^*)(\cdot)$ or $(e^{-t_i h_{\tau}/\tau} \varphi_{\tau})(\cdot)$. When $\delta = +1$, it is the former and when $\delta = -1$, it is the latter. The index r = 1, 2, 3 refers to the integration variable x_i, y_i, z_i respectively. Furthermore, when i = m + 1, we consider the factor $\varphi_{\tau}^*(x_{m+r})$ and when $\delta = +1$, and the factor $\varphi_{\tau}(y_{m+r})$ when $\delta = -1$; see Figure 4.1 for a graphical representation. Let us also note that

$$\alpha < \beta \quad \Rightarrow \quad 0 \le t_{\alpha} - t_{\beta} < 1.$$
(4.121)

Let us recall the quantum Wick theorem.

Lemma 4.2.19 (Quantum Wick theorem). Let A_1, \ldots, A_n be operators of the form $A_i = b(f_i)$ or $A_i = b^*(f_i)$, for $f_1, \ldots, f_n \in \mathfrak{h}$. Here, we recall (2.24)–(2.25). We then have

$$\rho_{\tau,0}(\mathcal{A}_1\cdots\mathcal{A}_n) = \sum_{\Pi \in M(n)} \prod_{(i,j)\in\Pi} \rho_{\tau,0}(\mathcal{A}_i\mathcal{A}_j)$$

⁴We emphasise that this is a different object than the $\alpha_{\#,m}^{\xi,\nu}$ in (4.105)–(4.106) above.



Figure 4.1: An unpaired graph from Definition 4.2.17 with m = p = 3. The black dots correspond to factors of $\varphi_{\tau}(\cdot)$ and $\varphi_{\tau}^{*}(\cdot)$. The wavy lines correspond to factors of the interaction potential w.

where, as in Proposition 2.1.2, M(n) denotes the set of complete pairings of $\{1, \ldots, n\}$. The edges of Π are now labelled using ordered pairs (i, j) with i < j. Here, we also recall (4.117).

For a self-contained proof of Lemma 4.2.19, we refer the reader to [29, Lemma B.1]. As in [29, Section 2], we use Lemma 4.2.19 to simplify the expression (4.119). Before proceeding, we define a few objects which we will use in the analysis.

Definition 4.2.20. Given Π a pairing of Σ , i.e. a one-regular graph on Σ , we regard its edges as ordered pairs (α, β) such that $\alpha < \beta$. We then define $\mathfrak{P} \equiv \mathfrak{P}(m, p)$ to be the set of pairings Π of Σ satisfying $\delta_{\alpha}\delta_{\beta} = -1$ whenever $(\alpha, \beta) \in \Pi$; see Figure 4.2.



Figure 4.2: A graph with m = p = 3 with a valid pairing from Definition 4.2.20.

Definition 4.2.21. For $\alpha \in \Sigma$ and $\mathbf{t} \in \mathfrak{V}$, define

$$\mathcal{B}_{\alpha}(\mathbf{x}, \mathbf{t}) := \begin{cases} \left(e^{t_{\alpha}h/\tau} \varphi_{\tau}^{*} \right) (x_{\alpha}) & \text{if } \delta = 1, \\ \left(e^{-t_{\alpha}h/\tau} \varphi_{\tau} \right) (x_{\alpha}) & \text{if } \delta = -1. \end{cases}$$
(4.122)

Definition 4.2.22. For $\Pi \in \mathfrak{P}$ and $\mathbf{t} \in \mathfrak{V}$, we define

$$I_{\tau,\Pi}^{\xi}(\mathbf{t}) := \int_{\Lambda^{\Sigma}} d\mathbf{x} \prod_{i=1}^{m} \left(w(x_{i,1,1} - x_{i,2,1}) w(x_{i,2,1} - x_{i,3,1}) w(x_{i,3,1} - x_{i,1,1}) \prod_{r=1}^{3} \delta(x_{i,r,1} - x_{i,r,-1}) \right) \\ \times \xi(x_{m+1}, \dots, x_{m+p}; y_{m+1}, \dots, y_{m+p}) \prod_{(\alpha,\beta)\in\Pi} \rho_{\tau,0}(\mathcal{B}_{\alpha}(\mathbf{x}, \mathbf{t})\mathcal{B}_{\beta}(\mathbf{x}, \mathbf{t})) .$$
(4.123)

We can now state the simplification of (4.119) that we will use in the sequel.

Lemma 4.2.23. For $\mathbf{t} \in \mathfrak{V}$, we have $g_{\tau,m}^{\xi}(\mathbf{t}) = \sum_{\Pi \in \mathfrak{P}} I_{\tau,\Pi}(\mathbf{t})$.

Proof. We follow the proof and use the notation used in [29, Lemma 2.8]. Namely for $\alpha \in \Sigma$, to each x_{α} , we define a corresponding spectral label $k_{\alpha} \equiv k_{i,r,\delta}$, and write $\mathbf{k} := (k_{\alpha})_{\alpha \in \Sigma}$. Let us recall (2.9). For $l \in \mathbb{N}$, we let

$$u_l^{\alpha} := \begin{cases} \overline{u}_l \text{ if } \delta = +1, \\ u_l \text{ if } \delta = -1. \end{cases}$$

From (4.122), it follows that

$$\mathcal{B}_{\alpha}(\mathbf{x}, \mathbf{t}) = \sum_{k_{\alpha} \in \mathbb{N}} e^{\delta_{\alpha} t_{\alpha} \lambda_{k_{\alpha}} / \tau} u_{k_{\alpha}}^{\alpha}(x_{\alpha}) \mathcal{A}_{\alpha}(\mathbf{k}) , \qquad (4.124)$$

where we define

$$\mathcal{A}_{\alpha}(\mathbf{k}) := \begin{cases} \varphi_{\tau,k_{\alpha}}^{*} \text{ if } \delta = +1 \,, \\ \varphi_{\tau,k_{\alpha}} \text{ if } \delta = -1 \,. \end{cases}$$
Using (4.124) in (4.119), we have

$$g_{\tau,m}^{\xi}(\mathbf{t}) = \int_{\Lambda^{\Sigma}} d\mathbf{x} \prod_{i=1}^{m} \left(w(x_{i,1,1} - x_{i,2,1}) w(x_{i,2,1} - x_{i,3,1}) w(x_{i,3,1} - x_{i,1,1}) \prod_{r=1}^{3} \delta(x_{i,r,1} - x_{i,r,-1}) \right) \\ \times \xi(x_{m+1,1,1}, \dots, x_{m+1,p,1}; y_{m+1,1,-1}, \dots, y_{m+1,p,-1}) \\ \times \sum_{\mathbf{k}} \left[\left(\prod_{\alpha \in \Sigma} e^{\delta_{\alpha} t_{\alpha} \lambda_{\tau,k_{\alpha}} / \tau} u_{k_{\alpha}}^{\alpha}(x_{\alpha}) \right) \rho_{\tau,0} \left(\prod_{\alpha \in \Sigma} \mathcal{A}_{\alpha}(\mathbf{k}) \right) \right]. \quad (4.125)$$

The result follows from applying Lemma 4.2.19 to (4.125) and using (4.124).

We define the following bounded operators.

$$S_{\tau,t} := e^{-th/\tau} \text{ if } t \ge 0,$$
$$G_{\tau,t} := \frac{e^{-th/\tau}}{\tau(e^{h/\tau} - 1)} \text{ if } t \ge -1$$

Here we note that $G_{\tau,t}$ is the time-evolved Green function.

Lemma 4.2.24. For $t \ge 0$, both $G_{\tau,t}$ and $S_{\tau,t}$ have symmetric, non-negative kernels.

Proof. Since both $S_{\tau,t}$ and $G_{\tau,t}$ are self-adjoint, it follows from Proposition 1.7.12 that their kernels are non-negative and thus symmetric. See [29, Lemma 2.9] for more details.

We thus have the following result for computing the free quantum states of products of pairs of \mathcal{B}_{α} ; see [29, Lemma 2.10].

Lemma 4.2.25. Suppose $\alpha, \beta \in \Sigma$ with $\alpha < \beta$.

1. If $\delta_{\alpha} = 1$ and $\delta_{\beta} = -1$ with $t_{\alpha} - t_{\beta} < 1$, then

$$\rho_{\tau,0}\left(\mathcal{B}_{\alpha}(\mathbf{x},\mathbf{t})\mathcal{B}_{\beta}(\mathbf{x},\mathbf{t})\right) = G_{\tau,-(t_{\alpha}-t_{\beta})}(x_{\alpha};x_{\beta}).$$

2. If $\delta_{\alpha} = -1$ and $\delta_{\beta} = 1$ with $t_{\alpha} - t_{\beta} \ge 0$, then

$$\rho_{\tau,0}\left(\mathcal{B}_{\alpha}(\mathbf{x},\mathbf{t})\mathcal{B}_{\beta}(\mathbf{x},\mathbf{t})\right) = G_{\tau,t_{\alpha}-t_{\beta}}(x_{\alpha};x_{\beta}) + \frac{1}{\tau}S_{\tau,t_{\alpha}-t_{\beta}}(x_{\alpha};x_{\beta}).$$

3. For both (1) and (2), by Lemma 4.2.24, we have $\rho_{\tau,0}\left(\mathcal{B}_{\alpha}(\mathbf{x},\mathbf{t})\mathcal{B}_{\beta}(\mathbf{x},\mathbf{t})\right) \geq 0$.

We define a second, coloured graph, where we collapse the left-hand side vertices of the original graph. In other words, we identify $x_{i,r,1}$ and $x_{i,r,-1}$ for $i \leq m$. In order to make this precise, we use first define the following equivalence relation.

Definition 4.2.26. For $\alpha, \beta \in \Sigma$, we write $\alpha \sim \beta$ if and only if $i_{\alpha} = i_{\beta} \leq m$ and $r_{\alpha} = r_{\beta}$.

Let us now implement Definition 4.2.26 in the graphical structure.

Definition 4.2.27. For each $\Pi \in \mathfrak{P}$, we define the coloured graph $(V_{\Pi}, E_{\Pi}, \sigma_{\Pi}) = (V, E, \sigma)$ as follows. $V := \Sigma / \sim$ is the set of equivalence classes of Σ under \sim . For $\alpha \in \Sigma$, let us denote by $[\alpha]$ its equivalence class under \sim . For each edge $(\alpha, \beta) \in \Pi$, we obtain an edge $e = \{[\alpha], [\beta]\}$, and we denote by E the set of edges obtained in this way. Finally, we define the colour of an edge $e \in E$ as

$$\sigma(e) := \delta_{\beta} \,. \tag{4.126}$$

This is well-defined by construction.



Figure 4.3: An example of the coloured graph from Definition 4.2.27 with the same pairing as Figure 4.2. The wider dotted edges have colour -1, and the finer dotted lines have colour 1.

Remark 4.2.28. We make the following observations about (V, E, σ) .

- 1. The set V inherits a well-defined total order from Σ defined by $[\alpha] \leq [\beta]$ if $\alpha \leq \beta$. We also adopt the same convention as before to write $[\alpha] < [\beta]$ if $\alpha < \beta$.
- 2. We can write V as the disjoint union $V = V_2 \sqcup V_1$, where

$$V_2 := \{(i, r) \mid i \in \{1, \dots, m\}, r \in \{1, 2, 3\}\}, V_1 := \{(m + 1, r, \pm 1)\}.$$

Note that then each vertex in V_j has degree j.

3. We write $\operatorname{conn}(E)$ to denote the set of connected components of E. Thus $E = \bigsqcup_{\mathcal{P} \in \operatorname{conn}(E)} \mathcal{P}$. We call $\mathcal{P} \in \operatorname{conn}(E)$ a *path* of E.

We now fix $m, p \in \mathbb{N}$ and $\Pi \in \mathfrak{P}$, and let (V, E, σ) be the associated graph defined in Definition 4.2.27. For each $\mathbf{x} \in \Lambda^{\Sigma}$ and $\mathbf{t} \in \mathbb{R}^{\Sigma}$, we associate integration labels $\mathbf{y} := (y_a)_{a \in V} \in \Lambda^V$ and $\mathbf{s} := (s_a)_{a \in V} \in \mathbb{R}^V$ defined by

$$y_{\alpha} := x_{[\alpha]}, \qquad s_{\alpha} := t_{[\alpha]} \tag{4.127}$$

for any $\alpha \in \Sigma$. It follows from Definition 4.2.17 that the definition above does not depend on the choice of vertex α . We note that (4.121) implies

$$a < b \quad \Rightarrow \quad 0 \le s_a - s_b < 1 \,, \tag{4.128}$$

and

$$s_a = s_b \text{ if and only if } i_a = i_b, \qquad (4.129)$$

where we have used a slight abuse of notation to write $i_{\alpha} := i_{[\alpha]}$.

Definition 4.2.29. We say that a path $\mathcal{P} \in \text{conn}(E)$ is *closed* if all of its vertices are in V_2 . Otherwise we call it *open*. We also denote by $V(\mathcal{P}) := \bigcup_{e \in \mathcal{P}} e$ and $V_i(\mathcal{P}) := V(\mathcal{P}) \cap V_i$.

Definition 4.2.30. For $\mathbf{y} \in \Lambda^V$ and $\mathbf{s} \in \mathbb{R}^V$ satisfying (4.128), and $e = \{a, b\} \in E$, we define $\mathbf{y}_e := (y_a; y_b) \in \Lambda^e$ and the integral kernels

$$\mathcal{J}_{\tau,e}(\mathbf{y}_e, \mathbf{s}) := G_{\tau,\sigma(e)(s_a - s_b)}(\mathbf{y}_e) + \frac{\chi(\sigma(e) = 1)\chi(s_a \neq s_b)}{\tau} S_{\tau,s_a - s_b}(\mathbf{y}_e), \quad (4.130)$$

$$\hat{\mathcal{J}}_{\tau,e}(\mathbf{y}_e, \mathbf{s}) := G_{\tau,\sigma(e)(s_a - s_b)}(\mathbf{y}_e) \,. \tag{4.131}$$

Here, we recall (4.126).

Although both of the operators (4.130)–(4.131) always Hilbert-Schmidt, we never estimate them in the Hilbert-Schmidt norm as we want to prove estimates which are uniform in **s**. In the sequel, let us write

$$\mathbf{y} = (\mathbf{y}_1, \mathbf{y}_2), \qquad (4.132)$$

with $\mathbf{y}_i := (\mathbf{y}_a)_{a \in V_i}$. Slightly abusing notation, let us also write

$$\xi(y_{m+1,1,1},\ldots,y_{m+1,p,1};y_{m+1,1,-1},\ldots,y_{m+1,p,-1}) =: \xi(\mathbf{y}_1).$$

Lemma 4.2.31. For $\mathbf{t} \in \mathfrak{V}$ and \mathbf{s} defined as in (4.127), we have

$$I_{\tau,\Pi}^{\xi}(\mathbf{t}) = \int_{\Lambda^{V}} d\mathbf{y} \bigg(\prod_{i=1}^{m} w(y_{i,1} - y_{i,2}) w(y_{i,2} - y_{i,3}) w(y_{i,3} - y_{i,1}) \bigg) \xi(\mathbf{y}_{1}) \prod_{e \in E} \mathcal{J}_{\tau,e}(\mathbf{y}_{e}, \mathbf{s}) \,.$$

$$(4.133)$$

Proof. We define a map $T : \Lambda^V \to \Lambda^{\Sigma}$ by $T\mathbf{y} := (y_{[\alpha]})_{\alpha \in \Sigma}$. Then T is a bijection from Λ^V onto the subset of Λ^{Σ} defined by

$$\{\mathbf{x} \in \Lambda^{\Sigma} \mid x_{i,r,1} = x_{i,r,-1} \text{ for all } i \in \{1, \dots, m\}, r \in \{1, 2, 3\}\}.$$

From Lemma 4.2.25 and Definition 4.2.30, it follows that

$$\rho_{\tau,0}\left(\mathcal{B}_{\alpha}(T\mathbf{y},\mathbf{t})\mathcal{B}_{\alpha}(T\mathbf{y},\mathbf{t})\right) = \mathcal{J}_{\tau,e}(\mathbf{y}_{e},\mathbf{s}), \qquad (4.134)$$

where $(\alpha, \beta) \in \Pi$ is chosen such that $[\alpha] = a$ and $[\beta] = b$. By making the change of variables, we obtain $\mathbf{x} = T\mathbf{y}$, (4.133), as was claimed.

Let us note the following corollary.

Corollary 4.2.32. For $t \in \mathfrak{V}$, we have

$$\left| I_{\tau,\Pi}^{\xi}(\mathbf{t}) \right| \le \|w\|_{L^{\infty}}^{3m} \int_{\Lambda^{V}} d\mathbf{y} \left| \xi(\mathbf{y}_{1}) \right| \prod_{e \in E} \mathcal{J}_{\tau,e}(\mathbf{y}_{e}, \mathbf{s}) \,. \tag{4.135}$$

Proof. This is a consequence of Lemmas 4.2.31 and 4.2.25 (3).

Lemma 4.2.33. Suppose that $\mathcal{P} \in \text{conn}(E)$ is a closed path as in Definition 4.2.29. Then, we have

$$\int_{\Lambda^{V(\mathcal{P})}} \prod_{a \in V(\mathcal{P})} dy_a \prod_{e \in \mathcal{P}} \mathcal{J}_{\tau,e}(\mathbf{y}_e, \mathbf{s}) \le C^{|V(\mathcal{P})|} \,. \tag{4.136}$$

Moreover

$$\left| \int_{\Lambda^{V(\mathcal{P})}} \prod_{a \in V(\mathcal{P})} dy_a \left[\prod_{e \in \mathcal{P}} \mathcal{J}_{\tau,e}(\mathbf{y}_e, \mathbf{s}) - \prod_{e \in \mathcal{P}} \hat{\mathcal{J}}_{\tau,e}(\mathbf{y}_e, \mathbf{s}) \right] \right| \to 0 \ as \ \tau \to \infty \,. \tag{4.137}$$

We note that h defined as in (2.6) satisfies h > 0 and $h^{-1} \in \mathfrak{S}^1(\mathfrak{h}) \subset \mathfrak{S}^2(\mathfrak{h})$.

Proof of Lemma 4.2.33. Since \mathcal{P} is closed, we need to consider two cases depending on the length of \mathcal{P} . If $|\mathcal{P}| = 1$, the path is a loop. Then the left-hand side of (4.136)

equals

$$\int_{\Lambda} dy_a G_{\tau}(y_a; y_a) = \|G_{\tau}\|_{\mathfrak{S}^1(\mathfrak{h})} = \operatorname{Tr}(G_{\tau}) = \sum_{k \ge 0} \frac{1}{\tau(e^{\lambda_k/\tau} - 1)}$$
$$\leq \sum_{k \ge 0} \frac{1}{\lambda_k} = \operatorname{Tr}(h^{-1}) = \|G\|_{\mathfrak{S}^1(\mathfrak{h})} < \infty. \quad (4.138)$$

Here we have used the positivity of G_{τ} as an operator as well as (2.8).

We henceforth need to consider the case when $|\mathcal{P}| \geq 2$. Here, we argue as in [29, Lemma 2.17]. Let $|\mathcal{P}| = q$, and write $\mathcal{P} = \{e_1, \ldots, e_q\}$, where e_j and e_{j+1} are incident for $j = 1, \ldots, q$. Throughout the proof we take the index j to be modulo q. We denote by a_j the unique vertex in $e_{j-1} \cap e_j$. Without loss of generality, we take $a_1 < a_2$.

An induction argument shows that the colour of e_j is determined by the colour of e_1 . Namely, recalling (4.126), we have

$$\sigma(e_j) = \begin{cases} \sigma(e_1) & \text{if } a_j < a_{j+1}, \\ -\sigma(e_1) & \text{if } a_j > a_{j+1}. \end{cases}$$
(4.139)

For j = 1, ..., q, we define $a_{j,-} := \min\{a_j, a_{j+1}\}$ and $a_{j,+} := \max\{a_j, a_{j+1}\}$. From (4.139), we have

$$\sigma(e_j)(s_{a_{j,-}} - s_{a_{j,+}}) = \sigma(e_1)(s_{a_j} - s_{a_{j+1}}).$$
(4.140)

Moreover, it is clear from (4.128) and (4.129) that $0 \leq \sigma(e_1)(s_{a_j} - s_{a_{j+1}}) < 1$. Thus, $G_{\tau,\sigma(e_1)(s_{a_j} - s_{a_{j+1}})}$ and $S_{\tau,\sigma(e_1)(s_{a_j} - s_{a_{j+1}})}$ are both well-defined. Substituting (4.140) into (4.130), we have

$$\mathcal{J}_{\tau,e_{j}}(\mathbf{y}_{e_{j}};\mathbf{s}) = G_{\tau,\sigma(e_{1})(s_{a_{j}}-s_{a_{j+1}})}(y_{a_{j}};y_{a_{j+1}}) + \frac{\chi(\sigma(e_{j})=1)\chi(s_{a_{j}}\neq s_{a_{j+1}})}{\tau}S_{\tau,\sigma(e_{1})(s_{a_{j}}-s_{a_{j+1}})}(y_{a_{j}};y_{a_{j+1}}). \quad (4.141)$$

Here we use that the kernels of $G_{\tau,t}$ and $S_{\tau,t}$ are symmetric. Rewriting the left-hand side of (4.136) using (4.141), we get

$$\operatorname{Tr}\left[\prod_{j=1}^{q} \left(G_{\tau,\sigma(e_1)(s_{a_j}-s_{a_{j+1}})} + \frac{\chi(\sigma(e_j)=1)\chi(s_{a_j}\neq s_{a_{j+1}})}{\tau} S_{\tau,\sigma(e_1)(s_{a_j}-s_{a_{j+1}})} \right) \right].$$
(4.142)

By definition, all of these operators commute, and hence the order of the above

product does not matter. We define

$$J_{\mathcal{P}} := \left\{ j \in \{1, \dots, q\} : \mathcal{J}_{\tau, e_j} \neq \hat{\mathcal{J}}_{\tau, e_j} \right\}.$$

We note that $J_{\mathcal{P}} \neq \{1, \ldots, q\}$. Namely, we recall (4.130)–(4.131) and note that by construction, the smallest vertex in \mathcal{P} (with respect to \leq) is incident to an edge e with $\sigma(e) = -1$. Therefore, we can rewrite (4.142) as

$$\sum_{I \subset J_{\mathcal{P}}} \operatorname{Tr} \left[\left(\prod_{j \in \{1, \dots, q\} \setminus I} G_{\tau, \sigma(e_1)(s_{a_j} - s_{a_{j+1}})} \right) \left(\prod_{j \in I} \frac{1}{\tau} S_{\tau, \sigma(e_1)(s_{a_j} - s_{a_{j+1}})} \right) \right]$$
$$= \sum_{I \subset J_{\mathcal{P}}} \operatorname{Tr} \left[\left(\prod_{j \in \{1, \dots, q\} \setminus I} G_{\tau, 0} \right) \left(\prod_{j \in \{1, \dots, q\} \setminus I} \frac{1}{\tau} S_{\tau, 0} \right) \right] = \sum_{J \subset J_{\mathcal{P}}} \frac{1}{\tau^{|I|}} \operatorname{Tr}(G_{\tau}^{q-|I|}).$$
(4.143)

The first equality holds since $\sum_{j=1}^{q} (s_{a_j} - s_{a_{j+1}}) = s_{a_1} - s_{a_{q+1}} = 0$ because $s_{a_1} = s_{a_{q+1}}$. By Lemma 4.2.4, we note that for $|I| \leq q - 2$,

$$\operatorname{Tr}(G_{\tau}^{q-|I|}) = \|G_{\tau}\|_{\mathfrak{S}^{q-|I|}(\mathfrak{h})}^{q-|I|} \le \|G_{\tau}\|_{\mathfrak{S}^{2}(\mathfrak{h})}^{q-|I|}.$$
(4.144)

Using (4.144) we have that (4.143) is

$$\leq \sum_{\substack{I \subset J_{\mathcal{P}} \\ |I| \leq q-2}} \frac{1}{\tau^{|I|}} \|G_{\tau}\|_{\mathfrak{S}^{2}(\mathfrak{h})}^{q-|I|} + \frac{q}{\tau^{q-1}} \|G_{\tau}\|_{\mathfrak{S}^{1}(\mathfrak{h})} \\
\leq C^{|V(\mathcal{P})|} \left(1 + \|G_{\tau}\|_{\mathfrak{S}^{2}(\mathfrak{h})} + \frac{1}{\tau} \|G_{\tau}\|_{\mathfrak{S}^{1}(\mathfrak{h})}\right)^{|V(\mathcal{P})|} .$$
(4.145)

In the first inequality above, we used $q \ge 2$. We now deduce (4.136) from (4.145), by using

$$\|G_{\tau}\|_{\mathfrak{S}^{2}(\mathfrak{h})} \leq \|G_{\tau}\|_{\mathfrak{S}^{1}(\mathfrak{h})} \leq C, \qquad (4.146)$$

for some C > 0 independent of τ , which follows from (4.138), and Lemma 4.2.4.

To obtain (4.137), we split into the same two cases. If $|\mathcal{P}| = 1$, then $\mathcal{P} = \{e\}$, so the path is a loop, so $s_a = s_a$. So $\mathcal{J}_e = \hat{\mathcal{J}}_e$, and there is nothing to prove. If $|\mathcal{P}| \geq 2$, we apply the same argument as used in the proof of (4.136). The only difference is that we now sum over *non-empty* subsets I of $\mathcal{J}_{\mathcal{P}}$ in (4.143). This results in the an extra power of $\frac{1}{\tau}$, and one less power of $(1 + ||G_{\tau}||_{\mathfrak{S}^2(\mathfrak{h})} + \frac{1}{\tau}||G_{\tau}||_{\mathfrak{S}^1(\mathfrak{h})})$ in (4.145). We hence deduce (4.137). **Lemma 4.2.34.** Suppose that $\mathcal{P} \in \text{conn}(E)$ is an open path with endpoints $b_1, b_2 \in V_2(\mathcal{P})$. Then, we have

$$\left\| \int_{\Lambda^{V_2(\mathcal{P})}} \prod_{a \in V_2(\mathcal{P})} dy_a \prod_{e \in \mathcal{P}} \mathcal{J}_{\tau,e}(\mathbf{y}_e, \mathbf{s}) \right\|_{L^2_{y_{b_1}, y_{b_2}}} \le C^{|V_2(\mathcal{P})|} \,. \tag{4.147}$$

Moreover

$$\left\| \int_{\Lambda^{V(\mathcal{P})}} \prod_{a \in V(\mathcal{P})} dy_a \left[\prod_{e \in \mathcal{P}} \mathcal{J}_{\tau,e}(\mathbf{y}_e; \mathbf{s}) - \prod_{e \in \mathcal{P}} \hat{\mathcal{J}}_{\tau,e}(\mathbf{y}_e; \mathbf{s}) \right] \right\|_{L^2_{y_{b_1}, y_{b_2}}} \to 0 \ as \ \tau \to \infty \,.$$

$$(4.148)$$

Proof. We argue similarly as for [29, Lemma 2.18]. We first prove (4.147). Let b_1, b_2 be as in the statement of the lemma. Without loss of generality, suppose $b_1 < b_2$, so $\delta_{b_1} = 1$ and $\delta_{b_2} = -1$. Let $q := |V_2(\mathcal{P})|$. If q = 0, then $\mathcal{J}_{\tau,e}(\mathbf{y}; \mathbf{s}) = G_{\tau}(y_{b_1}, y_{b_2})$, since $\sigma(e) = -1$. Hence, (4.147) follows from (4.146).

Suppose that $q \ge 1$. Write $\mathcal{P} := \{e_1, \ldots, e_{q+1}\}$, where $b_1 \in e_1, b_2 \in e_{q+1}$ and a_j is the unique vertex in $e_j \cap e_{j+1}$. An induction argument shows that the colour of e_j is determined by the colour of e_1 . Namely, we have

$$\sigma(e_j) = \begin{cases} -1 & \text{if } a_{j-1} < a_j, \\ 1 & \text{if } a_j < a_{j-1}. \end{cases}$$
(4.149)

Define $a_{j,-} := \min\{a_{j-1}, a_j\}$ and $a_{j,+} := \max\{a_{j-1}, a_j\}$. Then (4.149) implies

$$\sigma(e_j)(s_{a_{j,-}} - s_{a_{j,+}}) = s_{a_j} - s_{a_{j-1}}.$$
(4.150)

As in the proof of Lemma 4.2.33, we use (4.128) and (4.129) to deduce that $0 \leq \sigma(e_j)(s_{a_{j,-}} - s_{a_{j,+}}) < 1$. Substituting (4.149) into (4.2.30), we have

$$\mathcal{J}_{\tau,e}(\mathbf{y}_{e_j}, \mathbf{s}) = G_{\tau, s_{a_j} - s_{a_{j-1}}}(y_{a_{j-1}}; y_{a_j}) + \frac{\chi(\sigma(e_j) = 1)\chi(s_{a_j} \neq s_{a_{j-1}})}{\tau} S_{\tau, s_{a_j} - s_{a_{j-1}}}(y_{a_{j-1}}; y_{a_j}). \quad (4.151)$$

Here, we have used the symmetry of the kernels of $G_{\tau,t}$ and $S_{\tau,t}$. Using (4.151), we

have

$$\int_{\Lambda^{V_{2}(\mathcal{P})}} \prod_{a \in V_{2}(\mathcal{P})} dy_{a} \prod_{e \in \mathcal{P}} \mathcal{J}_{\tau,e}(\mathbf{y}, \mathbf{s}) \\
= \left[\prod_{j=1}^{q+1} \left(G_{\tau, s_{a_{j}} - s_{a_{j-1}}} \right) + \frac{\chi(\sigma(e_{j}) = 1)\chi(s_{a_{j}} \neq s_{a_{j-1}})}{\tau} S_{\tau, s_{a_{j}} - s_{a_{j-1}}} \right] (y_{b_{1}}; y_{b_{2}}).$$
(4.152)

Define $J_{\mathcal{P}} := \{j \in \{1, \ldots, q+1\} : \mathcal{J}_{\tau, e_j} \neq \hat{\mathcal{J}}_{\tau, e_j}\}$. Since $\sigma(e_1) = 1$ and $s_{b_1} \neq s_{a_1}$ (because $q \geq 1$), we have that $1 \in J_{\mathcal{P}}$. Moreover, $\sigma(e_{q+1}) = -1$, so $q+1 \notin J_{\mathcal{P}}$, so $1 \leq |J_{\mathcal{P}}| \leq q$. We rewrite (4.152) as

$$\sum_{I \subset J_{\mathcal{P}}} \left[\left(\prod_{j \in \{1, \dots, q\} \setminus I} G_{\tau, s_{a_{j}} - s_{a_{j-1}}} \right) \left(\prod_{j \in I} \frac{1}{\tau} S_{\tau, s_{a_{j}} - s_{a_{j-1}}} \right) \right] (y_{b_{1}}; y_{b_{2}}) \\ = \sum_{I \subset J_{\mathcal{P}}} \left[\left(\prod_{j \in \{1, \dots, q\} \setminus I} G_{\tau, 0} \right) \left(\prod_{j \in I} \frac{1}{\tau} S_{\tau, 0} \right) \right] (y_{b_{1}}; y_{b_{2}}) \\ = \sum_{I \subset J_{\mathcal{P}}} \frac{1}{\tau^{|I|}} \left(G_{\tau}^{q+1-|I|} \right) (y_{b_{1}}; y_{b_{2}}). \quad (4.153)$$

In the first inequality, we uses $\sum_{j=1}^{q+1} (s_{a_j} - s_{a_{j-1}}) = s_{a_{q+1}} - s_{a_0} = 0$, which is true since $s_{a_{q+1}} = s_{a_0} = 0$.

For $k \geq 2$, applying the Cauchy-Schwarz inequality to the operator kernels implies that

$$\begin{aligned} G_{\tau}^{k}(y_{b_{1}};y_{b_{2}}) \\ &:= \int_{\Lambda^{k-1}} \prod_{j=1}^{k-1} dx_{j} \, G_{\tau}(y_{b_{1}};x_{1}) G_{\tau}(x_{1};x_{2}) \dots G_{\tau}(x_{k-2};x_{k-1}) G_{\tau}(x_{k-1};y_{b_{2}}) \\ &\leq \|G_{\tau}\|_{\mathfrak{S}^{2}(\mathfrak{h})}^{k-2} \|G_{\tau}(y_{b_{1}},\cdot)\|_{\mathfrak{h}} \|G_{\tau}(y_{b_{2}},\cdot)\|_{\mathfrak{h}}. \end{aligned}$$
(4.154)

Applying (4.154), we deduce that (4.153) is

$$\leq \sum_{\substack{I \subset J_{\mathcal{P}} \\ |I| \leq q-1}} \frac{1}{\tau^{|I|}} \|G_{\tau}\|_{\mathfrak{S}^{2}(\mathfrak{h})}^{q+1-|I|} \|G_{\tau}(y_{b_{1}}, \cdot)\|_{\mathfrak{h}} \|G_{\tau}(y_{b_{2}}, \cdot)\|_{\mathfrak{h}} + \frac{q+1}{\tau^{q}} G_{\tau}(y_{b_{1}}; y_{b_{2}})$$

$$\leq C^{|V_{2}(\mathcal{P})|} \left(1 + \|G_{\tau}\|_{\mathfrak{S}^{2}(\mathfrak{h})}\right)^{|V_{2}(\mathcal{P})|} \left(\|G_{\tau}(y_{b_{1}}, \cdot)\|_{\mathfrak{h}} \|G_{\tau}(y_{b_{2}}, \cdot)\|_{\mathfrak{h}} + G_{\tau}(y_{b_{1}}; y_{b_{2}})\right).$$
(4.155)

Then (4.147) follows from (4.155) and (4.146).

To prove (4.148), we note that if q = 0, we have $\mathcal{J}_{\tau,e} = \hat{\mathcal{J}}_{\tau,e}$, since $\delta_{b_2} = -1$. In this case, (4.148) automatically holds. If $q \ge 1$, we argue as for (4.147), except we sum over *non-empty* subsets of $\mathcal{J}_{\mathcal{P}}$ in (4.153). This results in an extra factor of $\frac{1}{\tau}$ and one less power of $(1 + ||G_{\tau}||)_{\mathfrak{S}^2(\mathfrak{h})}$ in (4.155). (4.148) then follows from (4.146).

We can now bound the quantity (4.123).

Lemma 4.2.35. For $\Pi \in \mathfrak{P}$ and $\mathbf{t} \in \mathfrak{V}$, we have

$$\left| I_{\tau,\Pi}^{\xi}(\mathbf{t}) \right| \le C^{m+p} \|w\|_{L^{\infty}}^{3m} .$$
 (4.156)

Proof. We argue as in [29, Lemma 2.19]. We use the splitting (4.132) and Corollary 4.2.32 to rewrite (4.135) as

$$\left|I_{\tau,\Pi}^{\xi}(\mathbf{t})\right| \leq \|w\|_{L^{\infty}}^{3m} \int_{\Lambda^{V_1}} d\mathbf{y}_1 \left|\xi(\mathbf{y}_1)\right| \int_{\Lambda^{V_2}} d\mathbf{y}_2 \prod_{e \in E} \mathcal{J}_{\tau,e}(\mathbf{y}_e, \mathbf{s}).$$
(4.157)

Let us introduce the partition $\operatorname{Conn}(E) = \operatorname{Conn}_c(E) \sqcup \operatorname{Conn}_o(E)$. In other words, we partition $\operatorname{Conn}(E)$ into the closed connected paths $\operatorname{Conn}_c(E)$ and the open connected paths $\operatorname{Conn}_o(E)$. Then, we have

$$\int_{\Lambda^{V_2}} d\mathbf{y}_2 \prod_{e \in E} \mathcal{J}_{\tau,e}(\mathbf{y}_e, \mathbf{s}) = \prod_{\mathcal{P} \in \operatorname{Conn}_c(E)} \left(\int_{\Lambda^{V(\mathcal{P})}} \prod_{a \in V(\mathcal{P})} dy_a \prod_{e \in \mathcal{P}} \mathcal{J}_{\tau,e}(\mathbf{y}_e, \mathbf{s}) \right) \\ \times \prod_{\mathcal{P} \in \operatorname{Conn}_o(E)} \left(\int_{\Lambda^{V_2(\mathcal{P})}} \prod_{a \in V_2(\mathcal{P})} dy_a \prod_{e \in \mathcal{P}} \mathcal{J}_{\tau,e}(\mathbf{y}_e, \mathbf{s}) \right). \quad (4.158)$$

Substituting (4.158) into (4.157) and using (4.136), we have

$$|I_{\tau,\Pi}^{\xi}(\mathbf{t})| \leq C^{m} \|w\|_{L^{\infty}}^{3m} \int_{\Lambda^{V_{1}}} d\mathbf{y}_{1} |\xi(\mathbf{y}_{1})| \prod_{\mathcal{P}\in\operatorname{Conn}_{o}(E)} \left(\int_{\Lambda^{V_{2}(\mathcal{P})}} \prod_{a\in V_{2}(\mathcal{P})} dy_{a} \prod_{e\in\mathcal{P}} \mathcal{J}_{\tau,e}(\mathbf{y}_{e},\mathbf{s}) \right).$$

$$(4.159)$$

We note that (4.156) follows from (4.159) by applying the Cauchy-Schwarz inequality in the \mathbf{y}_1 variables, followed by (4.147). We also use that $|\text{Conn}_o(E)| = p$ to get the factor of C^p on the right-hand side of (4.156). Using Lemma 4.2.35, we can now bound the quantity $g_{\tau,m}^{\xi}(\mathbf{t})$ defined in (4.119).

Lemma 4.2.36. For $t \in \mathfrak{V}$, we have

$$\left|g_{\tau,m}^{\xi}(\mathbf{t})\right| \leq C^{m+p} \|w\|_{L^{\infty}}^{3m} (3m+p)!.$$

Proof. The claim follows from Lemma 4.2.23, (4.156), and the observation that $|\mathfrak{P}| \leq (3m+p)!$.

We can now bound the quantity $b_{\tau,m}^{\xi}$ given by (4.113) and (4.105) above and obtain (4.114) uniformly in $\xi \in \mathfrak{B}_p$.

Corollary 4.2.37. Uniformly in $\xi \in \mathfrak{B}_p$, we have

$$|b_{\tau,m}^{\xi}| \le (Cp)^p C^m (m!)^2 ||w||_{L^{\infty}}^{3m} =: C(m,p).$$
(4.160)

Proof. The claim follows from (4.118) and Lemma 4.2.36, after integrating over the simplex (4.120) (which gives a factor of $\frac{1}{m!}$) and using Stirling's formula.

Proof of (4.115) **uniformly in** $\xi \in \mathfrak{B}_p$

Let us make the following definition.

Definition 4.2.38. For $e = \{a, b\} \in E$, we define

$$\mathcal{J}_e(\mathbf{y}_e) := G(y_a; y_b) \,.$$

Proposition 1.7.12 implies the following lemma.

Lemma 4.2.39. The kernel of G is non-negative and symmetric.

Definition 4.2.40. For $\Pi \in \mathfrak{P}$, we define

$$I_{\Pi}^{\xi} := \int_{\Lambda^{V}} d\mathbf{y} \left(\prod_{i=1}^{m} w(y_{i,1} - y_{i,2}) w(y_{i,2} - y_{i,3}) w(y_{i,3} - y_{i,1}) \right) \xi(\mathbf{y}_{1}) \prod_{e \in E} \mathcal{J}_{e}(\mathbf{y}_{e}) .$$

$$(4.161)$$

Lemma 4.2.41. For each $\Pi \in \mathfrak{P}$ and $\mathbf{t} \in \mathfrak{V}$, we have

$$I_{\tau,m}^{\xi}(\mathbf{t}) \to I_m^{\xi} \text{ uniformly in } \xi \in \mathfrak{B}_p \text{ as } \tau \to \infty.$$
 (4.162)

Proof. The proof is similar to the proof of [29, Lemma 2.25]. For $\mathbf{t} \in \mathfrak{V}$, we define

$$\hat{I}_{\tau,\Pi}^{\xi}(\mathbf{t}) := \int_{\Lambda^{V}} d\mathbf{y} \left(\prod_{i=1}^{n} w(y_{i,1} - y_{i,2}) w(y_{i,2} - y_{i,3}) w(y_{i,3} - y_{i,1}) \right) \xi(\mathbf{y}_{1}) \prod_{e \in E} \hat{\mathcal{J}}_{\tau,e}(\mathbf{y}_{e}) . \quad (4.163)$$

Let first show that

$$\hat{I}^{\xi}_{\tau,\Pi}(\mathbf{t}) \to I^{\xi}_{\Pi}$$
 uniformly in $\xi \in \mathfrak{B}_p$ as $\tau \to \infty$. (4.164)

Namely, from (4.161) and (4.163), we have

$$\hat{I}_{\tau,\Pi}^{\xi}(\mathbf{t}) - I_{\Pi}^{\xi} = \int_{\Lambda^{V}} d\mathbf{y} \left(\prod_{i=1}^{m} w(y_{i,1} - y_{i,2}) w(y_{i,2} - y_{i,3}) w(y_{i,3} - y_{i,1}) \right) \xi(\mathbf{y}_{1}) \\ \times \left[\prod_{e \in E} \hat{\mathcal{J}}_{\tau,e}(\mathbf{y}_{e}) - \prod_{e \in E} \mathcal{J}_{e}(\mathbf{y}_{e}) \right]. \quad (4.165)$$

By telescoping, we can write

$$\prod_{e \in E} \hat{\mathcal{J}}_{\tau,e}(\mathbf{y}_{e}, \mathbf{s}) - \prod_{e \in E} \mathcal{J}_{e}(\mathbf{y}_{e}) = \sum_{e_{0} \in E} \left[\prod_{\substack{e \in E \\ e < e_{0}}} \hat{\mathcal{J}}_{\tau,e}(\mathbf{y}_{e}, \mathbf{s}) \left(\hat{\mathcal{J}}_{\tau,e_{0}}(\mathbf{y}_{e_{0}}, \mathbf{s}) - \mathcal{J}_{e_{0}}(\mathbf{y}_{e_{0}}) \right) \prod_{\substack{e \in E \\ e > e_{0}}} \mathcal{J}_{e}(\mathbf{y}_{e}) \right], \quad (4.166)$$

where we order the elements of E arbitrarily. Substituting (4.166) into (4.165), we have

$$\left| \hat{I}_{\tau,\Pi}^{\xi}(\mathbf{t}) - I_{\Pi}^{\xi} \right| \leq \sum_{e_0 \in E} \|w\|_{L^{\infty}}^{3m} \int_{\Lambda^V} d\mathbf{y} \left| \xi(\mathbf{y}_1) \right| \\ \times \left[\prod_{\substack{e \in E \\ e < e_0}} \hat{\mathcal{J}}_{\tau,e}(\mathbf{y}_e, \mathbf{s}) \left| \hat{\mathcal{J}}_{\tau,e_0}(\mathbf{y}_{e_0}, \mathbf{s}) - \mathcal{J}_{e_0}(\mathbf{y}_{e_0}) \right| \prod_{\substack{e \in E \\ e > e_0}} \mathcal{J}_{e}(\mathbf{y}_e) \right]. \quad (4.167)$$

Here, we have used that $\mathcal{J}_{\tau,e}(\mathbf{y}_e; \mathbf{s})$ and $\mathcal{J}_e(\mathbf{y}_e) \geq 0$ by Lemmas 4.2.24 and 4.2.39 (recalling Definitions 4.2.30 and 4.2.38).

Let us denote by $\sigma_{\tau,e_0}^{\xi}(\mathbf{t})$ the summand in (4.167) corresponding to e_0 . Since (4.167) is a finite sum, in order to obtain (4.2.41), it suffices to show that for each

 $e_0 \in E$, we have

$$\sigma_{\tau,e_0}^{\xi}(\mathbf{t}) \to 0 \text{ uniformly in } \xi \in \mathfrak{B}_p \text{ as } \tau \to \infty.$$
 (4.168)

We fix $e_0 \in E$. Let us define an integral kernel associated with an edge $e \in E$ by

$$\tilde{\mathcal{J}}_{\tau,e}(\mathbf{y}_e, \mathbf{s}) := \begin{cases} \hat{\mathcal{J}}_{\tau,e}(\mathbf{y}_e, \mathbf{s}) & \text{if } e < e_0, \\ \left| \hat{\mathcal{J}}_{\tau,e}(\mathbf{y}_e, \mathbf{s}) - \mathcal{J}_e(\mathbf{y}_e) \right| & \text{if } e = e_0, \\ \mathcal{J}_e(\mathbf{y}_e) & \text{if } e > e_0. \end{cases}$$
(4.169)

We have the following estimates for $\tilde{\mathcal{J}}_{\tau,e}$.

1. If $e = \{a, a\}$ (i.e. a loop in the graph) and $e \neq e_0$, then

$$\|\tilde{\mathcal{J}}_{\tau,e}(\cdot,\mathbf{s})\|_{\mathfrak{S}^{1}(\mathfrak{h})} \leq \|G_{\tau}\|_{\mathfrak{S}^{1}(\mathfrak{h})} + \|G\|_{\mathfrak{S}^{1}(\mathfrak{h})} \leq C, \qquad (4.170)$$

which holds by (4.138).

2. If $e = \{a, a\}$ and $e = e_0$, then

$$\begin{split} \|\tilde{\mathcal{J}}_{\tau,e}(\cdot,\mathbf{s})\|_{\mathfrak{S}^{1}(\mathfrak{h})} &= \int dy \, \left|\hat{\mathcal{J}}_{\tau,e_{0}}(\mathbf{y}_{e_{0}},\mathbf{s}) - \mathcal{J}_{e_{0}}(\mathbf{y}_{e_{0}})\right| \\ &= \int dy \, \left|\frac{1}{\tau(\mathrm{e}^{h/\tau}-1)}(y;y) - \frac{1}{h}(y;y)\right| = \int dy \sum_{k\in\mathbb{N}} \left(\frac{1}{\lambda_{k}} - \frac{1}{\tau(e^{\lambda_{k}/\tau}-1)}\right) \\ &= \|G_{\tau} - G\|_{\mathfrak{S}^{1}(\mathfrak{h})} \to 0, \quad (4.171) \end{split}$$

as $\tau \to \infty$ by spectral decomposition and the dominated convergence theorem. The third equality above follows by comparing Taylor series.

3. If $e = \{a, b\}$ with a < b and $e < e_0$, then $\tilde{\mathcal{J}}_{\tau,e}(\mathbf{y}_e, \mathbf{s}) = G_{\tau,\sigma(e)(s_a-s_b)}(y_a; y_b)$. Let us note that

$$\lim_{\tau \to \infty} (1+t) \left\| \frac{\mathrm{e}^{-th/\tau}}{\tau(\mathrm{e}^{h/\tau} - 1)} - h^{-1} \right\|_{\mathfrak{S}^{2}(\mathfrak{h})} \to 0, \qquad (4.172)$$

uniformly in $t \in (-1, 1)$. The claim (4.172) follows by a spectral argument; see [29, Lemma C.2] for the proof of a more general claim. Then (4.172) implies

$$\|\mathcal{J}_{\tau,e}(\cdot,\mathbf{s})\|_{\mathfrak{S}^{2}(\mathfrak{h})} \leq C_{\mathbf{s}},\qquad(4.173)$$

where the constant depends on \mathbf{s} .

4. If $e = \{a, b\}$ with a < b and $e = e_0$. Then

$$\tilde{\mathcal{J}}_{\tau,e}(\mathbf{y}_e, \mathbf{s}) = \left(G_{\tau,\sigma(e)(s_a - s_b)} - G \right) \left(y_a; y_b \right).$$
(4.174)

Therefore, (4.174)–(4.172) imply that

$$\|\tilde{\mathcal{J}}_{\tau,e}(\cdot,\mathbf{s})\|_{\mathfrak{S}^{2}(\mathfrak{h})} \to 0 \text{ as } \tau \to \infty.$$
(4.175)

5. If
$$e = \{a, b\}$$
 with $a < b$ and $e > e_0$. Then $\tilde{\mathcal{J}}_{\tau, e}(\mathbf{y}_e, \mathbf{s}) = G(y_a; y_b)$. Then
 $\|\tilde{\mathcal{J}}_{\tau, e}(\cdot, \mathbf{s})\|_{\mathfrak{S}^2(\mathfrak{h})} = \|G\|_{\mathfrak{S}^2(\mathfrak{h})} \le C$. (4.176)

Applying the same decomposition from the proof of Lemma 4.2.35, we have

$$\sigma_{\tau,e_0}^{\xi}(\mathbf{t}) = \|w\|_{L^{\infty}}^{3m} \int_{\Lambda^{V_1}} d\mathbf{y}_1 \prod_{\mathcal{P} \in \operatorname{Conn}_c(E)} \left(\int_{\Lambda^{V(\mathcal{P})}} \prod_{a \in V(\mathcal{P})} dy_a \prod_{e \in \mathcal{P}} \tilde{\mathcal{J}}_{\tau,e}(\mathbf{y}_e, \mathbf{s}) \right) \\ \times \prod_{\mathcal{P} \in \operatorname{Conn}_o(E)} \left(\int_{\Lambda^{V_2(\mathcal{P})}} \prod_{a \in V_2(\mathcal{P})} dy_a \prod_{e \in \mathcal{P}} \tilde{\mathcal{J}}_{\tau,e}(\mathbf{y}_e, \mathbf{s}) \right). \quad (4.177)$$

For $\mathcal{P} \in \operatorname{Conn}_{c}(E)$, apply the Cauchy-Schwarz inequality in y_{a} for $a \in V(\mathcal{P})$ to deduce that

$$\int_{\Lambda^{V(\mathcal{P})}} \prod_{a \in V(\mathcal{P})} dy_a \prod_{e \in \mathcal{P}} \tilde{\mathcal{J}}_{\tau,e}(\mathbf{y}_e, \mathbf{s}) \le \prod_{e \in \mathcal{P}} \|\tilde{\mathcal{J}}_{\tau,e}(\cdot, \mathbf{s})\|_{\mathfrak{S}^2(\mathfrak{h})}.$$
(4.178)

For $\mathcal{P} \in \operatorname{Conn}_{o}(E)$, we recall the notation from the proof of Lemma 4.2.34. Namely

$$\mathcal{P} = \{e_1, \dots, e_{q+1}\}, \quad V_1(\mathcal{P}) = \{b_1, b_2\}, \qquad V_2(\mathcal{P}) = \{a_1, \dots, a_q\}.$$

We apply the Cauchy-Schwarz inequality in the y_a for $a \in V_2(\mathcal{P})$ to obtain

$$\int_{\Lambda^{V_{2}(\mathcal{P})}} \prod_{a \in V_{2}(\mathcal{P})} dy_{a} \prod_{e \in \mathcal{P}} \tilde{\mathcal{J}}_{\tau,e}(\mathbf{y}_{e}, \mathbf{s}) \leq \prod_{j=2}^{q} \|\tilde{\mathcal{J}}_{\tau,e_{j}}(\cdot, \mathbf{s})\|_{\mathfrak{S}^{2}(\mathfrak{h})} \|\tilde{\mathcal{J}}_{\tau,e_{1}}(y_{b_{1}}, \cdot)\|_{\mathfrak{h}} \|\tilde{\mathcal{J}}_{\tau,e_{q+1}}(y_{b_{2}}, \cdot)\|_{\mathfrak{h}}.$$
(4.179)

Substituting (4.178) and (4.179) into (4.177) and applying the Cauchy-Schwarz in-

equality in the \mathbf{y}_1 variables yields

$$\sigma_{\tau,e_0}^{\xi}(\mathbf{t}) \le \|w\|_{L^{\infty}}^{3m} \|\xi\|_{\mathfrak{S}^2(\mathfrak{h})} \prod_{e \in E} \|\tilde{\mathcal{J}}_{\tau,e}(\cdot,\mathbf{s})\|_{\mathfrak{S}^2(\mathfrak{h})}.$$
(4.180)

Recalling (4.170)–(4.176) and Lemma 4.2.4, we obtain (4.168) from (4.180). Hence, (4.164) follows.

Let us also note that

$$I_{\tau,\Pi}^{\xi}(\mathbf{t}) \to \hat{I}_{\tau,\Pi}^{\xi}(\mathbf{t}) \text{ uniformly in } \xi \in \mathfrak{B}_p \text{ as } \tau \to \infty.$$
(4.181)

To obtain (4.181), we use a telescoping argument analogous to (4.166) above, perform a decomposition into open and closed paths as in the proof of Lemma 4.2.35, and use Lemmas 4.2.33 and 4.2.34. We omit the details, see [29, (2.62)]. We obtain the claim of Lemma 4.2.41 from (4.164) and (4.181).

For $m \in \mathbb{N}$, let us define

$$b_{\infty,m}^{\xi} := \frac{(-1)^m}{m! \, 3^m} \sum_{\Pi \in \mathfrak{B}} I_{\Pi}^{\xi} \,. \tag{4.182}$$

We now conclude the claimed convergence result.

Lemma 4.2.42. For $b_{\tau,m}^{\xi}$, b_m^{ξ} defined as in (4.113), we have

$$b_{\tau,m}^{\xi} \to b_m^{\xi} \text{ as } \tau \to \infty \text{ uniformly in } \xi \in \mathfrak{B}_p.$$
 (4.183)

Proof. By Lemma 4.2.23 and Proposition 4.2.35, we have that $g_{\tau,m}^{\xi}(\mathbf{t})$ is bounded uniformly in $\tau > 0, \xi \in \mathfrak{B}_p$, and $\mathbf{t} \in \mathfrak{V}$. Moreover, Lemmas 4.2.23 and 4.2.41 imply that for any $\mathbf{t} \in \mathfrak{V}$

$$g_{\tau,m}^{\xi}(\mathbf{t}) \to \sum_{\Pi \in \mathfrak{P}} I_{\Pi}^{\xi} \text{ as } \tau \to \infty \text{ uniformly in } \xi \in \mathfrak{B}_p.$$
 (4.184)

Recalling (4.118) and applying (4.182), as well as (4.184), combined with the dominated convergence theorem, we obtain that $b_{\tau,m}^{\xi} \to b_{\infty,m}^{\xi}$ as $\tau \to \infty$ uniformly in $\xi \in \mathfrak{B}_p$. The claim of the lemma follows by noting that, by Wick's theorem (Proposition 2.1.2), we have $b_{\infty,m}^{\xi} = b_m^{\xi}$. **Proof of** (4.114)–(4.115) **for** $\xi = \mathbf{1}_p$

To conclude the proof of Proposition 4.2.13, it remains to prove (4.114)–(4.115) for $\xi = \mathbf{1}_p$. We hence consider the operator with kernel

$$\xi(x_1, \dots, x_p; y_1, \dots, y_p) := \prod_{j=1}^p \delta(x_j - y_j).$$
(4.185)

To do this, we need to introduce a slightly modified version of the graphs defined in Definition 4.2.17 above. The modification is analogous to that used in [29, Section 4.2].

Definition 4.2.43. Let $m, p \in \mathbb{N}$ be given. We consider the same abstract vertex set $\Sigma \equiv \Sigma(m, p)$ with 6m + 2p elements and set of matchings of $\mathfrak{P} \equiv \mathfrak{P}(m, p)$ as in Definition 4.2.17. For each $\Pi \in \mathfrak{P}$, we consider a coloured multigraph $(\tilde{V}, \tilde{E}, \tilde{\sigma}) \equiv$ $(\tilde{V}_{\Pi}, \tilde{E}_{\Pi}, \tilde{\sigma}_{\Pi})$, with $\tilde{\sigma} : \tilde{E} \to {\pm 1}$, defined as follows.

1. We say $\alpha \sim \beta$ if and only if $i_{\alpha} = i_{\beta}$ and $r_{\alpha} = r_{\beta}$. We define the set $\tilde{V} := \{ [\alpha] : \alpha \in \Sigma \}$ and write $\tilde{V} = \tilde{V}_1 \sqcup \tilde{V}_2$, where

$$\tilde{V}_1 := \{(m+1,r) : r \in \{1,\ldots,p\}\}$$

and

$$V_2 := \{(i,r) : i \in \{1,\ldots,m\}, r \in \{1,2,3\}\}.$$

The set \tilde{V} inherits an order from the lexicographical ordering (given in Definition 4.2.17). Namely $[\alpha] \leq [\beta]$ if $\alpha \leq \beta$.

- 2. For $\Pi \in \mathfrak{P}$, we note that (α, β) induces an edge $e := \{[\alpha], [\beta]\}$ in \tilde{E} . Let us $\tilde{\sigma}(e) := \delta_{\beta}$. This is well-defined by construction.
- 3. Let $\operatorname{conn}(\tilde{E})$ denote the set of connected components of \tilde{E} . Note we can write $\tilde{E} = \bigsqcup_{\mathcal{P} \in \operatorname{conn}(\tilde{E})} \mathcal{P}$. We call the connected components \mathcal{P} of \tilde{E} paths.

Remark 4.2.44. In a slight abuse of notation, we denote the equivalence relation in Definition 4.2.27 and in Definition 4.2.43 by \sim . From context, it will always be clear to which equivalence relation we are referring. The same holds for the order \leq induced by lexicographical order on Σ .

Remark 4.2.45. The difference between the graph structure in Definition 4.2.43 and the one in Definition 4.2.27 is that, in the former, we identify the nodes corresponding to the observable ξ with kernel (4.185); see Figure 4.4.



Figure 4.4: An unpaired graph with m = 2, p = 3 corresponding to Definition 4.2.43.

By construction, every vertex in \tilde{V} has degree 2. Hence all paths $\mathcal{P} \in \operatorname{conn}(\tilde{E})$ are closed. For fixed $\Pi \in \mathfrak{P}$ and $\mathbf{t} \in \mathfrak{V}$, we define

$$I_{\tau,\Pi}^{\xi}(\mathbf{t}) := \int_{\Lambda^{\tilde{V}}} \prod_{a \in \tilde{V}} dy_a \left(\prod_{i=1}^m w(y_{i,1} - y_{i,2}) \, w(y_{i,2} - y_{i,3}) \, w(y_{i,3} - y_{i,1}) \right) \prod_{e \in \tilde{E}} \mathcal{J}_{\tau,e}(\mathbf{y}_e; \mathbf{s}) \,.$$

$$(4.186)$$

As in Definition 4.2.17, we consider the spatial labels $\mathbf{y} = (y_a)_{a \in \tilde{V}}$ and time labels $\mathbf{s} = (s_a)_{a \in \tilde{V}}$, except that we now adopt \tilde{V} as the vertex set and let the equivalence relation be the one from Definition 4.2.43 above. Moreover, we adopt the convention that $\mathbf{y}_i := (y_a)_{a \in \tilde{V}_i}$. Given $\mathcal{P} \in \operatorname{conn}(\tilde{E})$, we denote the set of vertices of \mathcal{P} by $\tilde{V}(\mathcal{P})$, and write $\tilde{V}_i(\mathcal{P}) := \tilde{V}(\mathcal{P}) \cap \tilde{V}_i$ for i = 1, 2, analogously as in Definition 4.2.29. Let us also define $\mathcal{J}_{\tau,e}$ analogously to (4.130), replacing (V, E) with (\tilde{V}, \tilde{E}) .

Lemma 4.2.46. Suppose that $\mathcal{P} \in \operatorname{conn}(E)$. Then

$$\int_{\Lambda^{\tilde{V}(\mathcal{P})}} \prod_{a \in \tilde{V}(\mathcal{P})} dy_a \prod_{e \in \mathcal{P}} \mathcal{J}_{\tau,e}(\mathbf{y}_e, \mathbf{s}) \le C^{|\tilde{V}(\mathcal{P})|} \,. \tag{4.187}$$

Proof. We follow the approach of the proof given in [29, Lemma 4.9]. We have two cases, depending on whether $\tilde{V}(\mathcal{P}) \subset \tilde{V}_2$ or $\tilde{V}_1(\mathcal{P}) \neq \emptyset$. In the first case, all the vertices lies in \tilde{V}_2 , so we can argue as in the proof of Lemma 4.2.33.

Let us henceforth assume that $\tilde{V}_1(\mathcal{P}) \neq \emptyset$. If $|\tilde{V}(\mathcal{P})| = 1$, then \mathcal{P} is a loop, and the left-hand side of (4.187) is $||G_\tau||_{\mathfrak{S}^1}$. Therefore, we can argue as in the proof of Lemma 4.2.33 to get the required bound. Let us now suppose that $|\tilde{V}(\mathcal{P})| > 1$. Since \mathcal{P} is a closed path, there exist $b_1 \dots, b_k \in \tilde{V}_1(\mathcal{P})$ such that $\mathcal{P} = \bigsqcup_{j=1}^k \mathcal{P}_j$, where for each $j = 1, \dots, k$, $\mathcal{P}_j = \{e_1^j, \dots, e_{q_j}^j\}$, with $b_j \in e_1^j$, $b_{j+1} \in e_{q_j}^j$, and $e_k^j \cap e_{k+1}^j \in \tilde{V}_2$ for $k \in \{1, \dots, q_{j-1}\}$. Since \mathcal{P} is closed, we set $b_{k+1} := b_1$. Let us note that if b_j and b_{j+1} are connected by a path of length one, then we have $q_j = 1$. Therefore, the left-hand side of (4.187) can be written as

$$\int_{\Lambda^k} dy_{b_1} \dots dy_{b_1} \prod_{j=1}^k \left[\int_{\Lambda^{\tilde{V}_2(\mathcal{P}_j)}} \prod_{a \in \tilde{V}_2(\mathcal{P}_j)} dy_a \prod_{e \in \mathcal{P}_j} \mathcal{J}_{\tau,e}(\mathbf{y}_e, \mathbf{s}) \right].$$
(4.188)

Arguing as in the proof of Lemma 4.2.34 (in particular as in the proof of (4.155)), we have that the j^{th} factor in (4.188) is less than or equal to

$$C^{|\tilde{V}_{2}(\mathcal{P}_{j})|}\left(1+\|G_{\tau}\|_{\mathfrak{S}^{2}(\mathfrak{h})}\right)^{|V_{2}(\mathcal{P}_{j})|}\left(\|G_{\tau}(y_{b_{j}},\cdot)\|_{\mathfrak{h}}\|G_{\tau}(y_{b_{j+1}},\cdot)\|_{\mathfrak{h}}+G_{\tau}(y_{b_{j}};y_{b_{j+1}})\right).$$
(4.189)

(4.187) follows from applying the Cauchy-Schwarz inequality in each of the y_{b_1}, \ldots, y_{b_k} variables, and using (4.146).

Let us note that Lemma 4.2.46 implies (4.114) for $\xi = \mathbf{1}_p$. For fixed $\Pi \in \mathfrak{P}$, we define

$$I_{\Pi}^{\xi} := \int_{\Lambda^{\tilde{V}}} d\mathbf{y} \left(\prod_{i=1}^{m} w(y_{i,1} - y_{i,2}) w(y_{i,2} - y_{i,3}) w(y_{i,3} - y_{i,1}) \right) \prod_{e \in \tilde{E}} \mathcal{J}_e(\mathbf{y}_e) \,. \tag{4.190}$$

Let $b_{\infty,m}^{\xi}$ be defined as in (4.182), where I_{Π}^{ξ} is now given by (4.190) instead of by (4.161). The same telescoping argument used in the proof of Lemma 4.2.41 (adapted to the framework of the proof of Lemma 4.2.46) implies that (4.115) holds for $\xi = \mathbf{1}_p$. We omit the details. This completes the proof of Proposition 4.2.13.

4.3 The time-independent problem with unbounded interaction potentials. Proof of Theorem 2.2.7

In this section, we consider w as in Assumption 2.2.2 above. Our goal is to prove Theorem 2.2.7 and thus complete the analysis of the time-independent problem outlined in Section 2.2.3. Let us first note the following claim, which follows from (4.109)-(4.110) and Theorem 2.2.6 by duality. **Lemma 4.3.1.** Suppose that $w \in L^{\infty}$ is even and real-valued (as in Assumption 2.2.1). Then, for all $p \in \mathbb{N}^*$, we have $\rho_{\tau}(\Theta_{\tau}(\xi)) \to \rho(\Theta(\xi))$ as $\tau \to \infty$ uniformly in $\xi \in C_p$. Here, we recall the definition (1.13) of C_p .

We now prove the following result for w as in Assumption 2.2.2 (the result in Lemma 4.3.1 will be applied for bounded approximations of the original interaction potential).

Lemma 4.3.2. Suppose that w is as in Assumption 2.2.2 and that $w^{\varepsilon} \in L^{\infty}$ is a sequence of interaction potentials as in Assumption 2.2.1 satisfying $w^{\varepsilon} \to w$ in $L^{3/2}$ as $\varepsilon \to 0$. Then, there is a sequence (ε_{τ}) tending to 0 as $\tau \to \infty$ such that for any $p \in \mathbb{N}^*$

$$\lim_{\tau \to \infty} \rho_{\tau}^{\varepsilon_{\tau}}(\Theta_{\tau}(\xi)) = \rho(\Theta(\xi))$$
(4.191)

uniformly in $\xi \in C_p$ with C_p defined as in (1.13).

Proof. By a diagonal argument, it suffices to prove that for fixed $\varepsilon > 0$, we have

$$\lim_{\tau \to \infty} \rho_{\tau}^{\varepsilon}(\Theta_{\tau}(\xi)) = \rho^{\varepsilon}(\Theta(\xi)) \tag{4.192}$$

uniformly in $\xi \in \mathcal{C}_p$, and that

$$\lim_{\varepsilon \to 0} \rho^{\varepsilon}(\Theta(\xi)) = \rho(\Theta(\xi)), \qquad (4.193)$$

uniformly in $\xi \in C_p$. The convergence (4.192) follows by noting that w^{ε} satisfies Assumption 2.2.1 for each fixed $\varepsilon > 0$ and by using Lemma 4.3.1.

Let us now prove the convergence (4.193). We recall (2.51) (and the corresponding analogue for interaction w^{ε}). Let us first show that

$$\lim_{\varepsilon \to 0} \mathcal{W}^{\varepsilon} = \mathcal{W}, \quad \mu\text{-almost surely}.$$
(4.194)

We have

$$3 |\mathcal{W} - \mathcal{W}^{\varepsilon}| \leq \int dx \, dy \, dz \, |\varphi(x)|^2 |\varphi(y)|^2 |\varphi(z)|^2 \\ \times |w(x-y)w(y-z)w(z-x) - w^{\varepsilon}(x-y)w^{\varepsilon}(y-z)w^{\varepsilon}(z-x)| \,. \tag{4.195}$$

Using the triangle inequality, we can rewrite the term inside the absolute value on

the right-hand side of (4.195) as

$$\leq |w(x-y)w(y-z)w(z-x) - w^{\varepsilon}(x-y)w(y-z)w(z-x)| + |w^{\varepsilon}(x-y)w(y-z)w(z-x) - w^{\varepsilon}(x-y)w^{\varepsilon}(y-z)w(z-x)| + |w^{\varepsilon}(x-y)w^{\varepsilon}(y-z)w(z-x) - w^{\varepsilon}(x-y)w^{\varepsilon}(y-z)w^{\varepsilon}(z-x)|.$$
(4.196)

To bound (4.195), we apply Lemma 4.1.3 separately to each of the terms obtained (4.196) to deduce that

$$|\mathcal{W} - \mathcal{W}^{\varepsilon}| \lesssim \|w - w^{\varepsilon}\|_{L^{\frac{3}{2}}} \left(\|w\|_{L^{\frac{3}{2}}}^{2} + \|w^{\varepsilon}\|_{L^{\frac{3}{2}}}^{2}\right) \|\varphi\|_{H^{\frac{1}{3}}}^{6}, \qquad (4.197)$$

We hence obtain (4.194) from (4.197), the Sobolev embedding theorem, and the fact that $w^{\varepsilon} \to w$ in $L^{3/2}$.

By using (4.28) and the fact that $w^{\varepsilon} \to w$ in $L^{3/2}$ as $\varepsilon \to 0$, it follows that there exists $c_0 > 0$ (depending on w) such that for $\varepsilon > 0$ sufficiently small, we have

$$\left| e^{-\mathcal{W}^{\varepsilon}} - e^{-\mathcal{W}} \right| \le 2 e^{c_0 \|\varphi\|_{H^{1/3}}^6}.$$
 (4.198)

By (4.29) and Assumption 2.2.4 (with K sufficiently small), we know

$$e^{c_0 \|\varphi\|_{H^{1/3}}^6} f^{\frac{1}{2}}(\mathcal{N}) \in L^1(d\mu).$$
(4.199)

By Lemma 4.2.2 and Assumption 2.2.4, we have

$$\Theta(\xi) f^{\frac{1}{2}}(\mathcal{N}) \in L^{\infty}(d\mu).$$
(4.200)

Combining (4.194) and (4.198)-(4.200), it follows that

$$\lim_{\varepsilon \to 0} \int d\mu \left| e^{-\mathcal{W}^{\varepsilon}} - e^{-\mathcal{W}} \right| \left| \Theta(\xi) \right| f(\mathcal{N}) = 0.$$
(4.201)

By analogous arguments, we obtain

$$\lim_{\varepsilon \to 0} z^{\varepsilon} = z \,. \tag{4.202}$$

We write

$$\rho^{\varepsilon}(\Theta(\xi)) - \rho(\Theta(\xi)) = \frac{1}{z} \int d\mu \left(\frac{z}{z^{\varepsilon}} e^{-\mathcal{W}^{\varepsilon}} - e^{-\mathcal{W}}\right) \Theta(\xi) f(\mathcal{N}), \qquad (4.203)$$

and deduce the claim of the lemma from (4.201)-(4.203). For the last step, we also

used (4.199) and $e^{-\mathcal{W}^{\varepsilon}} \leq e^{c_0 \|\varphi\|_{H^{1/3}}^6}$, which is proved as (4.198) above.

We now have all the tools necessary to prove Theorem 2.2.7.

Proof of Theorem 2.2.7. We deduce (2.76) (with the same subsequence (ε_{τ})) from Lemma 4.3.2 by a duality argument based on (4.109)–(4.110) as in the proof of Theorem 2.2.6.

The proof of (2.77) is similar. Let us first note that by (2.75), we have that for fixed $\varepsilon > 0$,

$$\lim_{\tau \to \infty} \mathcal{Z}_{\tau}^{\varepsilon} = z^{\varepsilon} \,. \tag{4.204}$$

The proof of (2.77) proceeds analogously as the proof of (2.76). The only difference is that instead of (4.192) and (4.193), we use (4.204) and (4.202) respectively.

4.4 The time-dependent problem. Proofs of Theorems 2.2.11 and 2.2.12

4.4.1 Bounded interaction potentials. Proof of Theorem 2.2.11

Let us consider w as in Assumption 2.2.1. To prove Theorem 2.2.11, we note the following analogues of the Schwinger-Dyson expansion results from [30, Sections 3.2 and 3.3].

Lemma 4.4.1 (Quantum Schwinger-Dyson Expansion). Let $\xi \in \mathcal{L}(\mathfrak{h}^{(p)}), \mathcal{K} > 0$, $\varepsilon > 0$, and $t \in \mathbb{R}$ be given. Then there exists $L(\mathcal{K}, \varepsilon, t, ||\xi||, p) \in \mathbb{N}$, $(e^l)_{l=0}^L$, where $e^l = e^l(x_i, t) \in \mathcal{L}(\mathfrak{h}^{(l)})$, and $\tau_0 = \tau_0(\mathcal{K}, \varepsilon, t, ||\xi||) > 0$ such that

$$\left\| \left(\Psi_\tau^t \Theta_\tau(\xi) - \sum_{l=0}^L \Theta_\tau(e^l) \right) \Big|_{\mathfrak{h}^{(\leq \mathcal{K}\tau)}} \right\| < \varepsilon$$

for all $\tau > \tau_0$.

Proof. We use the same proof and notation as [30, Lemma 3.9]. We begin by noting that all of the calculations up to [30, (3.44)] hold since they do not use the explicit form of \mathcal{W} . Instead of [29, 3.46], the same argument yields that the norm of $A_{\tau,\infty}^t(\xi)$ is bounded by

$$\frac{|t|^{j}}{j!}(p+j)^{j}2^{j}\left(\frac{n}{\tau}\right)^{p+j}\|w\|_{L^{\infty}}^{3}\|\xi\| \leq e^{p}\mathcal{K}^{p}\left(2e\mathcal{K}\|w\|_{L^{\infty}}^{3}|t|\right)^{j}\|\xi\|,$$

which differs only from [30, (3.47)] in that the $||w||_{L^{\infty}}$ is cubed. An analogous argument to [30, Lemma 3.9] yields the result. We refer the reader to [30, Lemma 3.9] for the precise definitions and arguments.

Lemma 4.4.2 (Classical Schwinger-Dyson Expansion). Let $\xi \in \mathcal{L}(\mathfrak{h}^{(p)}), \mathcal{K} > 0$, $\varepsilon > 0$, and $t \in \mathbb{R}$ be given. Then for $L(\mathcal{K}, \varepsilon, t, ||\xi||, p) \in \mathbb{N}$ and $\tau_0(\mathcal{K}, \varepsilon, t, ||\xi||) > 0$ chosen possibly larger than in Lemma 4.4.1 and the same choice of $e^l \in \mathcal{L}(\mathfrak{h}^{(l)})$ as in Lemma 4.4.1 we have

$$\left| \left(\Psi^t \Theta(\xi) - \sum_{l=0}^L \Theta(e^l) \right) \chi_{\{\mathcal{N} \le \mathcal{K}\}} \right| \le \varepsilon$$

for all $\tau \geq \tau_0$.

Proof. The proof is analogous to the proof of Lemma 4.4.1.

Proof of Theorem 2.2.11. By the proof of [30, Proposition 2.1], we note that the time-dependent claim follows from the corresponding time-independent claim, provided that we have the results of Lemmas 4.4.1 and 4.4.2 above. The time-independent claim was shown in Theorem 2.2.6. The claim now follows. \Box

Remark 4.4.3. The analogous reduction of the time-dependent result to the time-dependent result was also used in the proof of Theorem 2.1.11.

4.4.2 Unbounded interaction potentials. Proof of Theorem 2.2.12

In this section, we prove Theorem 2.2.12. Throughout the section, we consider w as in Assumption 2.2.2. Before proceeding with the proof of Theorem 2.2.12, we note an approximation result concerning the approximation of the flow of (2.45).

Let $s \in (0, \frac{1}{2})$ be given. We denote by $\mathcal{G} \subset H^s(\Lambda)$ the set constructed in Proposition 4.1.2 (ii) above. We know $\mathbb{P}^f_{\text{Gibbs}}(\mathcal{G}) = 1$ and initial data in \mathcal{G} gives rise to global solutions of (4.1). Here, we recall (2.55).

Lemma 4.4.4. Let $w^{\varepsilon} \in L^{\infty}(\Lambda)$ be a sequence such that

$$\lim_{\varepsilon \to 0} \|w^{\varepsilon} - w\|_{L^{3/2}} = 0.$$
(4.205)

Furthermore, fix $s \in (0, \frac{1}{2})$, T > 0, and consider $\psi \in \mathcal{G}$, with $\mathcal{G} \in H^s(\Lambda)$ as above.

(i) For $\varepsilon > 0$, the following Cauchy problem is well-posed on [-T, T].

$$\begin{cases} i\partial_t u^{\varepsilon} + (\Delta - \kappa)u^{\varepsilon} = \int dy \, dz \, w^{\varepsilon} (x - y)w^{\varepsilon} (y - z)w^{\varepsilon} (z - x)|u^{\varepsilon}(y)|^2 |u^{\varepsilon}(z)|^2 u^{\varepsilon}(x) \\ u_0^{\varepsilon} = \psi \,. \end{cases}$$

$$(4.206)$$

(ii) We have

$$\lim_{\varepsilon \to 0} \|u^{\varepsilon} - u\|_{L^{\infty}_{[-T,T]}H^s} = 0, \qquad (4.207)$$

where

$$\begin{cases} i\partial_t u + (\Delta - \kappa)u = \int dy \, dz \, w(x - y)w(y - z)w(z - x)|u(y)|^2 |u(z)|^2 u(x) \\ u_0 = \psi \,. \end{cases}$$
(4.208)

Proof. As in the proof of Proposition 4.1.1, it suffices to consider the case $\kappa = 0$. By symmetry, it suffices to consider only positive times. Using [9, Lemma 4.4] and [9, (4.10)], we have a more quantitative description of the set \mathcal{G} in Proposition 4.1.2 (ii) above⁵. Namely, given $\nu > 0$, we can write

$$\mathcal{G} = \bigcup_{\eta, A > 0} \mathcal{K}_{\eta, A} \,, \tag{4.209}$$

where

$$\mathcal{K}_{\eta,A} := \left\{ \psi \in H^s(\Lambda) : \|\psi\|_{L^2}^2 \le K, \quad \|u(t)\|_{H^s} \le A \log\left(\frac{1+|t|}{\eta}\right)^{s+\nu} \right\}.$$
(4.210)

In (4.210), the constant K is as in Assumption 2.2.4 and u is the solution of (4.208) with initial data ψ .

By (4.209), we deduce that the claim follows if we show that it holds for $\psi \in \mathcal{K}_{\eta,A}$ with $\eta, A > 0$ fixed. Throughout the proof, we consider ε sufficiently small such that

$$\|w^{\varepsilon}\|_{L^{3/2}} \le 2\|w\|_{L^{3/2}}.$$
(4.211)

For fixed η, A , we let

$$\mathcal{A} := A \log\left(\frac{1+|T|}{\eta}\right)^{s+\nu}.$$
(4.212)

⁵Once we have the setup of Section 4.1, we can directly apply these arguments from [9].

We fix $\theta > 0$ small and consider

$$b = \frac{1}{2} + \theta \,. \tag{4.213}$$

With parameters as above, we consider $\delta > 0$ such that

$$\delta^{\theta} \|w\|_{L^{3/2}}^{3} (\mathcal{A}+1)^{4} \ll 1, \qquad (4.214)$$

where the smallness of the right-hand side of (4.214) will be determined later (depending on the other parameters above). For the remainder of the proof, we let

$$n = \lfloor T/\delta \rfloor. \tag{4.215}$$

Let us now show that the following properties hold for k = 0, 1, ..., n - 1.

(1) For $\varepsilon > 0$ small, we have $u^{\varepsilon} \in L^{\infty}_{[0,k\delta]}H^s_x$ and

$$\lim_{\varepsilon \to 0} \|u^{\varepsilon}(k\delta) - u(k\delta)\|_{H^s} = 0.$$
(4.216)

(2) There exists $\varepsilon_k > 0$ such that for all $\varepsilon \in (0, \varepsilon_k)$, and for a constant C > 0 independent of k, we have $u^{\varepsilon} \in X^{s,b}_{[k\delta,(k+1)\delta]}$ and

$$\begin{aligned} \|u^{\varepsilon} - u\|_{X^{s,b}_{[k\delta,(k+1)\delta]}} &\leq C \|u^{\varepsilon}(k\delta) - u(k\delta)\|_{H^{s}} \\ &+ C\delta^{\theta} \|w - w^{\varepsilon}\|_{L^{3/2}} \|w\|_{L^{3/2}}^{2} \left(\mathcal{A} + 1\right)^{5}. \end{aligned}$$
(4.217)

We show (4.216)–(4.217) by induction on k.

Base We consider k = 0. Note that (4.216) is automatically satisfied since $u^{\varepsilon}(0) = u(0) = \psi$. Using (4.211)–(4.214) and arguing analogously as in the proof of Proposition 4.1.1, we deduce that

$$u^{\varepsilon} \in X^{s,b}_{[0,\delta]}, \qquad \left\| u^{\varepsilon} \right\|_{X^{s,b}_{[0,\delta]}} \lesssim \mathcal{A}.$$

$$(4.218)$$

Let us note that we also have

$$u \in X^{s,b}_{[0,\delta]}, \qquad \|u\|_{X^{s,b}_{[0,\delta]}} \lesssim \mathcal{A},$$
 (4.219)

by the same argument.

We use Duhamel's principle for the difference equation solved by $u^{\varepsilon}-u$ and write for $t\in[0,\delta]$

$$u^{\varepsilon}(t) - u(t) =$$

$$-i \int_{0}^{t} dt' e^{i(t-t')\Delta} \left(\int dy \, dz \, w^{\varepsilon}(x-y) \, w^{\varepsilon}(y-z) \, w^{\varepsilon}(z-x) |u^{\varepsilon}(y)|^{2} |u^{\varepsilon}(z)|^{2} u^{\varepsilon}(x) \right)$$

$$- \int dy \, dz \, w(x-y) \, w(y-z) \, w(z-x) \, |u(y)|^{2} \, |u(z)|^{2} \, u(x) \right) \quad (4.220)$$

By using telescoping, Lemma 1.7.6 (iii), Lemma 4.1.4, and (4.211), we deduce from (4.220) that

$$\begin{aligned} \|u^{\varepsilon} - u\|_{X^{s,b}_{[0,\delta]}} &\leq C\delta^{\theta} \, \|w^{\varepsilon} - w\|_{L^{3/2}} \, \|w\|^{2}_{L^{3/2}} \left(\|u^{\varepsilon}\|^{5}_{X^{s,b}_{[0,\delta]}} + \|u\|^{5}_{X^{s,b}_{[0,\delta]}} \right) \\ &+ C\delta^{\theta} \, \|w\|^{3}_{L^{3/2}} \, \|u^{\varepsilon} - u\|_{X^{s,b}_{[0,\delta]}} \left(\|u^{\varepsilon}\|^{4}_{X^{s,b}_{[0,\delta]}} + \|u\|^{4}_{X^{s,b}_{[0,\delta]}} \right). \end{aligned}$$
(4.221)

Using (4.218)–(4.219), followed by (4.214) (with sufficiently small right-hand side), we obtain from (4.221) that

$$\|u^{\varepsilon} - u\|_{X^{s,b}_{[0,\delta]}} \le C\delta^{\theta} \|w^{\varepsilon} - w\|_{L^{3/2}} \|w\|_{L^{3/2}}^{2} \mathcal{A}^{5}, \qquad (4.222)$$

with a different choice of C. We obtain (4.217) from (4.222).

Inductive Step Suppose that (4.216)–(4.217) hold for some $0 \le k \le n-2$. Let us observe that by Lemma 1.7.6 (i) and (4.217) for k, we have that for $\varepsilon \in (0, \varepsilon_k)$

$$\begin{aligned} \|u^{\varepsilon}((k+1)\delta) - u((k+1)\delta)\|_{H^{s}} &\lesssim \|u^{\varepsilon} - u\|_{X^{s,b}_{[k\delta,(k+1)\delta]}} \\ &\leq C \|u^{\varepsilon}(k\delta) - u(k\delta)\|_{H^{s}} + C\delta^{\theta} \|w^{\varepsilon} - w\|_{L^{3/2}} \|w\|_{L^{3/2}}^{2} (\mathcal{A}+1)^{5}. \end{aligned}$$
(4.223)

We deduce

$$\lim_{\varepsilon \to 0} \|u^{\varepsilon}((k+1)\delta) - u((k+1)\delta)\|_{H^s} = 0$$

$$(4.224)$$

from (4.223), combined with (4.216) for k and (4.205). This shows that (4.216) holds for k + 1.

By (4.224), it follows that there exists $\varepsilon_{k+1} \in (0, \varepsilon_k)$ small enough such that for all $\varepsilon \in (0, \varepsilon_{k+1})$, we have

$$\|u^{\varepsilon}((k+1)\delta) - u((k+1)\delta)\|_{H^{s}} \le 1.$$
(4.225)

From (4.225), we deduce that for all $\varepsilon \in (0, \varepsilon_{k+1})$, we have

$$\|u^{\varepsilon}((k+1)\delta)\|_{H^{s}} \le \|u((k+1)\delta)\|_{H^{s}} + 1 \le \mathcal{A} + 1.$$
(4.226)

For (4.226), we recalled 4.210 and (4.212) above. Using (4.226), recalling (4.211), (4.213)–(4.214) and arguing analogously as in the proof of Proposition 4.1.1, we deduce that

$$u^{\varepsilon} \in X^{s,b}_{[(k+1)\delta,(k+2)\delta]}, \qquad \|u^{\varepsilon}\|_{X^{s,b}_{[(k+1)\delta,(k+2)\delta]}} \lesssim \mathcal{A} + 1.$$
 (4.227)

As in (4.219) (with the same argument), we have

$$u \in X^{s,b}_{[(k+1)\delta,(k+2)\delta]}, \qquad ||u||_{X^{s,b}_{[(k+1)\delta,(k+2)\delta]}} \lesssim \mathcal{A}.$$
 (4.228)

Similarly as for (4.220), we use Duhamel's principle to write for $t \in [(k+1)\delta, (k+2)\delta]$

$$u^{\varepsilon}(t) - u(t) = e^{i(t - (k+1)\delta)\Delta} \left(u^{\varepsilon}((k+1)\delta) - u((k+1)\delta) \right)$$

$$-i \int_{(k+1)\delta}^{t} dt' e^{i(t-t')\Delta} \left(\int dy \, dz \, w^{\varepsilon}(x-y) \, w^{\varepsilon}(y-z) \, w^{\varepsilon}(z-x) |u^{\varepsilon}(y)|^{2} |u^{\varepsilon}(z)|^{2} u^{\varepsilon}(x) \right)$$

$$- \int dy \, dz \, w(x-y) \, w(y-z) \, w(z-x) \, |u(y)|^{2} \, |u(z)|^{2} \, u(x) \right). \quad (4.229)$$

Using Lemma 1.7.6 (ii) to estimate the first term on the right-hand side of (4.229), and using telescoping, Lemma 1.7.6 (iii), Lemma 4.1.4, and (4.211) (as for (4.221)) to estimate the second term, we deduce that

$$\begin{aligned} \|u^{\varepsilon} - u\|_{X_{[(k+1)\delta,(k+2)\delta]}^{s,b}} &\leq C \|u^{\varepsilon}((k+1)\delta) - u((k+1)\delta)\|_{H^{s}} \\ &+ C\delta^{\theta} \|w^{\varepsilon} - w\|_{L^{3/2}} \|w\|_{L^{3/2}}^{2} \left(\|u^{\varepsilon}\|_{X_{[(k+1)\delta,(k+2)\delta]}^{s,b}} + \|u\|_{X_{[(k+1)\delta,(k+2)\delta]}^{s,b}} \right) \\ &+ C\delta^{\theta} \|w\|_{L^{3/2}}^{3} \|u^{\varepsilon} - u\|_{X_{[(k+1)\delta,(k+2)\delta]}^{s,b}} \left(\|u^{\varepsilon}\|_{X_{[(k+1)\delta,(k+2)\delta]}^{s,b}} + \|u\|_{X_{[(k+1)\delta,(k+2)\delta]}^{s,b}} \right). \end{aligned}$$

$$(4.230)$$

We now use (4.227)–(4.228) and (4.214) (for sufficiently small right-hand side) to deduce that

$$\begin{aligned} \|u^{\varepsilon} - u\|_{X^{s,b}_{[(k+1)\delta,(k+2)\delta]}} &\leq C \|u^{\varepsilon}((k+1)\delta) - u((k+1)\delta)\|_{H^{s}} \\ &+ C\delta^{\theta} \|w^{\varepsilon} - w\|_{L^{3/2}} \|w\|_{L^{3/2}}^{2} \left(\mathcal{A} + 1\right)^{5}, \quad (4.231)\end{aligned}$$

for a different choice of C. From (4.231), we conclude the induction.

Using (4.216)–(4.217), Lemma 1.7.6 (i), and (4.205), it follows that

$$\lim_{\varepsilon \to 0} \|u^{\varepsilon} - u\|_{L^{\infty}_{[0,n\delta]}H^s_x} = 0, \qquad (4.232)$$

for n as in (4.215). By using $||u||_{L^{\infty}_{[n\delta,T]}H^s_x} \leq \mathcal{A}$, and by repeating the above argument, we also obtain

$$\lim_{\varepsilon \to 0} \|u^{\varepsilon} - u\|_{L^{\infty}_{[n\delta,T]}H^s_x} = 0.$$
(4.233)

The lemma follows from (4.232)-(4.233).

Remark 4.4.5. The proof of Lemma 4.4.4 above is more involved than that of the analogous cubic results [30, Proposition 5.1] and Lemma 3.5.4. It requires full use of the set \mathcal{G} of initial data leading to global solutions of (2.45).

Let us recall the following form of the diagonal argument [30, Lemma 5.5].

Lemma 4.4.6 (Diagonal argument). Let (Γ_k) be a sequence of sets with $\Gamma_k \subset \Gamma_{k+1}$ and let $\Gamma := \bigcup_k \Gamma_k$. Given $\varepsilon, \tau > 0$, let $g, g^{\varepsilon}, g^{\varepsilon}_{\tau} : \Gamma \to \mathbb{C}$ be functions satisfying the following properties.

- (i) For $k \in \mathbb{N}$ and $\varepsilon > 0$ fixed, we have $\lim_{\tau \to \infty} g_{\tau}^{\varepsilon}(\zeta) = g^{\varepsilon}(\zeta)$, uniformly in $\zeta \in \Gamma_k$.
- (ii) For fixed $k \in \mathbb{N}$, we have $\lim_{\varepsilon \to 0} g^{\varepsilon}(\zeta) = g(\zeta)$, uniformly in $\zeta \in \Gamma_k$.

Then, there exists a sequence (ε_{τ}) converging to zero as $\tau \to \infty$ such that

$$\lim_{\tau \to \infty} g_{\tau}^{\varepsilon_{\tau}}(\zeta) = g(\zeta) \,.$$

We now have all of the necessary tools to prove Theorem 2.2.12.

Proof of Theorem 2.2.12. Once we have Lemma 4.4.4 at our disposal, the proof is quite similar to that of Theorem 2.1.12. We just outline the main differences and refer the reader to Section 3.5.2 for more details. We adopt the convention that the superscript ε denotes an object defined using the interaction potential w^{ε} . Using Theorem 2.2.11, Lemma 4.4.6, and arguing analogously as in the proof of Theorem 2.1.12, the claim follows if we show that

$$\lim_{\varepsilon \to 0} \tilde{\rho}_1^{\varepsilon} \left(\Psi^{t_1,\varepsilon} \Theta(\xi^1) \cdots \Psi^{t_m,\varepsilon} \Theta(\xi^m) \right) = \tilde{\rho}_1 \left(\Psi^{t_1} \Theta(\xi^1) \cdots \Psi^{t_m} \Theta(\xi^m) \right) , \qquad (4.234)$$

uniformly in Γ_k

$$\Gamma_k := \left\{ (m, t_i, p_i, \xi^i) : m \le k, \, |t_i| \le k, \, p_i \in \mathbb{N}^*, p_i \le k, \, \|\xi^i\| \le k, \, 1 \le i \le m \right\}.$$
(4.235)

In (4.235), we take $\xi^i \in \mathcal{L}(\mathfrak{h}^{(p_i)})$. We also recall (2.62), (4.87), and define the quantities with superscript ε accordingly.

Let φ denote the classical free field given by (2.12). Recall we have that for $s \in (0, \frac{1}{2}), \varphi \in H^s \mu$ -almost surely. Let us recall the definition (2.59) of $\Theta(\xi)$ and the definition (2.50) of the flow map $S_t(\cdot)$ of (2.45). By suitably defining the quantities with superscript ε , we have for $\xi \in \mathcal{L}(\mathfrak{h}^{(k)})$

$$\Psi^{t,\varepsilon}\Theta(\xi) = \left\langle \left(S_t^{\varepsilon}\varphi\right)^{\otimes_k}, \xi\left(S_t^{\varepsilon}\varphi\right)^{\otimes_k}\right\rangle_{\mathfrak{h}^{\otimes_k}}, \quad \Psi^t\Theta(\xi) = \left\langle \left(S_t\varphi\right)^{\otimes_k}, \xi\left(S_t\varphi\right)^{\otimes_k}\right\rangle_{\mathfrak{h}^{\otimes_k}}.$$
(4.236)

When $\xi \in \mathcal{L}(\mathfrak{h}^{(k)})$, Lemma 4.4.4 and (4.236) imply that

$$\lim_{\varepsilon \to 0} \Psi^{t,\varepsilon} \Theta(\xi) = \Psi^t \Theta(\xi) , \qquad (4.237)$$

 μ -almost surely.

Recalling (4.194), and using (4.237), it follows that

$$\lim_{\varepsilon \to 0} \Psi^{t_1,\varepsilon} \Theta(\xi^1) \cdots \Psi^{t_m,\varepsilon} \Theta(\xi^m) \,\mathrm{e}^{-\mathcal{W}^{\varepsilon}} = \Psi^{t_1} \Theta(\xi^1) \cdots \Psi^{t_m} \Theta(\xi^m) \,\mathrm{e}^{-\mathcal{W}} \,, \qquad (4.238)$$

 μ -almost surely.

Using Lemma 4.1.3, Lemma 4.2.2, conservation of mass for (4.206) and (4.208), as well as (4.205), we have the following bounds for $\varepsilon > 0$ sufficiently small.

$$\left|\Psi^{t_{1},\varepsilon}\Theta(\xi^{1})\cdots\Psi^{t_{m},\varepsilon}\Theta(\xi^{m})\mathrm{e}^{-\mathcal{W}^{\varepsilon}}f\left(\mathcal{N}\right)\right| \leq \left(\prod_{j=1}^{m} \|\xi^{j}\| \|\varphi\|_{\mathfrak{h}}^{2p_{j}}\right)\mathrm{e}^{c\|w\|_{L^{3/2}}^{3}\|\varphi\|_{H^{1/3}}^{3}}f\left(\mathcal{N}\right)$$
$$\left|\Psi^{t_{1}}\Theta(\xi^{1})\ldots\Psi^{t_{m}}\Theta(\xi^{m})e^{-\mathcal{W}}f\left(\mathcal{N}\right)\right| \leq \left(\prod_{j=1}^{m} \|\xi^{j}\| \|\varphi\|_{\mathfrak{h}}^{2p_{j}}\right)\mathrm{e}^{c\|w\|_{L^{3/2}}^{3}\|\varphi\|_{H^{1/3}}^{3}}f\left(\mathcal{N}\right).$$
$$(4.239)$$

The claim now follows from (4.238)–(4.239), by using Proposition 4.1.2 (i), Assumption 2.2.4 (with K sufficiently small) and the dominated convergence theorem.

Appendix A

Outline of the Construction of the Gibbs Measure for the Cubic NLS

In this section we recall the construction of the Gibbs measure for the cubic NLS in one dimension in more detail, following [9]. We recall from (1.5) that for non-negative interaction potential w, we heuristically define

$$d\mathbb{P}_{\text{Gibbs}} = \frac{1}{z_{\text{Gibbs}}} e^{-H(u)} du, \qquad (A.1)$$

where $z_{\text{Gibbs}} \equiv z$ is a normalisation constant, H is the Hamiltonian and du is the formally defined infinite dimensional Lebesgue measure on the space of fields. We begin with some formal arguments to establish why we expect the measure to be invariant, and to establish the support of the measure. To rigorously construct the Gibbs measure, we need the following.

- 1. A way of realising the infinite Lebesgue measure as a rigorously defined infinite dimensional measure times a weight function.
- 2. To show that the weight function mentioned above is an L^1 function with respect to the infinite dimensional measure, allowing us to define z.
- 3. To establish the invariance of the measure.

To give a brief outline of how we establish 1. - 3., we will write the Gibbs measure as a weighted Wiener measure. In the case of a defocusing potential, we can use the Sobolev embedding theorem to show that z is well defined, and we use a Galerkin like approximation to establish the invariance of the measure. We also include the arguments for a focusing potential, which requires us to truncate the Gibbs measure defined in (A.1).

A.1 Formal arguments

In this section, we consider the defocusing cubic NLS given by

$$\begin{cases} i\partial_t u + (\Delta - \kappa)u = |u|^2 u\\ u(x, 0) = u_0(x), \end{cases}$$
(A.2)

which has associated Hamiltonian

$$H(u) = \int dx \, |\nabla u(x)|^2 + \kappa |u(x)|^2 + \frac{1}{2} \int dx \, |u(x)|^4$$

We can write (A.2) as an infinite dimensional Hamiltonian system. Namely

$$u_t = -i\frac{\partial H}{\partial \bar{u}},\tag{A.3}$$

Thus, drawing analogy with the finite dimensional system, Liouville's theorem means we would formally expect the measure $d\mathbb{P}_{\text{Gibbs}}$ to be invariant under the flow of (A.2).

We rewrite

$$d\mathbb{P}_{\text{Gibbs}} = \mathrm{e}^{-\int dx \, |u|^4} d\nu,$$

where $d\nu$ is the unormalised Wiener measure given by

$$d\nu := \mathrm{e}^{-\int dx \, |\nabla u|^2} du.$$

We define the normalised Wiener measure by

$$d\mu := \frac{1}{Z_{\text{Wiener}}} e^{-\int dx \, |\nabla u|^2} du,$$

where Z_{Wiener} is the normalisation constant associated with the Wiener measure. We note that in one dimension, the Wiener measure is equivalent to Brownian motion, and in higher dimensions it is a Gaussian Free Field (GFF). For a review of the GFF, we direct the reader to [74]. We wish to understand, heuristically, what functions lie in the support of the Wiener measure. To do this, we write $a_k := \hat{u}(k)$, and use Plancherel's theorem to write

$$\int_{\mathbb{T}^d} |\nabla u|^2 \, dx = c \sum_{k \in \mathbb{Z}^d} |k|^2 |a_k|^2.$$

Putting $\kappa = 0$, we have

$$d\mu = \frac{\exp\left(-c\sum_{k\in\mathbb{Z}^d} |k|^2 |a_k|^2\right) \prod_{k\in\mathbb{Z}^d} da_k}{\int \exp\left(-c\sum_{k\in\mathbb{Z}^d} |k|^2 |a_k|^2\right) \prod_{k\in\mathbb{Z}^d} da_k} \\ = \prod_{k\in\mathbb{Z}^d} \frac{\exp\left(-c|k|^2 |a_k|^2\right) da_k}{\int \exp\left(-c|k|^2 |a_k|^2\right) da_k}.$$

So we get a Gaussian distribution for

$$|k|a_k = |k|\hat{u}(k).$$

Thus we can think of each Fourier coefficient as a random variable, $\hat{u}(k) = g_k(\omega)/|k|$. So each φ in the support of $d\rho$ has the random Fourier series

$$\varphi(x) = \varphi^{\omega}(x) = \frac{1}{2\pi} \sum_{k \in \mathbb{Z}^d} \frac{\omega_k}{|k|} e^{2\pi i \langle k, x \rangle},$$
(A.4)

where (ω_k) are suitable i.i.d. complex Gaussian random variables. To get around the problem at the zero mode in (A.4), we take $\kappa > 0$. Repeating the computation for a non-zero value of κ , we find that a typical element in the support of μ is given by the random Fourier series

$$\varphi(x) = \varphi^{\omega}(x) = \sum_{k \in \mathbb{Z}^d} \frac{\omega_k}{\sqrt{\lambda_k}} e^{2\pi i \langle k, x \rangle},$$
(A.5)

where we recall $\lambda_k := 4\pi^2 |k|^2 + \kappa$ are the eigenvalues of the one-body Hamiltonian $h = -\Delta + \kappa$. Thus we recover (2.12). By Wick's theorem, we calculate that

$$\mathbb{E}_{\mu}\left[\|\varphi\|_{H^{s}(\mathbb{T}^{d})}^{2}\right] = \sum_{k \in \mathbb{Z}^{d}} \frac{\mathbb{E}[|\omega_{k}|^{2}]}{\lambda_{k}} (1+|k|^{2})^{s} \sim \sum_{k \in \mathbb{Z}^{d}} |k|^{2s-2}.$$

Comparing with the harmonic series, we find that this is summable if and only if $s < 1 - \frac{d}{2}$. So it follows that $\varphi \in H^s(\mathbb{T}^d)$ almost surely for s < 1 - d/2. So $d\mu$ is a probability measure on $H^s(\mathbb{T}^d)$ for any such s, and $d\mu(H^s(\mathbb{T}^d)) = 0$ for any $s \ge 1 - d/2$. For more details, we direct the reader to the exposition in [49, Section 3.1] and the classical texts [46,77].

We now fix d = 1 in (A.2) and continue to consider the defocusing case. Then the Sobolev embedding theorem implies that $\frac{2}{p} \int |u|^p dx$ is finite almost surely for $u = \varphi^{\omega}$. Therefore, $e^{-\frac{2}{p} \int |u|^p dx}$ is in $L^1(d\omega)$. Similarly, for a pointwise non-negative $V \in L^1(\mathbb{T})$, the Sobolev embedding theorem and Young's inequality imply that $e^{-\frac{1}{2} \int (V*|u|^2)|u|^2 dx}$ is in $L^1(d\omega)$. So, if we can make our arguments rigorous, then using the Radon-Nikodym theorem, we will be able to define the Gibbs measure for the defocusing NLS and defocusing Hartree equation using the Wiener measure. By normalising the Gibbs measure, we can show it is an invariant probability measure on $H^s(\mathbb{T})$ for s < 1/2. The focusing case and corresponding Hartree equation require more work to show the density functions are in L^1 , and are considered later in the appendix.

A.2 Rigorous arguments

We henceforth fix d = 1. We want to make our arguments in the previous section rigorous. To do this, we will consider the truncated NLS given by

$$\begin{cases} iv_t^N + \Delta v^N = \pm P_N \left(|v^N|^{p-2} v^N \right) \\ v_0^N = P_N \varphi. \end{cases}$$
(A.6)

We note that for $\varphi \in H^s(\mathbb{T})$, $P_N \varphi \in H^\infty(\mathbb{T})$, so (A.6) is locally well-posed for $p \in [4, 6]$. We then define the Wiener and Gibbs measures associated with (A.6) as truncations of what we had previously, so

$$d\nu_N := \prod_{|k| \le N} \exp\left(-c|k|^2 |a_k|^2\right) da_k,$$
$$d\mu_N := \prod_{|k| \le N} \frac{\exp\left(-c|k|^2 |a_k|^2\right) da_k}{\int \exp\left(-c|k|^2 |a_k|^2\right) da_k}$$
$$d\mathbb{P}_{\text{Gibbs},N} := \frac{\exp\left(\pm \frac{1}{p} \int_{\mathbb{T}} |P_N\left(u\right)|^p dx\right) d\nu_N}{\int \exp\left(\pm \frac{1}{p} \int_{\mathbb{T}} |P_N\left(u\right)|^p dx\right) d\nu_N}$$

We note that all of the previously formal arguments are rigorous in the case of $d\mu_N$, since we are working in a finite dimensional space (\mathbb{C}^{2N+1}). In particular, $d\mu_N$ is invariant under the flow of (A.6).

For U an open subset of $H^{\frac{1}{2}-}(\mathbb{T})$, we define the Wiener and Gibbs measures

of (A.2) by:

$$d\rho(U) := \lim_{N \to \infty} d\rho_N (U \cap E_N),$$

$$d\mu(U) := \lim_{N \to \infty} d\mu_N (U \cap E_N),$$
 (A.7)

where $E_N := \operatorname{Span}_{\mathbb{C}} \{ e^{2\pi i kx} : |k| \leq N \}$. It was shown by Zhidkov in [89] definition (A.7) of the Gibbs measure is equivalent to the original invariant measure shown to exist by Lebowitz, Rose, and Speer in [47]. We want to show that this measure is invariant under the flow of the nonlinear Schrödinger equation (A.2). A key ingredient for this is the approximation result given by Lemma A.2.1, stated in the next section, which relates the solutions of (A.6) to the solutions to (A.2).

Approximation lemma

Lemma A.2.1. Let s > 0 and suppose that $\varphi \in H^s(\mathbb{T})$, $\|\varphi\|_{H^s} \leq A$, and that $N \in \mathbb{N}$. Suppose that the solution of

$$\begin{cases} iv_t^N + \Delta v = \pm P_N(|v^N|^2 v^N) \\ v_0^N(x) = P_N \varphi(x) \end{cases}$$

satisfies $||v^N(t)||_s \leq A$ for all $t \in [0,T]$. Then for u satisfying

$$\begin{cases} iu_t + \Delta u = \pm |u|^2 u \\ u_0(x) = \varphi(x), \end{cases}$$

for $t \in [0,T]$, if $s_1 \in (0,s)$, we have the approximation

$$||u(t) - v^{N}(t)||_{H^{s_{1}}} \le \exp\left(C(A+1)^{C_{1}}T\right) N^{s_{1}-s}$$

provided that the quantity on the right hand side of (A.2.1) remains less than 1.

Lemma (A.2.1) is proved in [9, Lemma 2.27] and is similar to the proof of the approximation result for the quintic case proved Section 4.1.2. We do not include the details of the remainder of the construction of the measure for the defocusing case, which can be found in [9].

A.3 Focusing case

We now focus on the case of a non-positive nonlinearity. In particular, we recall the proof of Lemma 3.1.1, which was proven in [9, Lemma 3.10]. For the convenience of the reader, we present the full details of the proof in a self-contained way. For an alternative summary, see also [65, Section 2]. Before proceeding with the proof, we recall in Section A.4 several auxiliary results concerning Fourier multipliers in the periodic setting and concentration inequalities. In Section A.5, we recall the notion of a *norming set*, which we use to prove duality results in L^p spaces. The proof of Lemma 3.1.1 is given in Section A.6.

A.4 Auxiliary results

We recall the definition of a sub-gaussian random variable.

Definition A.4.1. Let $(\Omega, \mathcal{A}, \mathbb{P})$ be a probability space. We say a random variable X is sub-gaussian if there exist constants C, v > 0 such that for all t > 0 we have

$$\mathbb{P}(|X| > t) \le C \mathrm{e}^{-vt^2}.$$

We will use the following inequality about sub-gaussian random variables. For a proof, see [86, Proposition 5.10].

Lemma A.4.2 (Hoeffding's Inequality). Suppose that X_1, \ldots, X_N are all independent, centred sub-gaussian random variables. Let $Q := \max_i ||X_i||_{\psi_2}$ for

$$||X||_{\psi_2} := \sup_{p \ge 1} p^{-1/2} (\mathbb{E}|X|^p)^{1/p}$$

and let $a \in \mathbb{R}^N$. Then, for any t > 0, we have

$$\mathbb{P}\left[\left|\sum_{i=1}^{N} a_i X_i\right| > t\right] \lesssim \exp\left(-\frac{ct^2}{Q^2 \|a\|_{\ell^2}^2}\right) \,.$$

We will also need the Riesz-Thorin interpolation theorem, which is proved in [81, V].

Theorem A.4.3 (Riesz-Thorin theorem). Let (X, μ) and (Y, ν) be σ -finite measure spaces. Suppose that $p_0, p_1, q_0, q_1 \in [1, \infty]$ and suppose that $T : L^{p_j}(X, \mu) \to L^{q_j}(Y, \nu)$ is a bounded linear operator with norm K_j for j = 0, 1. Then $T : L^{p_\theta}(X, \mu) \to L^{q_\theta}(Y, \nu)$ is a bounded linear map with norm $K_\theta \leq K_0^{1-\theta} K_1^{\theta}$ for all

 $\theta \in [0, 1]$. Here

$$\begin{split} \frac{1}{p_{\theta}} &= \frac{1-\theta}{p_0} + \frac{\theta}{p_1}, \\ \frac{1}{q_{\theta}} &= \frac{1-\theta}{q_0} + \frac{\theta}{q_1}. \end{split}$$

A.5 Norming sets

To prove Lemma 3.1.1, we need the following result about duality in L^p spaces. We emphasise that this is a known result, but whose proof we could not find in the literature, so we write out the proof for the convenience of the reader.

Lemma A.5.1. Suppose that $\mathcal{M} \subset \mathbb{Z}$ has cardinality m, and let

$$S := \operatorname{Span}_{\mathbb{C}} \left\{ e^{2\pi i k x} : k \in \mathcal{M} \right\}$$

Then there is some subset Ξ of the unit sphere of $L^{p'}$ satisfying the following properties.

- 1. $\max_{\varphi \in \Xi} |\langle g, \varphi \rangle| \ge \frac{1}{2} ||g||_{L^p}$ for all $g \in S$.
- 2. $\log |\Xi| \leq Cm$ for some universal constant C > 0.

Remark. This result can be extended to finite dimensional subsets of normed vector spaces, but we do not need the result in full generality.

A.5.1 Norming sets and ε -nets

Before proceeding, we introduce several notions in Banach spaces.

Definition A.5.2. Let X be a Banach space, $Y \subset X$ a linear subspace, and $\theta \in (0, 1]$. We denote by X^* the (continuous) dual space of X. We say that a set $F \subset X^*$ is θ -norming over Y if

$$\sup_{g \in F \setminus \{0\}} \frac{g(y)}{\|g\|} \ge \theta \|y\|$$

for all $y \in Y$.

Definition A.5.3. Let X be a Banach space. Given $x \in X$ and $\varepsilon > 0$, we write $B_{\varepsilon}(x) = \{y \in X : ||x - y|| < \varepsilon\}$ for the ball in X of radius ε around x. Let $Y \subset X$ be a subset of X. Given $\varepsilon > 0$, we call $N_{\varepsilon} \subset Y$ an ε -net of Y if $Y \subset \bigcup_{x \in N_{\varepsilon}} B_{\varepsilon}(x)$.

We write $S_X := \{x \in X : ||x|| = 1\}$ for the unit sphere of X.

We want to relate norming sets to ε -nets. To do this, we take inspiration from the following result, the proof of which comes from [42, Section 17.2.4, Theorem 1].

Lemma A.5.4. Suppose X is a Banach space, $Y \subset X$ is a linear subspace, and $G \subset S_{X^*}$ a set that is 1-norming over Y. Let $\varepsilon \in (0, 1)$, and suppose that N_{ε} is an ε -net on the unit sphere of Y. For each element $x \in N_{\varepsilon}$, fix a functional $g_x \in G$ such that $g_x(x) > 1 - \varepsilon$ (which we can do since $N_{\varepsilon} \subset S_Y$ and G is 1-norming over Y). Then the set $F = \{g_x\}_{x \in N_{\varepsilon}}$ is θ -norming over S_Y for $\theta = 1 - 2\varepsilon$.

Proof. Let $y \in S_Y$. By definition, there is some $x_y \in N_{\varepsilon}$ satisfying $||y - x_y|| < \varepsilon$. Then, by definition of F, linearity, the definition of x_y , and $G \subset S_{X^*}$, we have

$$\sup_{g \in F} |g(y)| = \sup_{x \in N_{\varepsilon}} |g_x(y)| \ge |g_{x_y}(y)| = |g_{x_y}(x_y) - g_{x_y}(y - x_y)|$$

> $1 - \varepsilon - ||y - x_y|| > 1 - 2\varepsilon = \theta$.

A.5.2 Conclusion of the proof of Lemma A.5.1

We begin by bounding the size of an ε -net of \mathbb{C}^m . For $\mathcal{M} \subset \mathbb{Z}$ with $|\mathcal{M}| = m$, we consider the following norm on \mathbb{C}^m

$$\|\|(a)_{k\in\mathcal{M}}\|\| := \left\|\sum_{k\in\mathcal{M}} a_k \mathrm{e}^{2\pi i k x}\right\|_{L^{p'}}$$

We define $\Sigma := \{(a_k)_{k \in \mathcal{M}} : |||a||| = 1\}$. Notice that since \mathbb{C}^m is finite dimensional, the unit ball with respect to any norm is compact. Let N_{ε} be a maximal subset of Σ satisfying the property

$$x, y \in N_{\varepsilon}, x \neq y \implies |||x - y||| > \varepsilon.$$
 (A.8)

In other words, any subset of Σ strictly containing N_{ε} fails to have property (A.8). Such a set exists and is finite by the compactness of Σ . Any such set must be an ε net of Σ by maximality. We have the following bound, whose proof is an adaptation of [86, Lemma 5.2].

Lemma A.5.5. For $N_{\varepsilon} \subset \Sigma$ maximal satisfying (A.8), we have

$$|N_{\varepsilon}| \leq \left(1 + \frac{2}{\varepsilon}\right)^m =: C_{\varepsilon}^m.$$

Proof. The result follows from a volume bound. Since N_{ε} is ε -separated, it follows that $\{B_{\varepsilon/2}(x)\}_{x\in N_{\varepsilon}}$ are pairwise disjoint. Moreover, since $x \in \Sigma$, it follows from the triangle inequality that all such balls lie inside the ball of radius $1 + \varepsilon/2$ centred at the origin. So

$$\operatorname{vol}[B_{\varepsilon/2}(x)] \cdot |N_{\varepsilon}| \le \operatorname{vol}[B_{1+\varepsilon/2}(0)].$$
(A.9)

We also have the following identity

$$\operatorname{vol}[cB_1(0)] = \operatorname{vol}\left[\left\{(ca_k)_{k\in\mathcal{M}} : \left\|\sum_{k\in\mathcal{M}} a_k e^{2\pi i k x}\right\|_{L^{p'}} \le 1\right\}\right]$$
$$= c^m \operatorname{vol}[B_1(0)].$$

Combining this with (A.9) (and using translation invariance), we have

$$|N_{\varepsilon}| \le \left(\frac{1+\varepsilon/2}{\varepsilon/2}\right)^m = \left(1+\frac{2}{\varepsilon}\right)^m.$$

г			

We are now able to prove Lemma A.5.1.

Proof of Lemma A.5.1. We let $\Xi := N_{1/4} \subset \Sigma$ be obtained by setting $\varepsilon = \frac{1}{4}$ in the construction above. Let $g = \sum_{k \in \mathcal{M}} a_k e^{2\pi i k x}$. By duality, there is some $\psi \in \Sigma$ with $|\langle g, \psi \rangle| \geq \frac{3}{4} ||g||_{L^p}$. Moreover, since $N_{1/4}$ is a $\frac{1}{4}$ net of Σ , we can find $\varphi \in N_{1/4} \equiv \Xi$ with $||\psi - \varphi||_{L^{p'}} \leq \frac{1}{4}$. Hence, it follows that $|\langle g, \psi - \varphi \rangle| \leq \frac{1}{2} ||g||_{L^p}$. Therefore, we obtain

$$\frac{1}{2} \|g\|_{L^p} \le |\langle g, \psi \rangle| - |\langle g, \psi - \varphi \rangle| \le |\langle g, \psi \rangle - \langle g, \psi - \varphi \rangle| = |\langle g, \varphi \rangle|, \qquad (A.10)$$

where in the second step above, we used the reverse triangle inequality. The result follows from (A.10) and Lemma A.5.5. $\hfill \Box$

A.6 Proof of Lemma 3.1.1

We now prove Lemma 3.1.1, which was originally proved in [9, Lemma 3.10]. For the convenience of the reader, we present the full details of the proof in a self-contained way. Throughout, $(\mathbb{C}^{\mathbb{N}}, \mathcal{G}, \mu)$ is the probability space defined in (2.11) above.
Proof of Lemma 3.1.1. We show the following bound for large λ .

$$\mu\left[\left\|\sum_{k\in\mathbb{Z}} \left\|\frac{\omega_k}{\sqrt{\lambda_k}} e^{2\pi i k x}\right\|_{L^p} > \lambda, \left(\sum_{k\in\mathbb{Z}} \frac{|\omega_k|^2}{\lambda_k}\right)^{1/2} \le B\right] \lesssim \exp(-cM_0^{1+2/p}\lambda^2), \quad (A.11)$$

where

$$M_0 \sim \left(\frac{\lambda}{B}\right)^{\frac{1}{1/2 - 1/p}}.$$
(A.12)

Let us assume (A.11) and we show that it implies the claim. We write

$$F := \left\| \sum_{k \in \mathbb{Z}} \frac{\omega_k}{\sqrt{\lambda_k}} e^{2\pi i k x} \right\|_{L^p}^p \chi_{\left(\sum_{k \in \mathbb{Z}} \frac{|\omega_k|^2}{\lambda_k}\right)^{1/2} \le B}$$
$$G := e^{\frac{2}{p} \|\sum_{k \in \mathbb{Z}} \frac{\omega_k}{\sqrt{\lambda_k}} e^{2\pi i k x} \|_{L^p}^p} \chi_{\left(\sum_{k \in \mathbb{Z}} \frac{|\omega_k|^2}{\lambda_k}\right)^{1/2} \le B}$$

Then

$$\begin{split} \|G\|_{L^1} &= \int_{y>0} dy \, y \, \mu \, (|G| > y) \\ &\leq \int_{y>1} dy \, y \, \mu \, (|G| > y) + 1, \end{split}$$

where the inequality follows because μ is a probability measure. Now defining $y := \exp\left(\frac{2}{p}\lambda^p\right)$, we have

$$\|G\|_{L^{1}} \leq \int_{\lambda>0} d\lambda \, 2\lambda^{p-1} \mathrm{e}^{\frac{2}{p}\lambda^{p}} \mu\left(|F| > \lambda\right) + 1$$

$$\lesssim \int_{\lambda>0} d\lambda \, \exp\left(\frac{2}{p}\lambda^{p} - cB^{\frac{-2p+4}{p-2}}\lambda^{\frac{4p}{p-2}}\right) \lambda^{p-1} + 1. \tag{A.13}$$

Since $p < \frac{4p}{p-2}$ for $p \in [4,6)$ for $||G||_{L^1}$ to be finite, *B* can be arbitrary. We have $p = \frac{4p}{p-2}$ for p = 6, so in this case we have to take *B* sufficiently small.

We now prove (A.11). Throughout, M is a dyadic integer and $|k| \sim M$ means $\frac{3M}{4} \leq |k| < \frac{3M}{2}$. We make use of the following inequality.

$$\left\| \sum_{|k| \sim M} a_k e^{2\pi i k x} \right\|_{L^p} \lesssim M^{1/2 - 1/p} \left\| \sum_{|k| \sim M} a_k e^{2\pi i k x} \right\|_{L^2}.$$
 (A.14)

For p = 2, (A.14) is trivial, and for $p = \infty$, it follows from Cauchy-Schwarz and Plancherel's theorem. We then use the Riesz-Thorin interpolation theorem to deduce (A.14) for all $p \in (2, \infty)$.

With M_0 as in (A.12), we consider a sequence $(\sigma_M)_{M>M_0}$ of positive numbers with

$$\sum_{M>M_0} \sigma_M = \delta \,, \tag{A.15}$$

with $\delta > 0$ sufficiently small to be determined later.

Consider $\omega \in \Omega$ such that

$$\left\|\sum_{k\in\mathbb{Z}}\frac{\omega_k}{\sqrt{\lambda_k}}\mathrm{e}^{2\pi i k x}\right\|_{L^p} > \lambda, \quad \left(\sum_{k\in\mathbb{Z}}\frac{|\omega_k|^2}{\lambda_k}\right) \le B.$$
(A.16)

With ω as in (A.16), we show that there is some $M > M_0$ such that

$$\left\| \sum_{|k|\sim M} \omega_k \mathrm{e}^{2\pi i k x} \right\|_{L^p} > \sigma_M M \lambda \,. \tag{A.17}$$

We argue by contradiction. First, we note that for ω as in (A.16), we have

$$\sum_{M \le M_0} \left\| \sum_{|k| \sim M} \frac{\omega_k}{\sqrt{\lambda_k}} e^{2\pi i k x} \right\|_{L^p} \lesssim M_0^{1/2 - 1/p} \left(\sum_{k \in \mathbb{Z}} \frac{|\omega_k|^2}{\lambda_k} \right)^{1/2} \lesssim \lambda.$$
(A.18)

We used (A.14), Plancherel's theorem, and summed a geometric sequence for the first inequality in (A.18). For the second inequality in (A.18), we used the L^2 bound in (A.16), and (A.12). By taking the implied constant in (A.12) to be sufficiently small, let us note that the proof of (A.18) implies

$$\sum_{M \le M_0} \left\| \sum_{|k| \sim M} \frac{\omega_k}{\sqrt{\lambda_k}} e^{2\pi i k x} \right\|_{L^p} \le \frac{\lambda}{2}.$$
 (A.19)

We henceforth work with such a small implied constant in (A.12).

Suppose that (A.17) did not hold for any $M > M_0$. Then it would follow that, for an appropriate choice of δ in (A.15), we would have

$$\sum_{M>M_0} \left\| \sum_{|k|\sim M} \frac{\omega_k}{\sqrt{\lambda_k}} e^{2\pi i k x} \right\|_{L^p} \le \frac{\lambda}{2}.$$
 (A.20)

We note that (A.20) combined with (A.18) would give us a contradiction with the first inequality in (A.16). Let us explain how we have obtained (A.20). First we

note

$$\left\| \sum_{|k| \sim M} \frac{\omega_k}{\sqrt{\lambda_k}} e^{2\pi i k x} \right\|_{L^p} \le \frac{C}{M} \left\| \sum_{|k| \sim M} \omega_k e^{2\pi i k x} \right\|_{L^p}, \qquad (A.21)$$

which we justify as follows. Let $\Phi : \mathbb{R} \to \mathbb{C}$ be a smooth, compactly supported function which is equal to 1 on $1/2 \le |\xi| \le 2$ and zero for $|\xi| \le 1/4$ and $|\xi| \ge 4$. Consider the function

$$\Psi_M(\xi) := \frac{1}{\sqrt{\frac{4\pi^2 |\xi|^2}{M^2} + \frac{\kappa}{M^2}}} \Phi\left(\frac{\xi}{M}\right) \,. \tag{A.22}$$

Since

$$\tilde{\Phi}(\xi) := \frac{1}{\sqrt{4\pi^2 |\xi|^2 + (\kappa/M^2)}} \Phi(\xi)$$

has bounded derivatives of all order (with bound depending on κ), the same holds for $\Psi_M = \tilde{\Phi}(\xi/M)$ given by (A.22) above. Hence, the Mikhlin multiplier theorem (on \mathbb{R}) implies that the map T_M defined by $(T_M f)^{\hat{}}(\xi) := \Psi_M(\xi) \hat{f}(\xi)$ is bounded as a map on $L^p(\mathbb{R})$. Applying the support properties of Φ and using Lemma 1.7.2, we obtain

$$M \left\| \sum_{|k| \sim M} \frac{\omega_k}{\sqrt{\lambda_k}} e^{2\pi i k x} \right\|_{L^p} = M \left\| \sum_{|k| \sim M} \Phi_M(k) \frac{\omega_k}{\sqrt{\lambda_k}} e^{2\pi i k x} \right\|_{L^p}$$
$$= \left\| \sum_{|k| \sim M} \Psi_M(k) \omega_k e^{2\pi i k x} \right\|_{L^p}$$
$$\lesssim \left\| \sum_{|k| \sim M} \omega_k e^{2\pi i k x} \right\|_{L^p}.$$

Here we use the fact that the Fourier coefficients are supported on $|k| \sim M$, for which $\Phi_M(k) = 1$. We hence deduce (A.21). By summing in $M > M_0$ and applying (A.15) with δ sufficiently small, we obtain (A.20). Therefore (A.17) holds for some $M > M_0$.

To estimate the contribution for each dyadic M, we consider the subspace S of L^p given by $\operatorname{Span}_{\mathbb{C}}\{e^{2\pi i k x} : |k| \sim M\}$. We want to construct a $\frac{1}{2}$ -norming set, Ξ , contained in the unit sphere of $L^{p'}$ with the following properties.

- 1. $\max_{\varphi \in \Xi} |\langle g, \varphi \rangle| \ge \frac{1}{2} ||g||_{L^p}$ for all $g \in S$.
- 2. $\|\varphi\|_{L^2} \lesssim M^{1/2-1/p}$ for any $\varphi \in \Xi$.

3. $\log |\Xi| \lesssim M$.

To find this set, we apply Lemma A.5.1 and take the orthogonal projection of Ξ onto S. We obtain the first and third properties from Lemma A.5.1, and the second follows from Plancherel's theorem, Hölder's inequality, and the Hausdorff-Young inequality (applied to p'). Namely, for $\varphi \in \Xi$, we have

$$\begin{aligned} \|\varphi\|_{L^2} &= \|\hat{\varphi}\|_{\ell^2} \lesssim M^{1/2 - 1/p} \|\hat{\varphi}\|_{\ell^p} \\ &\lesssim M^{1/2 - 1/p} \|\varphi\|_{L^{p'}} = M^{1/2 - 1/p} \,. \end{aligned}$$

Having constructed the set, we now estimate the norm. We choose $M > M_0$ satisfying (A.17). Then

$$\begin{split} \sigma_M M \lambda < \left\| \sum_{|k| \sim M} \omega_k e^{2\pi i k x} \right\|_{L^p} &\leq 2 \max_{\varphi \in \Xi} \left| \sum_{|k| \sim M} \omega_k \hat{\varphi}(k) \right| \\ &= 2 \max_{\varphi \in \Xi} \left| \sum_{|k| \sim M} \omega_k \frac{\hat{\varphi}(k)}{\|\varphi\|_{L^2}} \right| \|\varphi\|_{L^2} \\ &\lesssim 2M^{1/2 - 1/p} \max_{\varphi \in \Xi} \left| \sum_{|k| \sim M} \omega_k \frac{\hat{\varphi}(k)}{\|\varphi\|_{L^2}} \right|, \end{split}$$

where the first line uses property (1) of Ξ and the final inequality follows from property (2) of Ξ . So

$$\sigma_M M^{1/2+1/p} \lambda \lesssim \max_{\varphi \in \Xi} \left| \sum_{|k| \sim M} \omega_k \frac{\hat{\varphi}(k)}{\|\varphi\|_{L^2}} \right|.$$
(A.23)

Let us take $(\sigma_M)_{M>M_0}$ satisfying (A.15) to be of the form

$$\sigma_M \sim M^{-1/p} + (M_0/M)^{1/2},$$
 (A.24)

for a suitable choice of implied constant. For $M > M_0$, let X_M denote the event

(A.23). Then

$$\mathbb{P}_{\omega}\left[\left\|\sum_{k\in\mathbb{Z}} \frac{\omega_{k}}{\sqrt{\lambda_{k}}} e^{2\pi i k x}\right\|_{L^{p}} > \lambda, \left(\sum_{k\in\mathbb{Z}} \frac{|\omega_{k}|^{2}}{\lambda_{k}}\right)^{1/2} \leq B\right]$$
$$\leq \mathbb{P}_{\omega}\left(\cup_{M>M_{0}} X_{M}\right) \tag{A.25}$$

$$\leq \sum_{M > M_0} \sum_{\varphi \in \Xi} \mathbb{P}_{\omega} \left[\left| \sum_{|k| \sim M} \omega_k \frac{\hat{\varphi}(k)}{\|\varphi\|_{L^2}} \right| \gtrsim \sigma_M M^{1/2 + 1/p} \lambda \right]$$
(A.26)

$$\lesssim \sum_{M > M_0} \sum_{\varphi \in \Xi} \exp\left(-cM^{1+2/p} \sigma_M^2 \lambda^2\right) \tag{A.27}$$

$$\lesssim \sum_{M > M_0} \exp\left(CM - cM^{1+2/p} \sigma_M^2 \lambda^2\right) \tag{A.28}$$

$$= \sum_{M>M_0} \exp\left(CM - c\left(M + M_0 M^{2/p} + 2M_0^{1/2} M^{1/2+1/p}\right)\lambda^2\right)$$
(A.29)

$$\lesssim \sum_{M > M_0} \exp\left(-cM_0 M^{2/p} \lambda^2\right)$$

$$\lesssim \exp\left(-cM_0^{1+2/p} \lambda^2\right).$$
(A.30)

$$\lesssim \exp\left(-cM_0^{1+2/p}\lambda^2
ight)\,.$$

Here, (A.25) follows from (A.23). (A.26) follows from a union bound. (A.27) comes from applying Lemma A.4.2 with $X_i = \omega_i$ and $a_i = \hat{\varphi}(i) / \|\varphi\|_{L^2}$ (so that $Q \sim 1$ and $||a||_{\ell^2} \leq 1$ by Plancherel's theorem), and for (A.28), we use property (3) of Ξ . (A.29) comes from (A.24). We obtain (A.30) from the fact that λ is large and noticing that the second term will give a factor less than one. The final inequality follows from the fact we have a geometric series with common ratio equal to $1 - \zeta_{M_0}$, with $\zeta_{M_0} > 0$. So we have shown (A.11), which completes the proof.

A.7 Higher dimensional constructions

In higher dimensions, the random Fourier series in (2.12) is less regular. In particular, in two spatial dimensions, a typical element in the support of the measure is contained in H^{0-} , meaning that the L^2 norm is almost surely infinite. To get around this issue, one has to (Wick) renormalise the interaction, in essence subtracting a diverging quantity to end up with an interaction which is non-infinite. This makes proving that the weight function is in L^1 significantly more difficult. It also means that there is no longer a deterministic global well-posedness theory for the underlying equation. One instead needs to prove probabilistic local well-posed results, showing that the equation is locally well-posed for typical element in the support of the Gibbs measure.

In two dimensions and for a defocusing potential, this is the content of [10]. It was also shown in [18] that Wick ordering and a suitable truncation does not produce a well-defined measure in the case of focusing local NLS, although Bourgain showed in [11] that given suitable assumption on the interaction potential, one can construct the measure for the Hartree equation.

In two dimensions and p an even integer, the construction of the measure for the suitably renormalised equation with nonlinearity $|u|^{p}u$ also began with the works of Nelson [58] (as well as [37,75]). A summary of the construction of the Gibbs measure in this case can be found in [64].

In three dimensions, the measure has been constructed for the Hartree equation for certain interaction potentials w_{β} which act like $|x|^{-(d-\beta)}$. This was done for $\beta > 2$ by Bourgain in [9] and by Deng-Nahmod-Yue for $\beta > 1 - \varepsilon$ in [23]. The construction of the measure in the case of general $\beta > 0$ and the local nonlinearity remains open in three dimensions.

In the cubic case, in dimensions greater than three, it is expected that the measure will be a Gaussian measure for any normalisation of the potential energy. This was proved for $d \ge 5$ independently by Aizenman and Fröhlich in [1] and [28] respectively. It was also proved for d = 4 in the case of the real-valued measure by Aizenman and Duminil-Copin in [2]. The same result is expected to hold for d = 4 in the case of the complex-valued measure.

For a summary of the construction results of Gibbs measures for non-Schrödinger equations, we point the reader for example to the summary in [16, Section 1.1].

Appendix B

Creation and annihilation operators

B.1 Basic computations

In this section, we record some basic computations. First we compute the commutation relations. We recall that for $f \in L^2$ and $\Psi = (\psi^{(0)}, \psi^{(1)}...) \in \mathcal{F}$, we define

$$(b(f)\Psi)^{(n)}(x_1,\ldots,x_n) := \sqrt{n+1} \int dx \,\overline{f(x)}\psi^{(n+1)}(x,x_1,\ldots,x_n),$$
$$(b^*(f)\Psi)^{(n)}(x_1,\ldots,x_n) := \frac{1}{\sqrt{n}} \sum_{i=1}^n f(x_i)\psi^{(n-1)}(x_1,\ldots,x_{i-1},x_{i+1},\ldots,x_n)$$

To simplify notation, we denote by $\mathbf{x} := x_1, \ldots, x_n$ and $\mathbf{x}_{\not{i}} := x_1, \ldots, x_{i-1}, x_{i+1}, \ldots, x_n$. We show that

$$[b(f), b^*(g)] = \langle f, g \rangle_{L^2} \mathbf{1}_{\mathcal{F}}.$$
(B.1)

We have

$$(b(f)b^{*}(g)\Psi)^{(n)}(\mathbf{x}) = \sqrt{n+1} \int dx \,\overline{f(x)} \, (b^{*}(g)\Psi)^{(n+1)}(x,\mathbf{x}) = \frac{\sqrt{n+1}}{\sqrt{n+1}} \int dx \left(\overline{f(x)} \sum_{i=1}^{n} g(x_{i})\psi^{(n)}(x,\mathbf{x}_{i}) + \overline{f(x)}g(x)\psi^{(n)}(\mathbf{x}) \right).$$

Similarly

$$(b^{*}(f)b(g)\Psi)^{(n)}(\mathbf{x}) := \frac{1}{\sqrt{n}} \sum_{i=1}^{n} g(x_{i}) (b(g)\Psi)^{(n-1)} (\mathbf{x}_{\not{i}})$$
$$= \frac{\sqrt{n}}{\sqrt{n}} \sum_{i=1}^{n} g(x_{i}) \int dx \,\overline{f(x)}\psi^{(n)}(x, \mathbf{x}_{\not{i}}).$$

(B.1) follows from the definition of the commutator.

We also show that

$$\left(\int dx \, b^*(x)b(x)\right)\Big|_{\mathfrak{h}^{(n)}} = n\mathbf{1}_{\mathfrak{h}^{(n)}},$$

where we recall that $b(x) = b(\delta_x)$ and similarly $b^*(x) = b^*(\delta_x)$. Then

$$\begin{split} \left[\left(\int dx \, b^*(x) b(x) \right) \Psi \right]^{(n)}(\mathbf{x}) &= \frac{1}{\sqrt{n}} \int dx \, \sum_{i=1}^n \delta(x-x_i) (b^*(x) \Psi)^{(n-1)}(\mathbf{x}_{\not i}) \\ &= \frac{\sqrt{n}}{\sqrt{n}} \int dx \, \sum_{i=1}^n \delta(x-x_i) \int dy \, \delta(x-y) \psi^{(n)}(y, \mathbf{x}_{\not i}) \\ &= \sum_{i=1}^n \int dx \, \delta(y-x_i) \psi^{(n)}(y, \mathbf{x}_{\not i}) = n \psi^{(n)}(\mathbf{x}), \end{split}$$

where the final line uses the fact bosonic wavefunctions are symmetric.

More details on creation and annihilation operators, both in the bosonic and fermionic case can be found for example in [13].

Bibliography

- [1] M. Aizenman, Proof of the triviality of φ_d^4 field theory and some mean-field features of Ising models for d > 4, Physical Review Letters **47** (1981): 1-4.
- M. Aizenman, H. Duminil-Copin. Marginal triviality of the scaling limits of critical 4D Ising and φ⁴₄ models, Ann. Math. 194 (2021), no. 1, 163–235.
- [3] Z. Ammari, V. Sohinger, Gibbs measures as unique KMS equilibrium states of Hamiltonian PDEs, Rev. Mat. Iberoam. 23 (2023), no. 1, 29–90.
- [4] V.I. Arnold, Mathematical Methods of Classical Mechanics, Springer, New York, (1978).
- [5] N. Barashkov, M. Gubinelli, A variational method for Φ⁴₃, Duke Math. J. 169 (2020), no. 17, 3339–3415.
- [6] N. Benedikter, M. Porta, B. Schlein, *Effective Evolution Equations from Quantum Dynamics*, Springer Briefs in Mathematical Physics, Springer, Berlin (2016).
- [7] A. Bényi, T. Oh, The Sobolev inequality on the torus revisited, Publ. Math. Debrecen 83 (2013), no. 3, 359–374.
- [8] J. Bourgain, Fourier transform restriction phenomena for certain lattice subsets and applications to nonlinear evolution equations, Geom. Funct. Anal. 3 (1993), no. 2, 107–156.
- J. Bourgain, Periodic nonlinear Schrödinger equation and invariant measures, Comm. Math. Phys. 166 (1994), no. 1, 1–26.
- [10] J. Bourgain, Invariant measures for the 2D-defocusing nonlinear Schrödinger equation, Comm. Math. Phys. 176 (1996), no. 8, 421–445.
- [11] J. Bourgain, Invariant measures for the Gross-Pitaevskii equation, J. Math. Pures Appl. (9) 76 (1997), no. 8, 649–702.
- [12] J. Bourgain, C. Demeter, The proof of the l² decoupling conjecture, Annals of mathematics, 351-389, (2015).
- [13] O. Bratteli, D. W. Robinson, Operator algebras and quantum statistical mechanics 2, Springer, (2002).
- [14] B. Bringmann, Invariant Gibbs measures for the three-dimensional wave equation with a Hartree nonlinearity I: measures, Stoch. Partial Differ. Equ. Anal. Comput. 10 (2022), no. 1, 1–89.
- [15] B. Bringmann, Invariant Gibbs measures for the three-dimensional wave equation with a Hartree nonlinearity II: Dynamics, Preprint arXiv: 2009.04616 (2020).

- [16] B. Bringmann, Y. Deng, A. Nahmod, H. Yue, Invariant Gibbs measures for the three dimensional cubic nonlinear wave equation, Preprint arXiv 2205.03893 (2022).
- [17] D. Brydges, J. Fröhlich, A.D. Sokal, A new proof of the existence and nontriviality of the continuum φ_2^4 and φ_3^4 theories, Comm. Math. Phys. **91** (1983), no. 2, 141–186.
- [18] D. Brydges, G. Slade, Statistical mechanics of the 2-dimensional focusing nonlinear Schrödinger equation, Comm. Math. Phys. 182 (1996), no. 2, 485–504.
- [19] N. Burq, L. Thomann, N. Tzvetkov, Remarks on the Gibbs measures for nonlinear dispersive equations, Ann. Fac. Sci. Toulouse Math. 6 (2018), 27(3): 527–597.
- [20] E. Carlen, J. Fröhlich, J. Lebowitz, Exponential relaxation to equilibrium for a onedimensional focusing non-linear Schrödinger equation with noise, Comm. Math. Phys. 342 (2016), no. 1, 303–332.
- [21] T. Chen, N. Pavlović, On the Cauchy problem for focusing and defocusing Gross-Pitaevskii hierarchies, Discrete Contin. Dyn. Syst. 27 (2010), no. 2, 715–739.
- [22] X. Chen, Second order corrections to mean field evolution for weakly interacting bosons in the case of three-body interactions, Arch. Ration. Mech. Anal. 203 (2012), no. 2, 455–497.
- [23] Y. Deng, A. R Nahmod, and H. Yue, Invariant Gibbs measure and global strong solutions for the Hartree NLS equation in dimension three, J. of Math. Phys. 62 (2021), no. 3, 031514.
- [24] Y. Deng, A. Nahmod, H. Yue, Random tensors, propagation of randomness, and nonlinear dispersive equations, Invent. Math. 228 (2022), no. 2, 539–686.
- [25] V. D. Dinh, N. Rougerie, Invariant Gibbs measures for 1D NLS in a trap, arXiv: 2301.02544, 2023.
- [26] M.B. Erdogan, N. Tzirakis, Dispersive Partial Differential Equations: Wellposedness and Applications, London Mathematical Society Student Texts 86.
- [27] J. Forlano, L. Tolomeo, Quasi-invariance of Gaussian measures of negative regularity for fractional nonlinear Schrödinger equations, Preprint, arXiv:2205.11453 (2022).
- [28] J. Fröhlich, On the triviality of $\lambda \varphi_d^4$ theories and the approach to the critical point in $d >_{(-)} 4$ dimensions, Nucl. Phys. B **200** (1982), no. 2, 281–296.
- [29] J. Fröhlich, A. Knowles, B. Schlein, V. Sohinger, Gibbs Measures of Nonlinear Schrödinger Equations as Limits of Many-Body Quantum States in Dimensions d ≤ 3, Comm. Math. Phys., 356(3), 2017, pp.883–980.
- [30] J. Fröhlich, A. Knowles, B. Schlein, V. Sohinger, A microscopic derivation of timedependent correlation functions of the 1D cubic nonlinear Schrödinger equation, Advances in Mathematics (2019), no. 353, 67–115.
- [31] J. Fröhlich, A. Knowles, B. Schlein, and V. Sohinger. The mean-field limit of quantum Bose gases at positive temperature, J. Amer. Math. Soc. 35 (2022), no. 4, 955–1030.
- [32] J. Fröhlich, A. Knowles, B. Schlein, and V. Sohinger. A path-integral analysis of interacting Bose gases and loop gases, J. Stat. Phys. 180 (2020), no. 1–6, 810–831.
- [33] J. Fröhlich, A. Knowles, B. Schlein, and V. Sohinger. Interacting loop ensembles and Bose gases, Ann. Henri Poincaré 24 (2023), no. 5, 1439–1503.

- [34] J. Fröhlich, A. Knowles, B. Schlein, and V. Sohinger. The Euclidean Φ₂⁴ theory as a limit of an interacting Bose gas, Preprint arXiv: 2201.07632 (2022). To appear in J. Eur. Math. Soc. (JEMS).
- [35] J. Ginibre, G. Velo, The classical field limit of scattering theory for non-relativistic many-boson systems. I and II, Commun.Math. Phys. 68, 45–68 (1979).
- [36] R.T. Glassey, Asymptotic behavior of solutions to certain nonlinear Schrödinger-Hartree equations, Comm. Math. Phys. 53 (1977), 9–18.
- [37] J. Glimm, A. Jaffe, Quantum Physics. A Functional Integral Point of View, Springer-Verlag, Second edition, 1987.
- [38] A. Grünrock, On the Cauchy- and periodic boundary value problem for a certain class of derivative nonlinear Schrödinger equations, Preprint arXiv: 0006195v1 (2000).
- [39] M. Gubinelli, M. Hofmanová, A PDE construction of the Euclidean φ₃⁴ quantum field theory, Comm. Math. Phys. 384 (2021), no. 1, 1–75.
- [40] T. Gunaratnam, T. Oh, N. Tzvetkov, H. Weber, Quasi-invariant Gaussian measures for the nonlinear wave equation in three dimensions, Probab. Math. Phys. 3 (2022), no. 2, 343–379.
- [41] K. Hepp, The classical limit for quantum mechanical correlation functions, Commun.Math. Phys. 35, 265–277 (1974).
- [42] V. Kadets, A Course in Functional Analysis and Measure Theory, Springer. (2018).
- [43] C. Kenig, G. Ponce, L. Vega, Well-posedeness and scattering results for the generalized Korteweg-de Vries equation via contraction principle, Comm. Pure Appl. Math. 46 (1993), no. 4, 527–620.
- [44] C. Kenig, G. Ponce, L. Vega, The Cauchy problem for the Korteweg-de Vries equation in Sobolev spaces of negative indices, Duke Math. J. 71 (1993), no. 1, 1–21.
- [45] A. Knowles, *Limiting dynamics in large quantum systems*, ETH Zürich Doctoral Thesis, 2009, ETHZ e-collection 18517.
- [46] H. H. Kuo, Gaussian measures in Banach spaces, Lecture Notes in Math. vol. 463 (1975), Springer-Verlag.
- [47] J. Lebowitz, H. Rose, E. Speer, Statistical mechanics of the nonlinear Schrödinger equation, J. Stat. Phys. 50 (1988), 657–687.
- [48] J. Lee, Rate of convergence towards mean-field evolution for weakly interacting bosons with singular three-body interactions, Preprint arXiv:2006.13040 (2020).
- [49] M. Lewin, P.-T. Nam, N. Rougerie, Derivation of nonlinear Gibbs measures from manybody quantum mechanics, Journal de l'École Polytechnique-Mathématiques 2 (2015), 65–115.
- [50] M. Lewin, P.-T. Nam, N. Rougerie, Gibbs measures based on 1D (an)harmonic oscillators as mean-field limits, J. Math. Phys. 59 (2018), no. 4, 041901.
- [51] M. Lewin, P.-T. Nam, N. Rougerie, Classical field theory limit of 2D many-body quantum Gibbs states, Preprint arXiv:1810.08370v1 (2018).
- [52] M. Lewin, P.-T. Nam, N. Rougerie, Derivation of renormalized Gibbs measures from equilibrium many-body quantum Bose gases, J. Math. Phys, 60 (2019), no.6, 061901, 11 pp.

- [53] M. Lewin, P.-T. Nam, N. Rougerie, Classical field theory limit of many-body quantum Gibbs states in 2D and 3D, Invent. Math. 224 (2021), no. 2, 315–444.
- [54] R. Liang, Y. Wang, Gibbs measure for the focusing fractional NLS on the torus, SIAM J. Math. Anal. 54 (2022), no. 6, 6096–6118.
- [55] F. Linares, G. Ponce, Introduction to Nonlinear Dispersive Equations, Universitext, Springer, Berlin, (2009).
- [56] H. P. McKean, K. L. Vaninsky, Action-angle variables for the cubic Schrödinger equation, Comm. Pure Appl. Math. 50 (1997), no. 6, 489–562.
- [57] H. P. McKean, K. L. Vaninsky, Cubic Schrödinger: the petit canonical ensemble in action-angle variables, Comm. Pure Appl. Math. 50 (1997), no. 7, 593–622.
- [58] E. Nelson The free Markoff field, J. Functional Anal. 12 (1973), no. 2, 211–227.
- [59] E. Nelson, Probability theory and Euclidean field theory, Constructive quantum field theory, Springer, 1973, pp. 94–124.
- [60] A. Nahmod, G. Staffilani, Randomness and nonlinear evolution equations, Acta Math. Sin. (Engl. Ser.) 35 (2019), no. 6, 903–932.
- [61] P. T. Nam, J. Ricaud, A. Triay, Ground state energy of the low density Bose gas with three-body interactions, J. Math. Phys. 63 (2022), no. 7, Paper No. 071903, 12 pp.
- [62] P. T. Nam, J. Ricaud, A. Triay, Dilute Bose gas with three-body interaction: recent results and open questions, J. Math. Phys. 63 (2022), no. 6, Paper No. 061103, 13 pp.
- [63] P. T. Nam, J. Ricaud, A. Triay, The condensation of a trapped dilute Bose gas with three-body interactions, Probab. Math. Phys. 4 (2023), no. 1, 91–149.
- [64] T. Oh, L. Thomann, A pedestrian approach to the invariant Gibbs measures for the 2-d defocusing nonlinear Schrödinger equations, Stoch. Partial Differ. Equ. Anal. Comput. 6 (2018), no. 3, 397–445.
- [65] T. Oh, P. Sosoe, L. Tolomeo, Optimal integrability threshold for Gibbs measures associated with focusing NLS on the torus, Invent. Math. 227 (2022), no. 3, 1323–1429.
- [66] M. Reed, B. Simon, Methods of modern Mathematical Physics I: Functional analysis, Revised and Enlarged Edition, Academic Press, 1980.
- [67] M. Reed, B. Simon, Methods of modern Mathematical Physics II: Fourier Analysis, Self-adjointness, Academic Press, 1975.
- [68] T. Robert, K. Seong, L. Tolomeo, Y. Wang, Focusing Gibbs measures with harmonic potential, Preprint arXiv: 2212.11386 (2022).
- [69] A. Rout, V. Sohinger, A microscopic derivation of Gibbs measures for the 1D focusing cubic nonlinear Schrödinger equation, Preprint arXiv: 2206.03392 (2022). To appear in Communications in Partial Differential Equations.
- [70] A. Rout, V. Sohinger A microscopic derivation of Gibbs measures for the 1D focusing quintic nonlinear Schrödinger equation, Preprint arXiv: 2308.06569 (2023).
- [71] M. Salmhofer, Functional Integral and Stochastic Representations for Ensembles of Identical Bosons on a Lattice, Comm. Math. Phys. 385 (2021), no. 2, 1163–1211.
- B. Schlein, Derivation of effective evolution equations from microscopic quantum dynamics, Chapter in Evolution Equations. Clay Mathematics Proceedings, Vol. 17 Eds.
 D. Ellwood, I. Rodnianski, G. Staffilani, J. Wunsch (2013).

- [73] S. Schwabik, G. Ye, Topics in Banach space integration (Vol. 10). World Scientific. (2005).
- [74] S. Sheffield, Gaussian free fields for mathematicians, Probab. Theory Relat. Fields, 139 (2007), 521–541.
- [75] B. Simon, The $P(\Phi)_2$ Euclidean (Quantum) Field Theory, Princeton Univ. Press, 1974.
- [76] B. Simon, Trace ideals and their Applications, Amer. Math. Soc., Second Edition, 2005.
- [77] A. Skorokhod, Integration in Hilbert space, Ergebnisse der Mathematik und ihrer Grenzgebiete, Springer-Verlag, 1974.
- [78] H. Spohn, Kinetic Equations from Hamiltonian Dynamics, Rev. Mod. Phys, 52 (1980), no. 3, 569–615.
- [79] V. Sohinger, Bounds on the growth of high Sobolev norms of solutions to nonlinear Schrödinger equations Ph.D. Thesis, MIT, 2011.
- [80] V. Sohinger, A microscopic derivation of Gibbs measures for nonlinear Schrödinger equations with unbounded interaction potentials, Int. Math. Res. Not. IMRN (2022), no. 19, 14964–15063.
- [81] E. M. Stein, G. Weiss, Introduction to Fourier Analysis on Euclidean Spaces (PMS-32), Volume 32, Princeton University Press, 2016.
- [82] T. Tao, Global behaviour of nonlinear dispersive and wave equations, Curr. Dev. in Math.(2006), 255-340
- [83] T. Tao, Nonlinear dispersive equations: Local and global analysis, AMS, CBMS Lecture Notes, 2006.
- [84] L. Tolomeo, H. Weber, Phase transition for invariant measures of the focusing Schrödinger equation, Preprint arXiv: 2306.07697 (2023).
- [85] N. Tzvetkov, Quasi-invariant Gaussian measures for one dimensional Hamiltonian PDE's, Forum Math. Sigma 3 (2015), e28, 35 pp.
- [86] R. Vershynin, Introduction to the non-asymptotic analysis of random matrices, In: Y.C. Eldar, G. Kutyniok, eds, Compressed Sensing: Theory and Applications, Cambridge: Cambridge University Press. (2012).
- [87] T. Xian, Optimal mass normalizability for Gibbs measure associated with NLS on the 2D disc, Preprint arXiv: 2204.09561 (2022).
- [88] Z. Xie, Derivation of a nonlinear Schrödinger equation with a general power-type nonlinearity in d = 1, 2, Differential Integral Equations 28 (2015), no. 5-6, 455–504.
- [89] P.E. Zhidkov, An invariant measure for the nonlinear Schrödinger equation (Russian), Dokl. Akad. Nauk. SSSR **317** (1991), 543–546; translation in Soviet Math. Dokl. **43**, 431–434.